The role of fibroblasts in chemoresistance in triple negative breast cancer

Robyn Victoria Broad

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

The University of Leeds School of Medicine

October 2018

Intellectual property and publication statement

The candidate confirms that the work submitted is her own and that appropriate credit has been given where reference has been made to the work of others.

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgment.

Acknowledgments

I would like to thank my supervisors Dr Thomas Hughes, Dr James Thorne and Professor Andy Hanby for their support throughout my PhD. I would also like to thank members of the Hughes group (past and present) for their advice and help as well as the rest of the staff on Level 9.

I would like to thank the University of Leeds for my scholarship and importantly BCRAG for funding my consumables throughout my PhD.

Thesis abstract

Triple negative breast cancer (TNBC) accounts for 10-15% of breast cancers and is associated with relatively poor prognosis. TNBC is characterised by lack of ER, PR and Her2 expression, and consequently does not respond to the targeted treatments routinely used to treat other breast cancers. Therefore, cytotoxic chemotherapy is the only systemic treatment for primary TNBC in routine use. Resistance to chemotherapy in TNBC is a major clinical issue and is reflected in the relatively high rate of early recurrences. Cancer associated fibroblasts (CAFs) are the most abundant cell type in the tumour stroma and have been shown to influence behaviour of the tumour cells. My aims were to determine whether CAFs modify responses of TNBC cells to chemotherapy, to identify molecular pathways responsible for this cross-talk, and to develop these pathways as potential therapeutic targets to enhance chemotherapy responses.

I have shown that breast CAFs, but not normal fibroblasts, dose-dependently protect the claudin-low TNBC cell lines MDA-MB-231 and MDA-MB-157 from the chemotherapeutic epirubicin. By contrast, the basal claudin-high TNBC line MDA-MB-468 was not protected, demonstrating that the CAFs' influence is specific to certain TNBC characteristics. Transcriptome profiling demonstrated that CAF-induced protection of MDA-MB-231 cells was associated with deregulation of specific genes and pathways, including activation of the interferon-signaling pathway. Recombinant interferons were sufficient to protect MDA-MB-231 and MDA-MB-157 cells, implicating IFN as the key mediator of CAF-induced protection. Inhibition of the activation of interferon signaling using blocking antibodies, JAK/STAT inhibitors or under-/over-expression of specific signaling intermediates (miR-155) reduced CAF-dependent protection of MDA-MB-231 and MDA-MB-157 cells, suggesting that CAF-induced up-regulation of interferon signaling is necessary for CAF-induced protection.

I concluded that CAF-induced activation of interferon signaling in certain TNBCs is a potential mechanism of chemoresistance, and therefore that addition of inhibitors of interferon signaling to chemotherapy regimens in patients with stromal-rich TNBCs may improve chemotherapy responses.

Contents

Intellectual property statement	2
Acknowledgements	3
Thesis abstract	4
Contents	5
Abbreviations	16
Chapter 1- Introduction	19
1.1 Breast cancer epidemiology and development	19
1.2 Breast cancer classification	20
1.2.1 Histological classification of breast cancers	21
1.2.1.1 Classification of in situ disease	21
1.2.1.2 Classification of invasive disease	22
1.2.2 Clinical molecular classification of breast cancers	23
1.2.2.1 Luminal A and B breast cancers	24
1.2.2.2 HER2 positive breast cancer	24
1.2.2.3 Triple negative breast cancer	24
1.2.2.4 Basal-like cancer	25
1.3 Breast cancer treatment	26
1.3.1 Local therapy	26
1.3.2 Cytotoxic chemotherapy	27
1.3.3 Targeted systemic treatments	28
1.4 Triple negative breast cancer	29
1.4.1 Triple Negative breast cancer epidemiology and clinical	
characteristics	29
1.4.2 Triple negative breast cancer classification	30
1.5 Triple negative breast cancer treatment	34
1.5.1 Triple negative breast cancer responses to treatment and	
prognosis	34
1.5.2 Current therapies for triple negative breast cancer	35
1.5.2.1 Mechanism of action of breast cancer cytotoxics	35
1.5.3 Therapies under investigation for triple negative breast cancer	36

1.6 Mechanisms of chemoresistance in breast cancer	38
1.6.1 Physical mechanisms of chemoresistance	39
1.6.2 Drug inactivation mechanisms of chemoresistance	39
1.6.3 Cellular mechanisms of chemoresistance	40
1.7 Tumour stroma	41
1.8 Fibroblasts	42
1.9 Origins of cancer associated fibroblasts	44
1.10 Cancer-associated fibroblasts in breast cancer progression	45
1.11 The interferon family	46
1.12 Interferon biogenesis	47
1.13 Interferon signaling	47
1.13.1 Type I interferon signaling	49
1.13.2 Type II interferon signaling	49
1.13.3 Regulation of interferon signaling	50
1.13.4 Interferon stimulated genes (ISGs)	51
1.14 Interferons in cancer	52
1.14.1 Extrinsic roles of interferons in cancer	52
1.14.2 Intrinsic roles of interferons in cancer	55
1.15 MiRNAs	57
1.16 MiRNA biogenesis	57
1.17 MiRNA silencing of RNA	59
1.18 MiRNAs in cancer	60
1.19 Hypothesis and aims	62
Chapter 2- Materials and Methods	63
2.1 Cell lines and tissue culture	63
2.2 Transfection of cell lines	65
2.3 Chemotherapy agents, chemical inhibitors, recombinant interferons,	
blocking antibodies	66
2.4 Dual luciferase reporter assays	67
2.5 Luciferase-based chemo-survival assays	67
2.5.1 Dose determination	67
2.5.2 Co-culture	68

2.6 MTT assays	68	
2.7 Mono-culture colony forming assays		
2.8 Co-culture colony forming assays		
2.9 Preparation of fibroblast conditioned media, and use in colony forming		
chemo-survival assays	73	
2.10 Anoikis assay	73	
2.11 Analysis of transcriptomes for MDA-MB-231 cells cultured with and with	hout	
CAF1 cells	74	
2.11.1 mRNA transcriptome analyses	74	
2.11.2 miRNA analyses	75	
2.12 RNA extractions and qPCR	76	
2.12.1 miRNA qPCR	76	
2.12.2 mRNA qPCR	78	
2.13 Flow cytometric assessment of epirubicin loading or cell cycle	79	
Chapter 3- Cancer-associated fibroblasts protect some triple negative breas	st	
cancer cell types from chemotherapeutics	83	
3.1 Abstract	83	
3.2 Introduction	84	
3.3 Results	86	
3.3.1 Breast fibroblasts can protect TNBC cell lines from the		
chemotherapeutic epirubicin	86	
3.3.2 Immortalised breast CAF1s protect claudin-low TNBC cells (MI	DA-	
MB-157 and MDA-MB-231), but not claudin-high MDA-MB-468-GFP	cells	
from epirubicin	90	
3.3.3 Primary breast CAFs, but not primary NFs, protect the claudin-	low	
TNBC cell line MDA-MB-231 from epirubicin	93	
3.3.4 Immortalised CAF1 cells protect MDA-MB-231s from an alterna	ative	
chemotherapeutic, docetaxel	95	
3.3.5 Fibroblast-dependent protection of claudin-low TNBC cell lines		
MDA-MB-157 and MDA-MB-231 is not reproduced by fibroblast		
conditioned medium	96	

3.4 Discussion	98
3.4.1 Methodology for assessing chemo-response	98
3.4.2 Conditioned media clonogenic assays – a soluble mediator of	
cross-talk between CAFs and cancer cells?	100
3.4.3 CAFs induce enhanced clonogenicity	102
3.5 Conclusions and aspects of future work	103
Chapter 4- Cancer-associated fibroblasts induce up-regulation of interferon	
signaling in breast epithelial cancer cells, leading to chemoresistance	104
4.1 Abstract	104
4.2 Introduction	105
4.3 Results	107
4.3.1 CAF1s induce changes to the mRNA transcriptome of MDA-ME	3-
231 cells during epirubicin treatment	107
4.3.2 CAF1s induce changes to the miRNA transcriptome of MDA-M	В-
231 cells during epirubicin treatment	109
4.3.3 MiR-155 and miR-27a do not have consistent, significant	
Influences on survival of MDA-MB-231 cells in the absence or	
presence of epirubicin treatment, although miR-155 demonstrated a	
potentially relevant trend	113
4.3.4. MiR-155 may have a role in chemo-response in MDA-MB-157	
cells	117
4.3.5 Co-culture with CAF1s during epirubicin treatment induces	
deregulation of two major biological pathways in MDA-MB-231 cells	118
4.3.6 Interferon signaling is stimulated by CAF1s in epithelial cell type	es
that are protected from epirubicin treatment by CAF1s, but not in cell	S
that are not	121
4.3.7 Recombinant IFNs are sufficient to induce chemoresistance in	
MDA-MB-231 and MDA-MB-157 cells, but not MDA-MB-468	123
4.3.8 CAF1s, but not NF1s, activate interferon signaling in co-culture	with
MDA-MB-231 cells	128
4.3.9 IFN- β is secreted by CAF1s, and not NFs, following co-culture	with
MDA-MB-231 cells, but not in CAF1 mono-cultures	130

4.4 Discussion	100
	100
4.4.1 The choice of expression profiling methodology	133
4.4.2 miR-155: an oncomiR, as well as potential role in	
chemoresistance?	135
4.4.3 miR-27a	138
4.5 Conclusions and aspects of future work	139
Chapter 5- Inhibition of interferon signaling reduces fibroblast-induced	
chemoresistance in triple negative breast cancer	140
5.1 Abstract	140
5.2 Introduction	141
5.3 Results	143
5.3.1 Type I and/or Type II interferon blocking antibodies individually	,
block CAF1-induced chemoresistance in TNBC cells but the combine	ation
of both antibodies has no increased effect	143
5.3.2 Type I, but not Type II, interferon blocking antibodies block CA	F1-
induced chemoresistance in MDA-MB-157 cells	147
5.3.3 Type I interferon blocking antibodies block pCAF-induced	
chemoresistance in MDA-MB-231 cells	150
5.3.4 The JAK-STAT inhibitor ruxolitinib reduces CAF1-induced	
chemoresistance in MDA-MB-231 and MDA-MB-157 cells	152
5.3.5 Ruxolitinib inhibits pCAF-induced chemoresistance in MDA-ME	3-231
cells	155
5.3.6 MiR-155 inhibition reduces CAF1-induced chemoresistance of	
MDA-MB-231 cells	156
5.4 Discussion	158
5.4.1 Use of antibodies for inhibition of interferon signaling	158
5.4.2 Use of small molecules for inhibition of interferon signaling	161
5.4.3 MiRNAs or miRNA inhibitors (specifically miR-155 inhibitor) as	
viable therapeutic options.	162
5.5 Conclusion and aspects of future work	164

Chapter 6- What are the cellular mechanisms of cancer-associated fibro	<u>blasts-</u>
induced chemoresistance in breast cancer cells?	165
6.1 Abstract	105
6. I Abstract	105
6.2 Introduction	166
6.3 Results	168
6.3.1 CAF1 cells increase intra-cellular epirubicin levels in co-cult	ured
MDA-MB-231 cells	168
6.3.2 CAF1s may induce changes in cell cycle progression	170
6.3.3 CAF1s significantly increase anoikis resistance in MDA-MB-	-231
and MDA-MB-157	172
6.4 Discussion	174
6.4.1 Anoikis resistance	174
6.4.2 Cell cycle changes and chemoresistance	177
Chapter 7- Discussion	179
7.1.1 CAF-induced therapy resistance in breast cancer	179
7.1.2 Interferons as mediators of chemoresistance in TNBC	183
7.1.3 Claudin-low TNBC phenotype in CAF-induced chemoresistance	187
7.2 Limitations	188
7.3 Final conclusions	189
Appendix	190
References	197

List of Tables

Table 1.1: Characteristics of the main clinical subtypes of invasive breast cancer	25
Table 1.2: Triple negative breast cancer subtypes	33
Table 1.3: Classes of interferons	48
Table 2.1: MiRNA mimics and inhibitors used for transfection	66
Table 2.2: Thermal cycling protocol required for Taqman MegaPlex miRNA cards reverse transcription	76
Table 2.3: Thermal cycling protocol required for TaqMan miRNA assays rev transcription	erse 77
<u>Table 2.4:</u> Thermal cycling protocol required for SYBR green mRNA assays reverse transcription	78
Table 2.5: Primer sequence for target and control mRNA qPCR primers	79
Table 2.6: Voltages for channels and compensation matrix used for drug loading	80
Table 2.7: Voltages for channels and compensation matrix used for cell cycl analysis	le 81
<u>Table 4.1:</u> Summary of gene expression changes in MDA-MB-231 cells between the 0% fibroblast and 20% fibroblast cultures	109
<u>Table 4.2:</u> Numbers of miRNAs differentially expressed between MDA-MB-2 cells cultured alone (0%) or with CAF1 cells (20%) during epirubicin	231
ισαιποπί	110

Table 4.3: Five miRNAs that were differentially expressed between MDA-M	B-
231 cells cultured alone (0%) or with CAF1s (20%) were selected for further	r
study	111
Table 4.4: Molecular pathways enriched in the differentially expressed gene	s in
MDA-MB-231 cells after co-culture with CAF1s, as identified using the	
ToppGene portal	119
<u>Table 4.5:</u> Genes in the interferon signaling pathway that were significantly	up-
regulated in MDA-MB-231 cells by co-culture with CAF1s during epirubicin	
treatment	120
Table 7.1: Key features of Ham et al and Boelens et al compared to work	
carried out in this thesis	182
<u>Appendix Table 1:</u> Full list of gene expression changes between 0% and 20)%
cultures following Affymetrix analysis	190
Appendix Table 2: List of miRNAs differentially expressed that are present i	n
0% cultures but not 20% and miRNAs present in 20% cultures but not 0%	191
Appendix Table 3: List of miDNAs differentially expressed that are decreased	ad
from 0% to 20% and miRNAs that are increased from 0% to 20%	-u 102
	192
Appendix Table 4: List of extracellular matrix genes up-regulated from 0% to	0
20% cultures from Affymetrix analysis	193
Appendix Table 5: List of dysregulated pathways taken from ToppGene	
pathway analysis. Top hits were then used to create Table 4.4	195

List of Figures

Figure 1.1: Breast cancer development from normal breast structures to invasive carcinoma	20
Figure 1.2: Type I and II Interferon signaling	50
Figure 1.3: Extrinsic roles of interferons on immune-regulation	54
Figure 1.4: Intrinsic roles of interferons on tumour cells	56
Figure 1.5: MiRNA biogenesis	59
Figure 2.1: Comparison of 2 independent counts of colony forming assay plat of MDA-MB-468 cells	ies 70
Figure 2.2: FACS cell sorting plots for MDA-MB-231 GFP/luc cells co-cultured with 20% CAF1 cells	ժ 71
Figure 2.3: FACS cell sorting plots for MDA-MB-468 GFP cells co-cultured wir 20% CAF1 cells	th 72
Figure 2.4: FACS cell sorting plots for MDA-MB-157 cells co-cultured with 20 ^o CAF1-GFP cells	% 72
Figure 2.5: Flow cytometry gating strategy for epirubicin loading assessments	80
Figure 2.6: Flow cytometry gating strategy for cell cycle assessments	82
Figure 3.1: CAF1 enhance chemo-survival in MDA-MB-231 cells with various proportions of fibroblasts after 24h epirubicin treatment	88
Figure 3.2: CAF1s enhance chemo-survival in MDA-MB-231 cells with various proportions of fibroblasts after 72h epirubicin treatment	s 89
Figure 3.3: CAF1s, but not NF1s, protect MDA-MB-231 cells, but not MDA-MI 468 cells, from epirubicin	B- 92
Figure 3.4: CAF1s protect MDA-MB-157 cells from epirubicin	93
Figure 3.5: pCAFs but not matched pNFs, protect MDA-MB-231 cells from epirubicin	94

Figure 3.6: CAF1s protect MDA-MB-231 cells from docetaxel	95
Figure 3.7: Fibroblast conditioned media does not protect MDA-MB-231, -46 and -157 cells when treated with epirubicin	68 97
Figure 4.1: Supervised hierarchical clustering of expression profiles from biological triplicate samples of MDA-MB-231 cells when in culture alone (0% with CAF1 cells (20%) following epirubicin treatment	%) or 108
Figure 4.2: MiR-155 and miR-27a are up-regulated in MDA-MB-231 cells in presence of CAF1s during epirubicin treatment	the 112
Figure 4.3: MiR-155 and miR-27a mimics and inhibitors successfully over- express or knockdown appropriate miRNA	113
Figure 4.4: MiR-155 and miR-27a do not have consistent, significant influen on survival of MDA-MB-231 cells in the absence or presence of epirubicin treatment using short-term assays	ices 115
Figure 4.5: MiR-155 did not have significant influences on survival of MDA-I 157 cells in the absence or presence of epirubicin treatment using short-tern assays	MB- m 117
Figure 4.6: Interferon response genes are up-regulated in MDA-MB-231 and MDA-MB-157 cells by CAF1s, but not in MDA-MB-468 cells	d 122
Figure 4.7: Recombinant IFNs are sufficient to stimulate chemoresistance ir MDA-MB-231	ו 125
Figure 4.8: Recombinant IFNs are sufficient to stimulate chemoresistance in MDA-MB-231 and MDA-MB-157 cells but not MDA-MB-468 cells	ר 126
Figure 4.9: MDA-MB-468 are unable to activate interferon signaling in the presence of recombinant IFNs	127
Figure 4.10: CAF1s activate ISRE and GAS activity in co-cultured MDA-MB cells	-231 129
<u>Figure 4.11</u> : IFN- β 1 expression in CAF1s is upregulated in the presence of MDA-MB-231 cells	132
Figure 5.1: Blocking Type I and Type II interferon signaling prevents CAF1- induced chemoresistance in MDA-MB-231 cells	145

<u>Figure 5.2</u> : Interferon neutralizing antibodies successfully blocked CAF1- dependent interferon activation in MDA-MB-231 cells when treated with	
epirubicin	146
<u>Figure 5.3</u> : Type I but not Type II interferon neutralisation blocks immortalis CAF1-induced chemoresistance in MDA-MB-157 cells	ed 148
<u>Figure 5.4</u> : Type I but not Type II interferon neutralizing antibodies block CA dependent interferon activation in MDA-MB-157 cells	+F1- 149
<u>Figure 5.5</u> : Type I interferon neutralisation blocks pCAF-induced chemoresistance, and pCAF-induced interferon activation in MDA-MB-231 cells	151
Figure 5.6: Ruxolitinib reduces, although not significantly, CAF1-induced chemoresistance in MDA-MB-231	153
Figure 5.7: Ruxolitinib significantly reduces CAF1-induced chemoresistance MDA-MB-157 cells	in 154
Figure 5.8: Ruxolitinib significantly inhibits pCAF-induced chemoresistance MDA-MB-231	in 155
Figure 5.9: MiR-155 inhibition significantly inhibits CAF1-induced chemoresistance in MDA-MB-231 cells	157
Figure 6.1: CAF1s enhance intra-cellular epirubicin levels in MDA-MB-231	169
Figure 6.2: CAF1s do not induce significant differences in epithelial cell cycl profile	e 171
Figure 6.3: CAF1s enhance anoikis resistance in MDA-MB-231- and -157 cells	173
Appendix Figure 1: Dose response curves used to calculate IC10, IC50 and	I
IC75 epirubicin values for MDA-MB-231 and -468 cells	196

Abbreviations

Ago2- Argonaute 2 protein ALDH- Aldehyde dehydrogenase Bad- Bcl-2-associated death promoter protein BASC- BRCA1-associated genome surveillance complex Bax- Bcl-2-like protein 4 Bcl-2- B cell lymphoma 2 Bcl-xl- B cell lymphoma-extra large BCRP- Breast cancer resistance protein BCS- Breast conserving surgery Bid- BH3 interacting-domain death agonist Bim- Bcl-2-like protein 11 BL1- Basal-like-1 BL2- Basal like-2 BLIA- Basal-like immune-activated BLIS- Basal-like immune-suppressed CAF- Cancer associated fibroblast CDK- cyclin-dependent kinase CRK/L- Crk adaptor protein/ligand CT- Computerised tomography CXCR/L- CXC-motif chemokine receptor/ ligand CYP- Cytochrome P450 **DC- Dendritic cells** DCIS- Ductal Carcinoma in situ DGSCR8- DiGeorge syndrome critical region gene 8 DNA- Deoxyribonucleic acid ECM- Extracellular matrix EGF- Epidermal growth Factor EGFR- Epidermal growth factor receptor 1 EMT- Epithelial mesenchymal transition ER- Oestrogen receptor FACS- Fluorescence activated cell sorting FAK-Focal adhesion kinase FAP- Fibroblast activated protein FGF- Fibroblast growth factor FOXO3a- Forkhead box O3 GAS- IFN-y- activated site GDP/GTP- Guanine di/triphosphate GFP- Green fluorescent protein HDAC- Histone deacetvlase HER2- Human epidermal growth factor receptor 2 HGF- Hepatocyte growth factor ICAM1- Intra-cellular adhesion molecule I IFITM1- Interferon induced transmembrane protein 1 **IFN-Interferon** IFNAR- Interferon-α receptor IFNGR- IFN-y receptor IFNLR- IFN- λ receptor 1 IGF/R- Insulin growth factor/receptor

IGFBP- Insulin-like growth factor binding proteins IKKε- IkB kinase ε IL-6- Interleukin-6 IM- Immunomodulatory INPP4B- Inositol polyphosphate 4-phosphotase Type II IRDS- IFN-related DNA damage resistance signature **IRF-IFN** response factors IRF- Interferon regulatory factor ISG- Interferon stimulated gene ISGF3- ISG Factor 3 ISRE- IFN-stimulated response element JAK- Janus activated kinase LAR- Luminal androgen receptor LOX- Lysyl oxidase M- Mesenchymal MAL- MYD88-adaptor like MAPK- Mitogen activated protein kinase Mcl-1- Myeloid leukemia cell differentiation protein 1 MDR1- Multidrug resistance protein 1 MDSC- Myeloid-derived suppressor cells MEK- Mitogen activated protein kinase miRNA- microRNA MLH1- MutL homolog 1 MMP- Matrix metalloproteinase MRI- Magnetic resonance imaging MRP1- Multi-drug resistance protein-1 MSC- Mesenchymal stem cells MSH2- MutS protein homolog 2 MSL- Mesenchymal-like mTOR- Mammalian targets of rapamycin MTT- 3-(4,5-dimethylthiazol-2-yl)-2,5-diphenyltetrazolium bromide MX- Myxovirus resistance MYD88- Myeloid differentiation primary response 88 NAC- Neoadjuvant chemotherapy NF(1/2)- Immortalised normal fibroblasts(1/2) NF-kB- nuclear factor-kB NHS- National Health Service NK cells- Natural killer cells NSCLC- Non small cell lung carcinoma NST- No special type OAS- Oligoadenylate synthase PACT- protein kinase R-activating protein PAMP- Pattern-associated molecular patterns PARP- Poly-ADP ribose polymerase pCAF- Primary cancer-associated fibroblast pCR- Pathological complete response pNF- Primary normal fibroblasts PD-1/PDL-1- Programmed cell death protein/ ligand 1 PDCD- Programmed cell death PDGF- Platelet derived growth factor PI3K- Phosphatidylinositide-3-kinase

PIP_{2/3}- Phosphatidylinositol-4,5- bis/triphosphate

PR- Progesterone receptor

PRR- Pattern recognition receptors

PTEN- Phosphatase and tensin homolog

q/RT/PCR- Quantitative/ Real time polymerase chain reaction

RAP1A- Ras related protein Rap-1A precursor

RISC- RNA-induced silencing complex

RNA- Ribonucleic acid

SDF-1- Stromal derived factor-1

SOCS- Suppressor of cytokine signaling

SPARC- Secreted protein, acidic and rich in cysteine

STAT- Signal transducers and activator of transcription proteins

TBK- TANK-binding kinase 1

TDLU- Terminal duct lobular unit

TGF- β - Transforming growth factor- β

TLR-Toll-like receptor

TNBC- Triple negative breast cancer

TNM- Tumour node metastasis

TPM1- Tropomyosin1

TRAM- TRIF-related adaptor molecule

TRBP- Trans-activating response RNA binding protein

TRIF- TIR-domain-containing adapter-inducing interferon-β

TYK- Tyrosine kinase

UNC-Unclassified

USP18- Ubiquitin specific peptidase 18

UTR- Un-translated region

VCAM1- Vascular adhesion molecule I

VEGF- Vascular endothelial growth factor

WNT- Wingless/Integrated

α-SMA- α-smooth muscle actin

Chapter 1– Introduction

1.1 Breast cancer epidemiology and development

Breast cancer is the most common cancer in the UK, with around 54,900 cases a year accounting for 15% of all new cancer cases in 2015; this means there are approximately 150 new breast cancer diagnoses every day (Cancer Research UK). Breast cancer is very prevalent in females, but also presents, rarely, in males amounting to approximately 380 new cases of male breast cancer each year (Cancer Research UK). Breast cancer is responsible for the 4th highest mortality of all cancer types and accounts for 7% of all cancer related deaths. In females, it is the 2nd highest cause of cancer related deaths, accounting for 15% of female cancer mortalities with 11,563 deaths associated with breast cancer in 2016 (Cancer Resarch UK). Breast cancer survival rates have almost doubled in the last 40 years. In the 1970s, 40% of women diagnosed with breast cancer survived longer than 10 years, however now this figure is now 78%. Currently 95% of women diagnosed with breast cancer survive a year, with 65% surviving 20 years or more (Cancer Research UK).

Cancers, including breast cancer, are formed from the poorly controlled division of cells, which is driven by an accumulation of genetic and epigenetic changes. Mammary epithelial cells that have acquired mutations over time divide rapidly and have the ability to sustain proliferation, evade growth suppression signals and resist cell death (Hanahan et al, 2000).

Breast cancers are initially confined to epithelial-lined compartments of the breast termed the lobules and ducts and in most cases are believed to arise in the Terminal Duct Lobular Unit (TDLU) (Wellings et al, 1973) (Figure 1.1). Whilst confined to epithelial-lined compartments, the cancer is known as *in-situ* breast cancer and is non-invasive (Cowell et al, 2013). Further proliferation of the breast cancer cells, confined in the lobules and ducts, leads to the integrity of the basement membrane of these breast structures being lost or fragmented, and it is at this stage that the cancer becomes invasive (Duffy et al, 2000). Cancer cells are then able to infiltrate into the surrounding breast tissue and

can continue to grow as well as migrate throughout the breast tissue (Weigelt et al, 2005). It is in the surrounding breast tissue where the tumour will create its own niche and microenvironment, which in turn will aid the tumour's growth and spread into lymph and blood vessels (Wiseman et al, 2002). Infiltration into blood and lymph vessels and circulatory systems can enable tumour cells to travel to secondary sites such as the brain, lung and bone (Lacroix et al, 2006). Here, the tumour cells can begin to grow and form metastases, which impair the normal functions of distant organs. It is commonly the formation of secondary breast cancers that is responsible for the death of breast cancer patients (Hart et al, 1980).



Figure 1.1: Breast cancer development from normal breast structures to invasive carcinoma

Normal breast tissues include normal epithelial cells (pink) surrounding open lumens. Breast cancer cells (purple) are initially confined to within the lumen of the breast ducts and lobules, and are referred to as carcinoma *in situ*. Breast carcinomas become invasive when they break down the basement membrane of the ducts/lobules, and invade into the surrounding tissue.

1.2 Breast cancer classification

Breast cancer is not one homogenous cancer type, and it is now commonly recognized that there are multiple different classes of breast cancer (Perou et al, 2000), which can be defined both histologically and molecularly. Both classifications are discussed in this section.

1.2.1 Histological classification of breast cancers

One major classification system for breast cancer is based on their microscopic appearance, as determined by histopathologists (Rosai, 2011). Importantly, histological classification is not just taxonomical, but the different subtypes it defines correlate with prognosis, and patients with these different disease subtypes are recommended for different treatment strategies (Makki et al, 2015).

Initially, malignant lesions are split into 2 groups: *in situ* carcinomas and invasive carcinomas (see Figure 1.1) (Burstein et al, 2004). This distinction is not one of type but how far the disease has progressed. *In situ* disease occurs prior to invasive carcinomas (Hu et al, 2008).

1.2.1.1 Classification of in situ disease

Whilst it was historically believed that *in situ* carcinomas originated from ductal (confined to breast ducts) or lobular (confined to breast lobules) compartments, this is not the case (Wellings et al, 1973). The difference between what is called lobular and ductal is merely derived from different molecular repertoires governing the morphology of lobular neoplasia in particular, where inactivation of the adhesion complex, normally via malfunction of E-cadherin, occurs (Mastracci et al, 2005). Nearly all breast cancer derives from the TDLU (Wellings et al, 1973).

At a practical/clinical level the most important determinant in ductal carcinoma *in situ* (DCIS) is nuclear grade (Lagios et al, 1989). DCIS also can grow in a variety of morphologies; according to Makki et al (Makki et al, 2015), DCIS can be further classified as:

- Comedo or solid (cancer cells have completely filled the duct)
- Cribform (gaps between cancer cells within the duct)
- Micropapillary (dilated ducts lined by stratified population of monotonous cells)
- Papillary (asymmetrical fern like pattern within the duct)

However, mixed patterns often occur making the usefulness of the architectural classifications minimal (Vinay et al, 2010). Modern practice would be to record the nuclear grade of the DCIS, which is generally mandatory in minimum data sets. The pattern is generally not mandatory data (Pinder et al, 2010).

1.2.1.2 Classification of invasive disease

Invasive carcinomas are also highly heterogenous and can be further classified into tumours with very distinct morphological features known as 'specical type'. (Shi et al, 2017). These 'special type' tumours include tubular, ductal lobular, invasive lobular, mucinous (colloid) and medullary. They all contain some reproducible characteristics that can be recognized down the microscope with varying degrees of agreement (Page, 2003).

The remainder of invasive carcinomas (70-80%) are put in the category of invasive ductal (infiltrating ductal) of no special type (NST) (Lakhani et al, 2012). It is important to recognize that invasive ductal carcinoma NST is not a specific type of invasive carcinoma but any carcinoma for which no specific morphological profile has yet been determined (Malhotra et al, 2010).

To attempt to predict better the behavior of breast cancers, histological grading was divised by Bloom and Richardson (Bloom et al, 1957) and later modified by Elston and Ellis (Elston et al, 1991). This grading system gives some guide to the biological aggression of the tumour. Grade is determined by a combined score derived from the degree of tubule formation, the number of mitotic figures in the tumour site, and the nuclear pleomorphism. Each are scored 1-3. Tumour grading is determined from the additive score from each of the components; 3-5 is assigned grade I, 6-7 grade II and 8-9 Grade III (Bloom et al, 1957, Elston et al, 1991).

As described by Bloom and Richarson, and Elston and Ellis, grade 1 breast cancer cells are normally well differentiated with a high proportion of the tumour (above 75%) showing normal breast tissue structures. Grade 1 tumours have a low number of mitotic cells and the cells are small and uniform with minimal

variation in size and shape. Grade 2 tumours are moderately differentiated, where the tumour has between 10%-75% of normal breast structures and a moderate mitotic rate. The tumour cells have moderate nuclear variation so are slightly bigger than normal cells and display some variation in shape. Grade 3 tumours are poorly differentiated with less than 10% of the tumour displaying normal breast structures. Grade 3 tumours have a high mitotic rate and the tumours cells look markedly different to normal cells with varied size and shape. Histological grading can be a strong predictor of patient outcome; for example, a study of 2219 breast cancer cases demonstrated that histological grade was strongly associated with both breast cancer specific survival and disease-free survival (Rakha et al, 2008).

The stage of the cancer is an assessment of the size of the tumour, lymph node involvement and spread of the cancer beyond the breast. The staging has been classified by the tumour, node, metastasis (TNM) system, which has been characterized by the American joint committee of cancer (American Cancer Society). Stage 1 breast cancer tumours are classically no larger than 2cm and in some cases there are also small clusters of cancer cells in the lymph node. Stage 2 breast tumours are typically between 2-5cm in size normally with axillary lymph node involvement. Stage 3 tumours are progressively larger than Stage 2 tumours and are more likely to have spread to multiple lymph nodes. Finally stage 4 breast cancers are commonly referred to as metastatic or advanced breast cancer. The breast cancer in stage 4 has spread beyond the breast tissue into secondary sites such as the brain, bone and lungs (American Cancer Society).

1.2.2 Clinical molecular classification of breast cancers

In seminal work by Perou, it was shown that molecular profiles could also be used to separate subtypes of breast cancer (Perou et al, 1999). Four subtypes of breast cancer are the mainstay of the current molecular classification used in clinical practice; these are Luminal A and B, HER2 subtype and triple negative breast cancers (Perou et al, 2000). Breast cancer subtypes were originally defined based on microarray-based mRNA expression profiles, but are classically defined clinically using the surrogate markers oestrogen receptor (ER), progesterone receptor (PR), HER2 and Ki67 (Perou et al, 2000). The characteristics of all these subtypes, and their standard treatment regimens, are summarized in Table 1.1.

1.2.2.1 Luminal A and B breast cancer

ER positive breast cancers account for around 70% of all invasive breast cancers and can be further divided in to Luminal A and Luminal B breast cancers. Luminal A and B cancers can also have progesterone receptor (PR) expression (Vallejos et al, 2010). Luminal A and B cancers are distinguished between using Ki67 expression, which is a proliferation marker. Luminal B breast cancers have higher levels of Ki67 and can also express high levels of HER2 (Inic et al, 2014). Both Luminal A and B breast cancer patients tend to have a relatively good prognosis, however Luminal B tends to have slightly worse prognosis than Luminal A, which is presumably related to the higher proliferation rate (Parise et al, 2014).

1.2.2.2 HER2 positive breast cancer.

HER2 invasive breast cancer accounts for 15% of all breast cancers and, likened to its name, is characterised by high gene and protein expression of HER2 and most, but not all, are ER negative. (Cooke et al, 2001).

1.2.2.3 Triple negative breast cancer

Another classified group of invasive breast cancer is triple negative breast cancer (TNBC); these will be discussed in more detail further on in section 1.4. TNBCs are ER and PR negative, and are also HER2 negative. Many TNBC have basal characteristics and as such express high molecular weight cytokeratins CK 5 and 14 and also EGFR1 (Sørlie et al, 2001). However, 25% of TNBC do not fall into the basal-subtype despite a substantial overlap of characteristics (Rakha et al, 2008). As I discuss later, the taxonomy of these tumours is quite complex with a great deal of hidden diversity.

1.2.2.4 Basal-like cancer

As indicated above, basal-like cancers express genes that are commonly associated with normal breast epithelial cells. In order for identification of basal like breast cancer, expression of growth factors and growth factor receptors (EGFR1, VEGF and IGFR) are determined as well as having a specific cytokeratin profile (Nielsen et al, 2004).

	Luminal A and B breast cancer	HER2 breast cancer	<u>Triple</u> negative breast cancer
Occurrence	70% of all invasive breast cancers	15% of all invasive breast cancers	10-15% of all invasive breast cancers
Classification	Presence of Oestrogen (ERa) and Progesterone (PR) receptors	No ER/PR HER2 genomic amplification, and high expression of Her2 protein	Lack of ER, PR and Her2
Systemic Treatment	Endocrine therapy (tamoxifen or aromatase inhibitors) Chemotherapy (mainly high grade or stage tumours)	Targeted Her2 treatment, such as trastuzumab Chemotherapy	No targeted treatment Chemotherapy
Typical Prognosis	Good	Poor	Poor

Table 1.1: Characteristics of the main clinical subtypes of invasive breast cancer

The main clinical molecular subtypes of invasive breast cancer are shown highlighting their occurrence, classification, treatment options and prognosis.

1.3 Breast cancer treatment

A diverse range of treatments are recommended for primary breast cancer patients, with the treatment plan dependent substantially on the breast cancer subtype as well as the patient's choice, age and overall health. The various treatment options available to patients will be covered in 3 different subsections below.

1.3.1 Local therapy

Most breast cancer patients will be treated surgically by either a mastectomy or breast conserving surgery (BCS). Mastectomies remove all breast tissue including the nipple and most of the skin that covers the breast (Sarrazin et al, 1989). In breast conserving resections, the surgery is not as drastic as a mastectomy. The cancerous cells are removed as well as a margin of normal tissue surrounding the cancer, but the whole breast is not removed (Chen et al, 2016). The recommendation to either undergo breast conserving surgery or breast mastectomy is determined by clinicopathological findings and radiology. Patient choice is equally as important with the patients' quality of life also taken into account regarding which surgery suits their needs best (Morrow et al, 2009). For larger tumours, patients could be advised to have a mastectomy, although in some cases shinking the tumour with neoadjuvant chemotherapy (NAC) can allow for more conservative surgery if the lesion shrinks sufficiently (Trimble et al, 1993).

Smaller tumours are more likely to be treated with BCS. In BCS, adjuvant (after surgery) radiotherapy is recommended for BCS patients to provide a localized therapy to target any remaining tumour cells within the breast tissue (Dean et al, 2009). Trials have shown that this combination of BCS and radiotherapy has a local recurrence rate that is not different from mastectomy (Chen et al, 2017), demonstrating that the approach is safe. Mastectomy patients may also receive radiotherapy if they show factors that are indicative of poor prognosis such as positive lymph nodes (Tsoutsou et al, 2013).

1.3.2 Cytotoxic chemotherapy

Despite targeted treatments being available for specific breast cancer subtypes, some breast cancer patients also receive traditional cytotoxic chemotherapy in addition to targeted treatments (Slamon et al, 2001). Around 34% of breast cancer patients receive chemotherapy for treatment of primary breast cancer (National Cancer Registration & Analysis Service and Cancer Research UK, 2017). The decision to undergo chemotherapy treatment is strongly influenced by stage, grade and size at diagnosis. For example, only 17% of stage I breast cancer patients will receive chemotherapy in comparison to 70.5% of stage III breast cancer patients (National Cancer Registration & Analysis Service and Cancer patients will receive chemotherapy in comparison to 70.5% of stage III breast cancer patients (National Cancer Registration & Analysis Service and Cancer Research UK, 2017).

Breast cancer patients can either receive systemic treatment prior to surgery (neoadjuvant) or after surgery (adjuvant); in both cases, the primary aim is to reduce the chances of metastatic recurrences by targeting sub-clinical micrometastases (EBCTCG, 2018). In addition, neoadjuvant chemotherapy (NAC) aims to reduce the primary tumour size to enable the option of breast conserving surgery instead of mastectomy. NAC can also be used to gauge drug response of the tumour (Thompson et al, 2012). An advantage of NAC is the monitoring of the primary tumour response to chemotherapy as a different chemotherapy drug could be used if the response in the primary tumour is minimal; the assumption being that the response of micrometastases will mirror the primary tumour (Buchholz et al, 2003).

Response rates can be determined through non-invasive imaging (MRI or ultrasound), to determine if the primary tumour has changed in size (Chen et al, 2007). In the metastatic setting, response can also be determined by imaging of overall tumour burden using scintigraphy (bone scan), MRI and CT scans as well as assessing circulating tumour markers (Woolf et al, 2015). In the case of adjuvant chemotherapy it is not possible to monitor response as the primary tumour has been removed and any sub-clinical micrometastases are, by definition, not detectable (Chew et al, 2001). Effectiveness of adjuvant

treatment can only be assessed from subsequent frequency and time post treatment of any recurrences (Isakoff et al, 2010).

Luminal A and B patients receive chemotherapy as well as endocrine therapy if there are indications of poor prognosis such as involved lymph nodes, which are indicative of likely cancer metastasis (NCCN guidelines, 2016). HER2 positive patients are given chemotherapy in combination with targeted Herceptin treatment (Piccart-Gebhart et al, 2005). However, because of the lack of molecular targets for precision therapy in TNBC patients, chemotherapy is the only systemic therapy routinely available to them. TNBC patients will therefore receive chemotherapy, while chemotherapy is only given to a minority of patients with other subtypes of primary breast cancer (Yershal et al, 2014).

1.3.3 Targeted systemic treatments

Molecularly-targeted treatments are also available to breast cancer patients but they are dependent on the subtype of breast cancer. Two classes of targeted agent are commonly used: endocrine agents, or HER2-targeted agents (Masoud et al, 2017).

Luminal A and B breast cancer patients can receive targeted endocrine therapy because Luminal A and B breast cancers express the ER, which is the target of endocrine agents. An example of endocrine therapy is tamoxifen, which is a selective ER modulator (Nolvadex Adjuvant Trial Organisation, 1998). Tamoxifen recognizes and competitively binds to the ER thereby reducing ER driven proliferation of the cancer cells (McDonnell et al, 2010).

HER2 expression, in Luminal B and HER2 subtypes of breast cancer, also provides a route for targeted therapy. Herceptin (trastuzumab) targets HER2 positive cells and binds to the extracellular domain of HER2 on the membrane of the cell (Baselga et al, 1998). Upon binding to HER2, Herceptin prevents receptor dimerization and therefore prevents the activation of the HER2 signaling pathway (Clarke et al, 2015). Consequently, inhibition of HER2 dimerization prevents activation of the MAPK and PI3K/Akt pathway leading to an increase in cell cycle arrest and suppression of tumour growth and survival (Vu et al, 2012). Herceptin also has the ability to attract immune cells to HER2 positive cancer cells, such as natural killer cells, which leads to antibody dependent cellular cytotoxicity and lysis of HER2 positive breast cancer cells (Park et al, 2010). Herceptin can be given alongside chemotherapy (Baselga et al, 2001) and can also be used in combination with another agent, pertuzumab (Harbeck et al, 2013). Pertuzumab is a targeted therapy that binds specifically to HER2 to prevent HER2 dimerisation. Pertuzumab, therefore, prevents activation of signaling pathways that normally stimulate cell proliferation and tumour growth (Baselga et al, 2012). The use of Herceptin and pertuzumab when combined with chemotherapy in both the neoadjuvant and adjuvant setting have been shown to increase cell death in early HER2 positive breast cancer patients (Gianni et al, 2016, von Minckwitz et al, 2017).

1.4 Triple negative breast cancer

TNBC epidemiology, classification, treatment and prognosis will be discussed in the next sections to highlight the importance of further investigation into treatment options for TNBC patients.

1.4.1 Triple negative breast cancer epidemiology and clinical characteristics

TNBC accounts for 10-15% of all breast cancers in caucasians (Badve et al, 2011), and commonly affects a younger population of women who are often pre-menopausal. The mean diagnosis age of TNBC is 46 years (±12.2 years) in comparison to other breast cancer patients for whom the mean diagnosis age is 53 years (±9.78 years) (Sajid et al, 2014). In the TNBC group, 65.9% were younger than 50 years old with 34.1% older than 50 years old (Sajid et al, 2014).

Studies have also shown a higher proportion of TNBC cases in African-American who are pre-menopausal. 39% of pre-menopausal breast cancer cases in African-American women are TNBC. This is compared to 14% in postmenopausal African-American women and 16% any aged non-African-American women (Carey et al, 2006).

BRCA1/2 mutation carriers have also shown to have a link to TNBC with a lifetime risk of developing TNBC at 50%-80%. From a recent study analyzing 3,791 BRCA1/2 mutation carriers diagnosed with invasive breast cancer, 68% were TNBC and 32% non-TNBC (Mavaddat et al, 2012). However, it is important to note that not all TNBCs are derived from BRCA1/2 mutations (Young et al, 2009, Peshkin et al, 2010). In addition, somatic mutations of TP53 (82%) frequently occur in TNBC cancers in comparison to Luminal A, Luminal B and HER2 cancers (12%, 29% and 72%) (Cancer Genome Altas Network, 2012).

TNBCs are highly aggressive with high proliferation and migratory properties. Because of their high proliferation rate, TNBCs grow quickly in comparison to other breast cancer subtypes (Matos et al, 2005). TNBC patients commonly present with metastatic recurrences, with secondary tumours presenting themselves in common breast metastases sites such as the brain, bone and lungs (Fulford et al, 2007). 5-year disease free survival of TNBC is the worst for all breast cancer subtypes, at 69.1% in comparison to the highest 5-year disease free survival of Luminal A cancers at 92.1%. At 5 years, overall survival of TNBC is 78.5% in comparison to Luminal A 95.1% (Hennigs et al, 2016, Zhang et al, 2013).

1.4.2 Triple negative breast cancer classification

As previously described, TNBCs lack expression of ER, PR and HER2 but even with these characteristics in common, TNBCs define a very diverse group of tumours, which are heterogenous in morphological and molecular terms (Perou et al, 2000). In previous years, there has been large debate over the subclasses within TNBC even now there is still some dispute between scientists on the classifications.

For example, using gene expression profiles and clustering analysis, Lehmann et al (Lehman et al, 2011) determined there are 6 TNBC subtypes: luminal androgen receptor (LAR), basal-like 1 (BL1), basal-like 2(BL2), immunomodulatory (IM), mesenchymal (M), mesenchymal stem-like (MSL), and unclassified (UNC). However, using RNA and DNA profiling analysis, Burstein et al (Burstein et al, 2015) distinguished 4 groups; LAR, M, basal-like immunesuppressed (BLIS) and basal-like immune-activated (BLIA) (see Table 1.2).

The LAR subtype (in both papers) is the most distinct and abundant subtype In TNBC, with estrogen/androgen pathways up-regulated compared to other TNBC subtypes. It is characterized by luminal gene expression and is driven by the androgen receptor.

M (in both papers) and MSL (Lehman et al, 2011) tumours have high levels of certain epithelial genes, particularly those involved in epithelial mesenchymal transition (EMT) and signaling pathways involved in cell motility, growth and differentiation. For example, increased levels of genes involved in the TGF- β pathway, growth factor pathways (FGF, IGF and PDGF) and Wnt pathway. A further subtype, "claudin-low", has also been identified within the M and MSL subtypes (Dias et al, 2017). The claudin-low subtype is characterized by a lack of cell-cell junction proteins and has high expression of EMT, immune response (interferon pathway) and cancer stem-like genes (Dias et al, 2017). The M and MSL groups were distinguished by the MSL group showing decreased expression of genes required for cell proliferation and increased expression of genes associated with stem cells (eg ABC genes (A8 and B1) and PROCR).

The BL1 subtype (Lehmann et al, 2011) has increased cell cycle and DNA damage response gene expression, which differs from BL2 (Lehmann et al, 2011), which has high expression of growth factor signaling components and myoepithelial genes. IM (Lehmann et al, 2011) mainly has genes involved in immune and cytokine signaling pathways as well as immune antigens. The basal like groups, defined by Lehmann et al, are comparable to the BLIS (BL1 and M) and BLIA (BL2 and IM) groups identified by Burstein et al (see Table 2).

Lehmann et al recently redefined their 6 groups to take into account transcripts from normal stromal cells and immune cells in the tumour microenvironment, thereby creating a reduced number of only 4 distinct groups (BL1, BL2, LAR and M), with cases from the previous IM and MSL groups being re-assigned between these 4 groups (Lehmann et al, 2016).

TNBC subtype	Classified by	Description
Luminal androgen	Lehman et al (2011)	Hormone receptor
receptor (LAR)	Burstein et al (2015)	pathway expression
Mesenchymal (M)	Lehman et al (2011)	High level of epithelial
	Burstein et al (2015)	genes and growth factor
		pathways
Mesenchymal	Lehman et al (2011)	Similar to M subtype with
stem-like (MSL)		high level of epithelial
		genes
		Decreased expression of
		genes required for cell
		proliferation compared to
		other TNBC subtypes
Basal-like 1 (BL-1)	Lehman et al (2011);	Increased cell cycle and
	Comparable to Basal-like	DNA damage response
	immune suppressed (BLIS)	gene expression
	in Burstein et al (2015)	
Basal-like 2 (BL-2)	Lehman et al (2011);	High growth factor
	Comparable to some	signaling and
	features of Basal-like	myoepithelial genes.
	immune-activated (BLIA) in	
	Burstein et al (2015)	
Immunomodulatory	Lehman et al (2011);	Immune and cytokine
(IM)	Comparable to some	signaling as well as
	features of BLIA in Burstein	immune antigen
	et al (2015)	presence
Unclassified (UNC)	Lehman et al (2011)	
		INBUS that do not fit into
		the other subtypes

Table 1.2: Triple negative breast cancer subtypes

A comparison of the different TNBC subtypes, according to Lehmann et al (2011) and Burstein et al (2015) classifications.

1.5 Triple negative breast cancer treatment

Unlike other breast cancer subtypes that have targeted treatments, there are currently no effective targeted treatments routinely available for primary TNBCs, and the targeted therapies used for other breast cancers are ineffective in this context because of the lack of ER and HER2 expression (Chittaranjan et al, 2014). Along with radiotherapy and the surgical options available to them, TNBC patients will therefore receive cytotoxic chemotherapy (Wahba et al, 2015).

1.5.1 Triple negative breast cancer responses to treatment and prognosis

Chemotherapy treatment can be very effective in reducing the size of the primary tumours when given in the neoadjuvant context, presumably because of the relatively high proliferation rate of TNBC (Carey et al, 2007). In the context of primary disease treated with NAC, either a decrease in tumour size (partial response) or in some cases a pathological complete response (pCR) can be seen (Rouzier et al, 2005). A pCR is determined by a lack of all morphological evidence of residual cancer. pCRs are important as patients who have achieved a pCR are more likely to have a better long-term prognosis (Cortazar et al, 2014). NAC is more effective in TNBC than any other subtype of breast cancer as TNBC cancers showed a pCR of 45% when treated with NAC in comparison to a pCR of 6% for luminal tumours (Rouzier et al, 2005). In another study carried out, TNBC patients again showed higher pCR rates (22%) compared to non-TNBC patients (11%). If pCR was achieved, TNBC patients, in this study, had similar survival to non-TNBC patients (Liedtke et al, 2008).

However, although patients may have a decrease in tumour size and partial response, more than 50% of patients will not achieve a pCR (Cortazar et al, 2014). Patients who have residual disease have worse overall survival if they had TNBC compared to non-TNBC (68% vs 88%) (Liedtke et al, 2008), which is likely due to chemoresistance in TNBC. Unfortunately, if only a partial response is seen, it is these patients who are more likely to relapse within 3 years of completion of their chemotherapy (Cortazar et al, 2014), as TNBC patients have

significantly decreased 3-year progression free survival rates when compared to non-TNBC patients (63% vs 76%) (Liedtke et al, 2008). The prognosis is also normally poor with 3-year overall survival lower than non-TNBC patients (74% vs 89%) (Liedtke et al, 2008). Treatment of the metastatic secondary tumour is always incomplete and TNBC patients can expect less than 14 months median survival (Kassam et al, 2009).

Therefore, an aim in development of neoadjuvant therapy currently is to improve pCRs of TNBC, and thereby kill any distant metastatic cells to prevent metastatic recurrences. Once a clinical metastatic reoccurrence is present, regardless of the primary response of the cancer, the prognosis is poor.

1.5.2 Current therapies for triple negative breast cancer

Chemotherapy commonly comprises of a combination of cytotoxic drugs that target proliferating cells. There are different families of chemotherapeutic agents, for example the anthracyclines, the taxanes and the alkylating agents (Isakoff et al, 2010). As determined by MRI, clinical response to neoadjuvant anthracycline treatment in TNBC cancer was 85%, compared to 70% in HER+ ER-, and 47% in luminal subtypes (Carey et al, 2007).

1.5.2.1 Mechanism of action of breast cancer cytotoxics

The anthracycline drugs that are commonly used for breast cancer treatment are doxorubicin and epirubicin (Kaklamani et al, 2003). Both doxorubicin and epirubicin target the cancer cells, along with any other actively proliferating cells, by inhibiting DNA synthesis and thereby replication, by intercalating between the base pairs in cellular DNA (Reinert, 1983). Anthracyclines can also inhibit topoisomerase activity. Topoisomerase enzymes are required for DNA strand cleavage and unwinding for efficient DNA replication to take place (Osheroff, 1989). Anthracyclines intercalate into the DNA and inhibit the DNAtopoisomerase complex after the nicking phase, and prevent the resealing stage in DNA synthesis (Han et al, 2015). Taxanes are also commonly used in breast cancer chemotherapy, with the two specific agents being paclitaxel or docetaxel (Nabholtz et al, 2005). Taxanes have an effect on the microtubules required for cell division by stabilizing GDP-tubulin, which prevents microtubule depolymerisation and subsequent chromosome separation in mitosis (Sachdev et al, 2016).

Alkylating agents, such as cyclophosphamide, are also used in breast cancer therapy and are commonly given with anthracyclines and taxanes (Fisher et al, 1990). Alkylating agents function by addition of an alkyl group to nitrogen in the purine ring of the guanine base of the DNA, forming intra-molecular DNA strand crosslinks between guanine bases to prevent the unwinding of DNA during DNA replication (Hall et al, 1992).

Multiple combinations of these drugs are commonly used for TNBC treatment. Treatment options are largely based on the NICE guidelines, the practice within the NHS trust where treatment is occurring, and the preference of individual oncologists in consultation with the patient (NICE, 2009, NICE, 2018). Drug classes anthracyclines (doxorubicin and epirubucin) and taxanes (docetaxel and paclitaxel) can be used separately or in combination with one another. In addition, cyclophosphamide can also be given in combination with either anthracyclines or taxanes. All 3 drug classes can also be combined for TNBC patients (Cleator et al, 2009).

1.5.3 Therapies under investigation for triple negative breast cancer

Because of the relatively high recurrence rate and poor prognosis, there is still an unmet clinical need for new and targeted therapies for TNBC. Poly-ADP ribose polymerase (PARP) inhibitors are one of the emerging therapies and are currently undergoing clinical trials in TNBC (Pettitt et al, 2018). PARP proteins play an important role in the repair of DNA strand breaks by being responsible for the recruitment of DNA repair enzymes (Chambon et al, 1963). The inhibitors of PARP work as cytotoxics by inhibiting the recruitment of DNA repair enzymes, leading to an accumulation of further double-stranded breaks resulting in apoptosis and cell death (Carey, 2011). PARP inhibitors look to be
promising drugs in TNBC with BRCA1 or BRCA2 mutations (Bryant et al, 2005). BRCA1/2 mutated cells are unable to form the BRCA1-associated genome surveillance complex (BASC) which therefore prevents the interaction of the BASC complex with RNA polymerase II (Wang et al, 2000, Scully et al, 1997). Subsequently, DNA strand breaks are therefore not fixed through homology directed repair. PARP inhibitors are effective in BRCA1/2 mututated tumours as the BRCA1/2 mutations render the cells uniquely dependent on DNA repair pathways that require PARP (Silver et al, 2012). PARP inhibitor, olaparib has been used in the treatment of metastatic breast cancer patients with BRCA1/2 mutations. Patients, who received olaparib, response rate was 59.9% when compared to 28.8% response rate in patients who received standard therapy. Additionally, patient progression free survival increased from 4.2 months to 7 months (Robson et al, 2017). Therefore, olaparib has recently been FDA approved for treatment of BRCA1/2 mutated metastatic breast cancers (FDA, 2018). However there does not appear to be any benefit of PARP inhibitors to other TNBC patients who do not have BRCA1/2 mutated TNBC (Tutt et al, 2010).

The addition of platinum salts to adjuvant chemotherapy has also been explored. Platinum based drugs such as cisplatin and carboplatin have shown promising responses. In primary TNBCs, cisplatin addition to docetaxel, following prior epirubicin/cyclophosphamide treatment, produced overall response rate of 77% compared to overall response rate without the addition of cisplatin to docetaxel of 70% (Alba et al, 2012). For metastatic TNBC, cisplatin addition to chemotherapy improved overall survival by 4 months (Bhattacharyya et al, 2009) and carboplatin has increased patients response rate by 11% in advanced TNBC patients and improved overall survival by 3 months (Carey et al, 2012).

PIK3CA mutations are present in 10% of TNBC and are attributed to the LAR and M subtype as described by Lehmann et al (Lehmann et al, 2011) (section 1.4.2). PIK3CA mutations lead to activation of the PI3K/Akt pathways, which can mediate many of the hallmarks of cancer (Fruman et al, 2014). In addition, expression of the tumour suppressors inositol polyphosphate 4-phosphotase type II (INPP4B) and phosphatase and tensin homolog (PTEN) are frequently lost in TNBC, also contributing to activation of the PI3K/Akt pathway (Li et al, 2017). PIK3CA mutations and loss of INPP4B and PTEN provided the impetus for the development of ipataserib, which is an ATP competitive small molecule inhibitor of Akt (Saura et al, 2017). Ipataserib is currently being used in clinical trials to determine efficacy in combination with other chemotherapeutic agents. Ipataserib clinical trials are in the early stages at phase Ib and II in patients with TNBC (Lehmann et al, 2015).

Multiple other therapeutic strategies are being explored, such as blockade of the androgen receptor in LAR TNBC (Lehmann et al, 2015), histone deactylase (HDAC) inhibitors (Uehara et al, 2012), mammalian targets of rapamycin (mTOR) inhibitors (Basho et al, 2017) and the development of immunotherapies (Stagg et al, 2013). It is hoped that blockade of AR, HDAC and mTOR inhibitors will increase the options and effectiveness of treatments available to TNBC patients.

With a lack of well-established targeted treatment available to TNBC patients, chemotherapy remains the predominant treatment for TNBC patients. Although response to chemotherapy can be promising, chemoresistance is a real clinical problem in TNBC so there is a need to understand chemoresistance pathways. Targeting cancer specific chemoresistance pathways in combination with traditional chemotherapy, could be a viable clinical route to improve survival of TNBC patients. One particular opportunity for such a combination therapy is explored in Chapter 5.

1.6 Mechanisms of chemoresistance in breast cancer

Chemotherapy remains the predominant treatment for TNBC patients and chemoresistance in TNBC, reflected in cancer recurrences and deaths, is a major clinical problem. Identification of chemoresistance pathways, and potentially inhibition of these pathways in combination with chemotherapy could represent a treatment option for TNBC patients and forms the basis of research in this thesis.

1.6.1 Physical mechanisms of chemoresistance

Extracellular matrix (ECM) proteins, particularly collagens, have been implicated in chemoresistance. Increased levels of collagen have been implicated in limited delivery and diffusion of chemotherapeutic into cancer tissues (Senthebane et al, 2017). In breast cancer cells, high levels of expression of collagen genes is associated with increased drug resistance (Iseri et al, 2010). Collagen type I has also recently shown to promote survival of breast cancer cells *in vitro* as well as enhancing metastasis (Badaoui et al, 2018). Increased levels of collagen and expression of Iysyl oxidase (LOX), a protein involved in collagen cross-linking (Herchenhan et al, 2015), can increase ECM stiffness in breast cancer. It is thought that ECM stiffness can induce chemoresistance and increase cell survival through activation of PI3K and adhesion signaling (Levental et al, 2009).

Cytoskeletal proteins have also been implicated in chemoresistance, in particular β tubulin. β tubulin III subunit overexpression has correlated with increased resistance to taxane-based chemotherapy (paclitaxel and docetaxel) in advanced breast cancer (Paradiso et al, 2005).

1.6.2 Drug inactivation mechanisms of chemoresistance

Chemoresistance can occur in epithelial cells through drug inactivation/detoxification or through increased expression of drug pumps. ABC transporters such as the P-glycoprotein pump MDR1 and breast cancer resistant protein (BCRP/ABCG2) have been shown to play a role in chemoresistance in TNBC by increasing the efflux of chemotherapeutics (Zheng, 2017). MDR1 has the ability to pump out multiple different types of chemotherapeutics in breast cancer, including paclitaxel and docetaxel (Mechetner et al, 1998). In addition, chemotherapy induced activation of Notch leads to the up-regulation of multi-drug resistance protein-1 (MRP1), with inhibition of Notch-dependent MRP1 sensitizing cells to chemotherapy (Kim et al, 2015). BCRP/ABCG2 can also pump out chemotherapeutics including anthracyclines such as doxorubicin (Doyle et al, 1998). More recently, NAC has been shown to increase expression of BCRP and has been implicated in determining diseasefree survival (Kim et al, 2013). Patients displaying low expression of BCRP, following chemotherapy, had significantly longer disease free survival in comparison to patients with high expression of BCRP (Kim et al, 2013).

Drug inactivation or detoxification is also a mechanism employed by breast cancer cells to enable chemoresistance. ALDH1A1 and ALDH3A1 have been shown to lead to inactivation of cyclophosphamide, which is commonly used in combination with anthracyclines (Sladek et al, 2002). Drug inactivation has also been attributed to altered metabolism of drugs by cytochrome P450s (CYPs). CYP2D6 is responsible for the metabolism of tamoxifen in breast cancer and therefore reduced the efficacy of tamoxifen (Ali et al, 2016). CYP3A/5 has been shown to metabolize docetaxel into less effective compounds for the treatment of breast cancer (Bedard et al 2010).

1.6.3 Cellular mechanisms of chemoresistance

Alterations in genes involved in apoptosis have been implicated in chemoresistance. In TNBC, mutations in p53 are often present and overexpression of anti-apoptotic proteins Bcl-2 is also observed (Vegran et al, 2006, Aas et al, 1996). Loss of p53 and up-regulation of Bcl-2 have been linked to chemoresistance in breast cancer cells following anthracycline chemotherapy (p53) or cyclophosphamide, epirubicin and 5-fluorouracil (Bcl-2) respectively (Vegran et al, 2006, Aas et al, 1996). Loss of the mismatch repair proteins MSH2 and MLH1 has also been linked to resistance to epirubicin and doxorubicin (Fedier et al, 2001).

1.7 Tumour stroma

The role of the tumour stroma, in particular of cancer-associated fibroblasts, in chemoresistance in TNBC is a key focus throughout this thesis. I will, therefore, provide background on what is already known about the tumour stroma and cancer-assocoiated fibroblasts as well as their role in cancer progression.

For a cancer to survive, grow and thrive it requires the right environment. The tumour stroma is responsible for this as the tumour stroma infiltrates in and around the tumour mass and provides an environmental niche for the tumour cells (Egebald et al, 2008). Tumour stroma consists of a large number of different types of stromal cells, including fibroblasts, pericytes, adipocytes and immune cells. In addition, to deliver the nutrients that the tumour requires, the tumour stroma contains blood vessels and also contains extracellular matrix proteins such as collagen, fibronectin and laminin (Kharaishvili et al, 2014). In the presence of tumour cells, components of the stroma undergo complex alterations, which alter the characteristics of the stroma. The altered characteristics of the tumour stroma are exploited by tumours to aid in cancer progression (Lu et al, 2012).

The ratio of tumour stroma to tumour cells in TNBC has been investigated to see if it correlates with prognosis of TNBC patients. TNBC patients with cancers containing greater than 50% stroma, as determined by point-counting, had greater chances of recurrences within 5 years in comparison to those with tumours that had less than 50% tumour stroma (85% vs 45%) (Moorman et al, 2012). High tumour stroma content in TNBC was also associated with poorer overall survival (89% vs 65%), however in ER positive disease, the influence on overall survival was reversed, with prolonged survival associated with stromal rich tumours (Downey et al, 2015).

The most abundant cell type within the tumour stroma is the stromal fibroblasts (Bussard et al, 2016), and fibroblasts are a key focus of my work. Fibroblasts are present normally within the body, however fibroblasts in tumour stroma show an altered phenotype and are referred to as cancer-associated fibroblasts

(CAFs) (Sadlonova et al, 2009). Investigation into the role of CAFs has increased understanding of the role that CAFs play in the progression of breast and other carcinomas. CAFs can enhance cell proliferation, angiogenesis, invasion and cancer metastasis to secondary sites (Orimo et al, 2005, Dumont et al, 2013).

1.8 Fibroblasts

Fibroblasts are non-vascular, non-epithelial and non-inflammatory cells that are commonly found with the fibrillar matrix of the connective tissue (Tarin et al, 1970). Fibroblasts remain quiescent until they are required for their functions, which include deposition of the ECM, control of epithelial cell differentiation, regulation of inflammation, and most commonly wound healing (Parsonage et al, 2005). Fibroblasts also have the ability to produce multiple types of collagen (I, III, IV and V) as well as fibronectin and laminin (Qureshi et al, 2017). Matrix metalloproteinases (MMPs) are another product that fibroblasts secrete to enable ECM degradation and cell movement through the extracellular matrix (Tomasek et al, 2002).

A key role of fibroblasts in normal physiology is in wound healing. When a wound occurs, fibroblasts are responsible for invading into the wound, producing ECM components such as collagen, fibronectin and laminin to provide a scaffold for epithelial cells to move onto and grow (Darby et al, 2014). Fibroblasts are also responsible for producing cytoskeletal components to enable contraction to pull the wound together (Gabbiani et al, 1971). As previously mentioned, fibroblasts remain quiescent until they are activated and required for their function (Gabbiani et al, 1971). When fibroblasts are in this quiescent state, they are spindle-shaped and metabolically relatively inactive, with little active secretome (Tarin et al, 1970). Fibroblasts also do not actively divide to any great extent, being commonly in G_0/G_1 arrest, although there is a slow rate of self-renewal (Marthandan et al, 2014). In addition, fibroblasts are non-migratory, and are unable to produce components of the ECM (Kalluri, 2016). Fibroblasts are activated in order to carry out their cellular function from multiple stimuli such as growth factors, chemokines, cytokines as well as

physical stress and hypoxic conditions (Bainbridge et al, 2013). Transforming growth factor- β (TGF- β) is well established as an activator of fibroblasts, particularly for fibroblast involvement in wound healing (Montesano et al, 1998). Platelet-derived growth factors are also important for both fibroblast activation and the wound healing process. Release of platelet-derived growth factor (PDGF), recruits and activates fibroblasts at the wound site for effective wound healing (Rajkumar et al, 2006). When activated, the morphology of fibroblasts changes as well as the secretory phenotype. Fibroblasts become cruciform or steallate in shape as well as becoming metabolically active, proliferative, and motile (Ravikanth et al, 2011). Once activated, fibroblasts have a dynamic secretome that includes growth factors (eg EGF, FGF, HGF) as well as the ECM components previously described. Fibroblasts may go through several cycles of active remodelling, healing or inflammatory functions, followed by successive periods of quiescence and activation depending on external stimuli (Kalluri and Zeisberg, 2006).

By contrast, CAFs are constantly active and never revert back to the quiescent inactive fibroblast state. CAFs are thought to have undergone irreversible epigenetic modifications from the normal activated fibroblast phenotype where they have acquired a higher proliferation rate and enhanced secretory phenotype (Littlepage et al, 2005). CAFs also have the ability to self-renew in comparison to the activated fibroblast phenotype as well as producing a greater amount of growth factors, cytokines and chemokines. CAFs also produce more ECM proteins, such as tenascin C (Hanamura et al, 1997), periostin (Kikuchi et al, 2008) and SPARC (Gao et al, 2017) as well as collagen (Pankova et al, 2016). Both activated fibroblasts and CAFs express enhanced levels of vimentin and α -SMA (Shiga et al, 2015). However, CAFs produce fibroblast-activated protein (FAP) (Park et al, 1999), which is not present in activated fibroblasts.

1.9 Origins of cancer-associated fibroblasts

The origins of CAFs have been debated with multiple different suggestions as to where they arise from and how they become activated.

Firstly, it is thought that the majority of CAFs are derived from the conversion of normal activated fibroblasts into CAFs, potentially driven by cancer cell secreted cytokines (Kojima et al, 2010). The mechanisms by which this conversion is induced are thought to be similar to activation of normal fibroblasts in wound healing as well as some epigenetic changes occurring (Qiu et al, 2008). TGF-β is responsible for activation of fibroblasts in wound healing but is also commonly secreted by cancer cells (Knabbe et al, 1987). Cancer cell secretion of TGF- β activates TGF- β signaling in CAFs, which in turn leads to an activated CAF phenotype (Kojima et al, 2010). Epigenetic changes are also required for CAF activation, with loss of p53 and PTEN required. Interestingly, loss of p53 and PTEN occurred in CAFs in close proximity to tumour cells indicating the potential cancer cell driven activation of CAFs (Mao et al, 2013). Direct cell contact between cancer cells and fibroblast can also lead to the activation of CAFs through adhesion molecules such as intra-cellular adhesion molecule I (ICAM1) and vascular adhesion molecule I (VCAM1) (Clayton et al, 1998). CAFs have a different miRNA expression profile in comparison to normal fibroblasts, which can also play a regulatory role in transformation of normal activated fibroblasts to CAFs. Recent studies have implicated miR-31, miR-214 and miR-155 in the transition from normal fibroblasts to CAFs (Mitra et al, 2012).

There is also evidence that bone marrow derived mesenchymal stem cells (MSCs) can differentiate into CAFs. It is thought that this differentiation could also be TGF- β driven. Characteristic CAF markers like α -SMA, FAP and tenascin-C are expressed in these MSC derived differentiated CAFs (Spaeth et al, 2009).

Finally, another mechanism by which fibroblast-like cells are produced is when malignant epithelial cells undergo epithelial mesenchymal transition (EMT).

EMT is commonly triggered by the expression of TGF- β , EGF, PDGF, FGF2 and CXC-motif chemokine ligand (CXCL) (Spaeth et al, 2009). Not only can epithelial cells transition into CAF-like cells but endothelial cells have also undergone endothelial to mesenchymal transition to produce a CAF phenotype, which has been confirmed by the expression of CAF markers (Zeisberg et al, 2007).

Despite all these potential origins of CAFs, the precise origin of CAFs in breast cancer is not fully understood and it is plausible that it is a combination of CAFs from multiple origins that comprise the tumour mass.

1.10 Cancer associated fibroblasts in breast cancer progression

CAFs have been implicated in multiple aspects of cancer progression including tumour growth, angiogenesis and metastasis.

In terms of tumour growth, CAFs can secrete the growth factors EGF, HGF and FGF, which can activate growth factor signaling in the cancer cells and thereby enhance tumour growth (Matsumoto et al, 2006). Other novel secretory factors from CAFs have been implicated in tumour growth and survival. Secreted frizzled related protein 1 and IGF like family member (IGF) have also been shown to directly or indirectly enhance tumour growth (Ostman et al, 2009). Stromal derived factor-1 (SDF-1) can also contribute to tumour growth and can also have an impact on migration and invasion to secondary sites (Orimo et al, 2005). SDF-1 binds to its receptor CXCR4 on the surface of breast cancer cells to activate signaling pathways to enhance proliferation of the breast cancer cells (Orimo et al, 2005).

CAFs can also aid in migration, invasion and metastasis (Verghese et al, 2013). Interleukin-6 (IL-6) has been shown to promote migration in ER positive breast cancer and IL-6 also has the ability to induce EMT in ER positive breast cancer cells (Hugo et al, 2012). One of the main classes of secretory factors that CAFs release that aid in migration, invasion and metastasis are MMPs (Singer et al, 2002). MMPs degrade ECM proteins such as collagen and laminin and remodel the ECM to allow cancer cells to migrate through the stroma and surrounding tissue to aid cells in reaching the lymph vessel and blood vessels to travel to secondary sites (Gianneli et al, 1997). MMP2, MMP9 and MMP14 have been shown to have increase migration and invasion in breast cancer (Kessenbrock et al, 2010, Rothschild et al, 2015).

Another area CAFs can aid in is angiogenesis. CAFs can secrete VEGF, which upon binding to its receptor, activates VEGF signaling in endothelial cells, which in turn promotes the formation of new blood vessels to deliver oxygen and nutrients to the tumour to aid in its growth and survival (Bachelder et al, 2002).

The roles of CAFs, in breast cancer, are best understood in growth, migration, invasion and angiogenesis, however there is still a lack of understanding of how CAFs contribute to other hallmarks of cancer such as evading growth suppression, immune destruction and resisting cell death. In addition, there are large gaps in knowledge of whether and how CAFs contribute to chemoresistance.

1.11 The interferon family

Interferons, their signaling pathways, and their roles in cancer progression will be highlighted as it became evident in my results that interferon signaling could play a role in chemoresistance. The interferon (IFN) family is a group of secreted cytokines that are widely expressed, predominantly by immune cells, and are typically produced in response to viral infection to induce potent antiviral and anti-proliferative effects (Le Page et al, 2000).

There are multiple different interferon paralogues, each expressed from its own gene. These can be categorised into 3 different classes: Type I, Type II and Type III (Pestka et al, 1987) (summarized in Table 1.3). Within Type I, there are nine different paralogues including IFN- α (13 known isoforms), IFN- β , IFN- ϵ , IFN- κ , IFN- ω , IFN- δ , IFN- τ , IFN- ν and IFN- ζ (Pestka et al, 2004). Interestingly, the genes for Type I interferons are found within a single cluster on one chromosome (9p21) (Diaz et al, 1994). IFN- γ is the only representative of Type

II interferon (Pestka et al, 1987). IFN- λ 1, IFN- λ 2 and IFN- λ 3 form the Type III paralogues (Ank et al, 2006). Each class of interferon binds to its own subset of receptors. Type I interferons bind to the heterodimeric receptor complex of IFN- α receptor 1 (IFNAR1) and IFNAR2 subunits (Mogensen et al, 1999). Type II interferons bind to the IFN- γ receptor IFNGR1 and IFNGR2 complex, which is a tetramer of 2 molecules of each of the two subunits (Pestka et al, 1997). Finally, the Type III class binds to the IFN- λ receptor 1 (IFNLR1) (Kotenko et al, 2003)

1.12 Interferon biogenesis

Type I IFNs are produced by a number of cell types in response to various stimuli (Gibbert et al, 2013), whereas expression of Type II and III IFNs are far more limited and only commonly expressed in immune cells, normally T cells and NK cells (Samuel et al, 2001).

Pattern-associated molecular patterns (PAMP), from viruses or other foreign bodies, are recognized by cells, which leads to interferon production (Baig et al, 2008). PAMPs then activate pattern recognition receptors (PRRs) (Medzhitov et al, 1997). The most predominant PRR are the Toll-like receptors (TLR). TLR1, TLR2, TLR4, TLR5 and TLR6 can react to bacterial lipopolysaccharides, whereas TLR3, TLR7, TLR8 and TLR9 have the ability to react to nucleic acids (Uematsu et al, 2007). MYD88, MAL, TRIF and TRAM are adaptors of response and are recruited to the TLRs when TLRs become activated (O'Neill et al, 2007). Recruitment of adaptors to the TLRs leads to the activation of TANKbinding kinase 1 (TBK) and IKK ϵ and IkB (Ma et al, 2012), which then bind and activate transcription factors; IFN response factors (IRFs) and nuclear factor-kB (NF-kB) (Ma et al, 2012, Pomerantz et al,1999). IRFs and NF-kB transcription factors upregulate the expression of IFNs (Randall and Goodbourn, 2008).

1.13 Interferon signaling

Following interferon production, interferons carry out the late phase viral response through activation of the interferon signaling pathway using the classical JAK-STAT pathway (Silvennoinen et al, 1993). The different interferon

/interferon receptor pathways also lead to varying activation of the classical JAK-STAT pathway (summarized in Table 1.3). Activation of the JAK-STAT pathway occurs relatively quickly as there is no need to synthesize new protein for the signaling cascade to proceed (Pestka, 2007).

	Type I	Type II	Type III
	Interferon	Interferon	Interferon
Interferons	IFN-α (13	IFN-γ	IFN-λ1, IFN-λ2, IFN-
	subtypes), IFN-		λ3
	β, IFN-ε, IFN-κ,		
	IFN-ω, IFN-δ,		
	IFN-т, IFN-v		
	and IFN-ζ		
Receptor	IFNAR1 and	IFNGR1 and	IFNLR1 and IL10R2
	IFNAR2	IFNGR2	
STATs	STAT1, STAT2,	STAT1	STAT1, STAT2,
	STAT3 and		STAT3, STAT4 and
	STAT5		STAT5
	STAT4 and		
	STAT6 - IFN-α		
	only		
	Must form		
	ISGF3 complex		
	with IRF9 for		
	signaling		
Promoter	ISRE and GAS	GAS	ISRE and GAS
element			

Table 1.3: Classes of interferons

A comparison of the 3 different types of interferon, their receptors, the STATS used for signal transduction and the promoter elements (IFN-stimulated response element (ISRE) and IFN- γ - activated site (GAS)) the STATs are able to bind to for transcription of interferon-stimulated genes (ISGs)

1.13.1 Type I interferon signaling

Upon binding of the Type I IFNs to the IFNAR1 and IFNAR2 receptor complex, ligand-dependent rearrangement and dimerization of the receptor occurs (Pattyn et al, 1999). Receptor dimerization leads to trans-phosphorylation events of the receptor associated JAK kinases, TYK2 and JAK. Activation of TYK2 and JAK enables further phosphorylation events and phosphorylation of STATs (Darnell et al, 1994) (Figure 1.2). There are multiple forms of STATs that can be phosphorylated in the Type I signaling pathway which include STAT1, STAT2, STAT3 and STAT5. In addition, IFN-α can also activate STAT4 and STAT6. All of these STAT subunits are able to form homodimers and heterodimers with one another upon phosphorylation by TYK2 and JAK (Darnell et al, 1997). STAT1 and STAT3 are known to be phosphorylated on serine 727 (Wen et al, 1995), however the location of phosphorylation of the other STATs has not been explored. STAT homo or heterodimers can translocate to the nucleus and act as transcription factors, binding at specific binding sites in target promoters (Meinke et al, 1996). There are 2 distinct canonical promoter elements that STAT dimers bind to: either the IFN-stimulated response element (ISRE) or IFN-y- activated site (GAS) element (Yang et al, 2005). Most of the STAT homo and heterodimers have the ability to bind to the GAS element and enable transcription. However, only STAT1-STAT2 heterodimers have the ability to bind to ISREs in combination with IRF9, as the ISG Factor 3 (ISGF3) complex (Fu et al, 1992). IRF9 does not require phosphorylation to bind to the STAT1-STAT2 heterodimer. The ISGF3 complex then translocates to the nucleus to bind to the ISRE and enable transcription of interferon stimulated genes (ISGs) (Tsuno et al, 2009).

1.13.2 Type II interferon signaling

IFN-γ is the only Type II interferon and upon binding to its receptors, IFNGR1 and IFNGR2, IFN-γ also activates the JAK-STAT pathway shown in Figure 1.2. However, in this case the 2 receptor subunits, IFNGR1 and IFNGR2, are each constitutively associated with JAK1 and JAK2 respectively (Bach et al, 1997). Activation of JAK1 and JAK2 leads to the phosphorylation of STAT1 on tyrosine 701, which enables the formation of a STAT1 homodimer. The STAT1 homodimer then translocates to the nucleus and binds to the GAS elements within target promoters to permit transcription of ISGs (de Weerd and Nguyen, 2012).



Figure 1.2: Type I and II Interferon signaling

The interferon signaling pathway is represented schematically from extracellular at the top, down to intra-nuclear at the bottom, with the plasma membrane (blue double-barrier) and nuclear membrane (non-continuous blue line). Purple elements are specific to Type II with red components specific with Type I, although cross-talk with Type II should be noted.

1.13.3 Regulation of interferon signaling

Interferon signaling is tightly regulated, which may be important for overall organism homeostasis since, for example, prolonged signaling has been associated with autoimmune diseases (Feng et al, 2012). Part of interferon regulation involves negative feedback, as some ISGs that are up-regulated by interferon signaling act to inhibit the JAK-STAT pathway. The 2 main examples

of this are USP18 and SOCS (Ivashkiv et al, 2014). USP18 is an isopeptidase that is normally responsible for the degradation of proteins (Honke et al, 2016). However, this role of USP18 is not responsible for the inhibition of the interferon signaling pathway. USP18 specifically binds to the IFNAR2 subunit, thereby preventing the binding of JAK1 to IFNAR2 and subsequently inhibiting JAK-STAT activation and downstream interferon signaling (Sarasin-Filipowicz et al, 2009). The family of SOCS proteins are also inhibitors of the JAK-STAT pathway, particularly SOCS1 and 3. SOCS proteins bind to phosphorylated residues on the IFN receptors or on JAK components, and prevent subsequent binding of STATs and thereby inhibit activation of the signaling pathway (Yoshimura et al, 2007).

1.13.4 Interferon stimulated genes

Transcription of many different genes is stimulated upon binding of STATs to either the ISRE or GAS elements of their promoters, which are known as interferon stimaulted genes (ISGs).

One family of genes that is up-regulated by IFN signaling is the oligoadenylate synthase (OAS) family. This comprises OAS1, OAS2 and OAS3 (Rebouillat et al, 1999). The OAS family stimulates cleavage of double stranded RNA structures. OAS enzymes convert adenosine triphosphate (ATP) into 2'-5'-linked oligoadenylate (2-5A) (Kerr et al, 1978). 2-5A in turn promotes dimerization and activation of the latent endoribonuclease, RNaseL (Zhou et al, 1993). Activated RNaseL is then responsible for degradation of double-stranded viral and cellular RNA, as well as promoting apoptosis, attenuating proliferation and inhibiting protein synthesis (Sadler et al, 2008).

Another family of proteins that are prominent ISGs is the myxovirus resistance (MX) family (Horisberger et al, 1991). MX family is made up of MX1 and MX2, and the proteins are responsible for the inhibition of virus entry into cells. MX1 and MX2 are structurally similar to dynamin-like large guanosine triphosphatase (GTPases) (Haller et al, 2007). MX1 and MX2 have N-terminal GTPase domains, bundle signaling elements and C-terminal stalk domains. Both N-

terminal and C-terminal domains are critical for the function of MX1 and MX2 as they enable self-oligomerisation for the formation of multiple ring like structures (Gao et al, 2011). It is thought that these ring like structures interact with and surround viral nucleocaspids, which activates the GTPase activity of the MX proteins. Activation of GTPase activity leads to the degradation of the viral nucleocaspids, however this mechanism has yet to be fully determined (Haller et al, 2011).

Both the OAS and MX family of proteins are classical ISGs that are activated upon interferon signaling and are involved in viral response. OAS1 and MX1 have been used in my work to confirm interferon signaling activation.

1.14 Interferons in cancer

In addition to interferons role in anti-viral response, interferons have also been implicated in cancer. It is thought that they can play a role in extrinsic and intrinsic mechanisms of cancer cells. Extrinsic mechanisms have an impact on other cell types that are not directly the tumour cell (section 1.14.1 and Figure 1.3), whereas intrinsic mechanisms impact the tumour cells directly (section 1.14.2 and Figure 1.4) (Platanais et al, 2013).

1.14.1 Extrinsic roles of interferons in cancer

Interferons can modulate immunoregulatory effects. Interferons have been shown to regulate the activity of a number of different classes of immune cells and can potentially activate anti-tumour immunity. Activation of natural killer cells (NK) cells and CD8-positive cytotoxic T cells by interferons can reduce tumour burden and enable anti-metastatic effects to prevent cancer spread (Swann et al, 2007).

Interferons can also suppress proliferation of effector T cells preventing immune response against cancer cells. The presence of interferons inhibits the ability of T-regulatory cells (Tregs) to divide, thereby reducing their suppression of cytotoxic T cell activity (Pace et al, 2010). Interferons can also have an impact

on myeloid-derived suppressor cells (MDSC) by reducing their local levels, and therefore reducing MDSC subsequent suppressive function on immune responses (Zoglmeier et al, 2011). Production of interferons can also influence the production of further chemokines required for immune responses against cancer cells, such as IL-15. IL-15 is responsible for activation and proliferation of T and NK cells as well as providing survival signals for memory T cells to maintain the lymphocyte population (Munger et al, 1995).

Interferon driven activation of dendritic cells (DC) can also increase presentation of tumour antigens to T cells as well as enhancing the expression of tumour antigens on the surface of cells (Schiavoni et al, 2013). Interferons can also increase the expression of programmed cell death protein 1 (PD1) and its ligand PD-L1 on immune cells and tumour cells respectively. Upon binding of PD1 to its ligand PD-L1, T-cell receptor signaling is inhibited and subsequently T cell proliferation is stalled. It is currently thought that up-regulation of PD-L1 and PD1 enable evasion of immune responses (Parker et al, 2016).



Figure 1.3: Extrinsic roles of interferons on immune-regulation in cancer

1.14.2 Intrinsic roles of interferons in cancer

Interferons can also have an impact on the tumour cells themselves (Figure 1.4) and have been implicated in many core biological functions. Interferons have the ability to prolong and stall the cell cycle. Interferons have been shown to delay entry into S phase and G2/mitosis by lengthening the time cells spend in G1, S and G2./M, therefore inhibiting proliferative ability of tumour cells. In addition, quiescent cells in G_0/G_1 were delayed entry into the cell cycle in the presence of interferons (Balkwill et al, 1978). Interferons have also been shown to activate another pathway by activating CRK proteins. The CRK protein CRKL interacts with RAP1A, which has previously been shown to be a tumour suppressor of RAS GTPases, which are predominantly activated in cancer and activate multiple signaling pathways (Fathers et al, 2012). Presence of both Type I and Type II interferons has been shown to block cell cycle progression through up-regulation of the cyclin-dependent kinase inhibitor (CDK1) p21. p21 is able to block cell cycle progression by binding directly to cyclin-dependent kinase 2-cyclin complexes (cyclin A and E), which are required for passage through G1 and the S phase to G2/M transition (Hobeika et al, 1997).

Interferons can also have an impact on apoptosis of tumour cells. IFN- α has been shown to induce apoptosis of cells. In order to do this, IFN- α activated multiple different caspases (-1,-2,-3, -8 and -9). Caspases are required for apoptosis, with caspase-8 and -9 being the major initiator caspases (McIlwain et al, 2013). Normally caspase-8 is activated by death ligands binding to their receptors and caspase-9 is activated by mitochondrial cytochrome c. However, IFN- α has been shown to activate casapase-8 and -9 in the absence of death ligands and thereby induce apoptosis (Thyrell et al, 2002).



Figure 1.4: Intrinsic roles of interferons on tumour cells

1.15 MiRNAs

In my results, I have attempted to identify miRNAs with potential roles in chemoresistance, therefore, here I describe miRNA biogenesis, the function of miRNA in gene silencing and the relevance of their expression in cancer.

MiRNAs (miRNAs) are a subset of the transcriptome, being short non-coding RNAs of approximately 20-22 nucleotides in length, with 5'- phosphate and 3' hydroxyl ends (Krol et al, 2004). MiRNA genes are found in multiple different regions of the genome, including in the introns of coding genes as well as in inter-genic regions. MiRNA genes are also commonly found in clusters within both these regions (Lagos-Quintana et al, 2001). It is thought that transcription of miRNAs within a gene is driven by the same promoter that is responsible for the transcription of the host gene. However, transcription of miRNAs present in inter-genic regions is driven by their own promoters (Bartel et al, 2004).

1.16 MiRNA biogenesis

A number of modification and processing stages are required to convert the initial transcript of a miRNA into a mature miRNA (summarized in Figure 1.5). Transcription of miRNAs normally occurs using RNA polymerase II (Lee at al, 2004).

Initial miRNA transcripts that are produced following gene transcription are referred to as primary miRNAs, or pri-miRNAs. Pri-miRNAs have a specific structure as they have 2 segments of single stranded RNA, that are complimentary to one another, which link together by a hairpin loop and 3 spiral turns (Cai et al, 2004). In addition, the pri-miRNA must contain a 5' end cap and a poly A tail (Cai et al, 2004). If the pri-miRNA does not contain the hairpin loop and 5' and 3' modifications, the RNA will not be processed correctly for the formation of mature functional miRNA (MacFarlane and Murphy, 2010).

Nuclear processing of the pri-miRNA is then undertaken. Nuclear protein DiGeorge syndrome critical region gene 8 (DGSCR8) recognizes the double stranded hairpin loops in the pri-miRNA and binds to the double-stranded RNA (Han et al, 2004). DGSCR8 associates with the RNA III endonuclease Drosha to form the microprocessor complex (Denli et al, 2004). Drosha then cuts around 11 nucleotides from the pri-miRNA to produce an isolated hairpin species, referred to as the pre-miRNA (Lee et al, 2003).

MiRNA transcripts can also be produced from an intron in a host-coding gene. The production of these miRNAs is initially different from the transcripts described above. Intronic miRNAs are again transcribed by RNA polymerase II when the host-coding gene is transcribed (Kim et al, 2007). This therefore produces precursor mRNA (pre-mRNA). Pre-mRNA contains the 5' UTR, the host gene protein-coding region, non-coding introns (in which the intronic miRNA is present) and the 3'UTR (Lin et al, 2006). The pre-mRNA then must undergo splicing by the spliceosome to remove the intron region from the host gene-coding region to produce the intron containing miRNA transcript and the mRNA for protein translation (Ruby et al, 2007). Further splicing and processing enables the spliced intron to function as the pri-miRNA. Following production of the pri-miRNA, the processing of the intronic and inter-genic miRNA is exactly the same (Shomron et al, 2009).

For further processing, the pre-miRNA hairpins must be exported to the cytoplasm (Lund et al, 2004). For this to occur the pre-miRNA assembles into a complex with the nucleo-cytoplasmic transporter factor exportin-5. Exportin-5 recognises a 2 nucleotide overhang on the 3' end of the hairpin (Bohnsack et al, 2004). For exportin-5 mediated transport across the nuclear barrier, GTP is required, which is bound to a Ran protein. The Ran-GTP is converted to Ran-GDP to actively transport the pre-miRNA to the cytoplasm (Yi et al, 2003).

For the production of mature miRNA, the pre-miRNA is cleaved by the RNase III endonuclease, Dicer (Bernstein et al, 2001). Dicer forms a complex with argonaute 2 protein (Ago2) and trans-activating response RNA binding protein (TRBP) (Chendrimada et al, 2005). Dicer interacts with the 5' and 3' ends of the pre-miRNA hairpin and cuts the long flexible loop that joins the 5' and 3' arms of the hairpin to create an imperfect miRNA: miRNA* duplex (Zhang et al, 2002).

Unwinding of the miRNA: miRNA* duplex is then required, which is performed by the Dicer, TRBP and Ago2 complex, to produce mature miRNA (Kawamata et al, 2009). The single-stranded mature miRNA that is bound to the Dicer, TRBP and Ago2 complex is then incorporated into the RNA-induced silencing complex (RISC) for miRNA silencing (Provost et al, 2002).



Figure 1.5: MiRNA biogenesis

A schematic flow chart of miRNA biogenesis starting from miRNA transcribed from within its own miRNA gene or intronic miRNA gene, and the processing and export steps until the miRNA binds to its target for gene silencing.

1.17 MiRNA silencing of RNA

The major role of miRNAs is in RNA silencing and post-transcriptional regulation of gene expression. MiRNAs have the ability to impair the production of proteins from target mRNAs and thereby reduce expression of particular mRNA targets (Figure 1.6) (Shivdasani et al, 2006). It is thought that 60% of all mRNAs are targets for regulation by miRNAs (Friedman et al, 2009). For RNA silencing to occur, the formation of the RISC complex is vital as the RISC complex guides the miRNA towards its correct target mRNA (Pratt et al, 2009).

The core components of the RISC complex are Dicer, Ago2, TRBP and protein kinase R-activating protein (PACT) (Lee et al, 2006). Once the mature miRNA has been formed, the mature miRNA is loaded into the RISC complex that then guides the miRNA to target sequences in mRNAs, normally in the 3' UTRs (Chendrimada et al, 2005).

MiRNA driven mRNA silencing can occur in different ways. Firstly, miRNA can be responsible for the cleavage of the mRNA strand and subsequent degradation (MacFarlane et al, 2010). Here the miRNA, that has been guided to the target mRNA by the RISC complex, is usually – but not always - fully complimentary and base pairs with the mRNA (Meister et al, 2004). Cleavage of the mRNA strand is then catalyzed by Ago2 in the RISC complex leading to degradation of the RNA (Liu et al, 2004)

Secondly, miRNAs can be responsible for inhibition of mRNA translation (Valencia-Sanchez et al, 2006). Inhibition of mRNA translation can occur when nucleotides 2-7 of the miRNA, which is known as the seed region, are complementary to the mRNA while the remainder of the miRNA is only partially complementary and the base pairing is mismatched (Meister et al, 2004). It is thought that complimentary and mismatched base pairing creates bulges in the miRNA and mRNA pairing to prevent translation occurring. However, further understanding is required (Bartel et al, 2004). Complimentary and mismatched base pairing between the miRNA and mRNA can also lead to an increase in the speed of deadenylation and removal of the mRNA poly A tail leading to more rapid mRNA degradation (Pillai, 2005).

1.18 MiRNAs in cancer

Cancer is driven by mutations and dysregulation of genes, so, given the roles miRNAs play in gene regulation in normal cells, the roles miRNAs have in gene dysregulation in cancer have been studied intensively (Peng et al, 2016).

Firstly, expression profiles of miRNAs have been assessed to identify any increased or decreased expression of miRNAs in cancers (Espinosa et al,

2006). MiRNA expression profiles are commonly carried out using sequencing platforms (miRNA-SEQ) to screen a large number of miRNAs, with a large number of changes seen in miRNA expression in many different tumour types (Zhu et al, 2014). For example, miR-155 has been shown to overexpressed in many types of cancer such as haematopoietic cancers, breast, lung and colon (Faraoni et al, 2009). In addition, miR-21 has been found to be overexpressed in almost all cancers studied such as breast, colon, liver, brain, pancreas and prostate (Volinia et al., 2006). The miR-17-92 cluster, which encodes 6 different miRNAs, is overexpressed in multiple cancers such as lung, colon and gastric cancers (Concepcion et al, 2012).

It is evident that the expression of miRNAs change during malignant transformation and further work has been carried out to identify what role miRNAs have in cancer progression (Adams et al, 2014). Because of miRNAs role in dysregulation of gene expression, miRNAs can act as either oncomiRs or tumour suppressor miRs. MiRNAs act as oncomiRs when they aid in cancer progression and promote tumor development, normally by targeting inhibitory regulators of cancer as well as genes that stimulate cell differentiation and apoptosis (Jansson et al, 2012). On the other hand, tumour suppressor miRs prevent cancer progression by reducing expression of oncogenes (Price et al, 2014).

There are a large number of examples of oncomiRs in a variety of different cancers that can play a role in different hallmarks of cancer. miR-21 is responsible for the down-regulation of four tumour suppressor genes: mapsin, programmed cell death 4 (PDCD4), tropomyosin1 (TPM1) and phosphatase and tensin homolog (PTEN) (Volinia et al., 2006). Additionally, it is thought that PTEN is also a target of the miR-17-92 cluster (Zhu et al, 2014).

The miR-17-92 cluster contains the genes for the following six miRNAs: miR-17, miR-18a, miR-19a, miR-19b-1, miR-20a and miR-92-1. The miR-17-92 cluster has also been shown to be an oncomiR (Mendell, 2008). The miR-17-92 cluster has been linked to the expression of c-myc, which is required for proliferation. Both c-myc and the miR-17-92 cluster regulate the expression of the

transcription factor E2F1 leading to increased cell survival (O'Donnell et al, 2005). There are also numerous examples of miRNAs acting as tumour suppressors. For example, let-7, which is one of the first-identified members of the miRNA family, has been shown to negatively regulate the oncogene RAS (Johnson et al, 2005).

1.19 Hypothesis and aims

Having reviewed the current literature, my hypothesis was that CAFs enable chemoresistance in TNBC through the release of specific mediators that act on epithelial cancer cells.

I aimed to:

- Identify whether CAFs enable chemoresistance in TNBC cells;
- Characterize potential mediators (miRNA) and signaling pathways responsible for CAF-induced chemoresistance;
- Determine if the CAF-induced chemoresistance pathways could be inhibited to improve TNBC response to chemotherapy;
- Highlight any potential cellular mechanisms responsible for CAF-induced chemoresistance.

Chapter 2: Methods and materials

2.1 Cell lines and tissue culture

Triple negative (ER, PR and HER2 negative) breast epithelial cancer cell lines MDA-MB-231 (wild-type [without either luciferase or GFP expression]), MDA-MB-468 and MDA-MB-157 cells were acquired from American Type Culture Collection (ATCC HTB-26, HTB-132 and HTB-24, Manassas, USA). MDA-MB-231 cells that stably express firefly luciferase (MDA-MB-231-luc) were purchased (Cat# AKR-231, Cell Biolabs, San Diego, USA), and MDA-MB-231 cells that stably express GFP and firefly luciferase (MDA-MB-231-GFP/luc) were developed in-house by transduction with lentiviruses coding for overexpression of the appropriate reading frames (Lorger et al, 2010). MDA-MB-468 that stably express firefly luciferase (MDA-MB-468-luc) or GFP (MDA-MB-468-GFP) were also developed in house following transduction with lentiviruses (Lorger et al, 2010). Normal breast fibroblasts (NFs) or breast cancerassociated fibroblasts (CAFs) were extracted from breast cancer patient samples by the Hughes group and immortalised by viral transduction to allow stable over-expression of hTERT (Verghese et al, 2013) to create the lines NF1 and NF2 (normal fibroblasts) or CAF1 (cancer-associated fibroblasts). CAF1-GFP cells were also developed by transduction with lentiviruses coding for overexpression of the appropriate reading frame (Verghese et al, 2013). Primary NF (pNF) and CAF (pCAF) were also extracted from breast cancer patient samples and were used at relatively early passage (between 5 and 10 passages). Both the pNFs and pCAFs were taken from a TNBC patient with the pCAFs being taken from within the tumour mass, and the pNFs being extracted from at least 1cm outside the tumour margin. Fibroblasts were extracted from samples provided by breast cancer patients undergoing treatment within the Leeds Teaching Hospitals NHS Trust; patients were recruited and informed consent was taken under ethical permission from the Leeds (East) Research Ethics Committee reference 09/H1326/108.

MDA-MB-231 (all derivatives), MDA-MB-468 (all derivatives), NF1, NF2, CAF1 and CAF1-GFP cells were all grown in the same media: DMEM (Cat# 31966-

021, Thermofisher Scientific, Waltham, USA), 10% FCS (Cat# F7524, Sigma, St Louis, USA) and 1% penicillin/streptomycin at final concentrations of 100 U/ml penicillin and 100µg/ml Streptomycin (Cat# 15070-063, Thermofisher Scientific, Waltham, USA). Cells were cultured at 37°C in 5% CO₂/air and were kept subconfluent by sub-culturing every 2-4 days as appropriate. MDA-MB-157 cells were grown in Leibovitz L-15 media (Cat# 11415-049, Thermofisher Scientific, Waltham, USA) with 10% FCS and 1% penicillin/streptomycin as above, and were cultured at 37[°]C in sealed flasks or plates in 100% air. MDA-MB-157 cells were kept sub-confluent by sub-culturing every 7 days, with 2 media changes during this period. pNFs and pCAFs were cultured in DMEM F12 media (Cat# 31331-028, Thermofisher Scientific, Waltham, USA) supplemented with 10% FBS, 1% penicillin/streptomycin as above, and 1% Fungizone at a final concentration of 5µg/ml (Cat# A2942, Sigma, St Louis, USA). Cells were cultured at 37° C in 5% CO₂/air and were kept sub-confluent by sub-culturing every 2-4 days as appropriate. All cells were maintained routinely in 150cm² tissue culture flasks (Cat# CLS430825, Corning, Sigma, St Louis, USA).

For the passaging of cells, medium was removed and cells were washed with 10ml Dulbecco's phosphate-buffered saline (DPBS, Cat# 14190-094, Thermofisher Scientific, Waltham, USA). 2ml of 0.05% trypsin (Cat#1540054, Thermofisher Scientific, Waltham, USA) was added and incubated for 5min at 37^oC, or until all the cells had dissociated from the plastic. To inactivate the trypsin, 10ml of the correct culture media was added to the cells. 2-10ml of the cell suspension was then added to 20ml of fresh culture media in a new 150cm² tissue culture flask. The volume of cell suspension varied dependent on the cell line used, with – for example - MDA-MB-157 cells and primary fibroblasts requiring higher seeding densities, therefore 10ml of cell suspension was added to the 20ml fresh culture media.

To remove cells from plastic mid-experiment for replating or for further downstream analysis, cells were trypsinised using 2ml of 0.05% trypsin and resuspended in 10ml cell culture media. Cells were then centrifuged at 400g for 5 min to pellet the cells. Cells were then resuspended in a further 10ml of culture media ready for replating. For down-stream analysis, following pelleting, cells were resuspend in 10ml PBS instead.

2.2 Transfection of cell lines

ISRE/GAS reporter plasmids and renilla plasmid (pRL-TK using HSV thymidine kinase promoter (Mankouri et al, 2010)) were kind gifts from Dr Andrew Macdonald (University of Leeds). MiRNA mimics and inhibitors and appropriate controls were purchased from Dharmacon (Lafayette, USA), and are listed in Table 2.1.

MDA-MB-231 cells (all derivatives) were seeded for transfection in 24 well plates (Cat# CLS3527, Corning Costar, Sigma, St Louis, USA) at 50,000 cells per well, or in 96-well plates (Cat# CLS9102, Corning Costar, Sigma, St Louis, USA) at 20,000 cells per well and were incubated in standard culture for 24h before transfection. MDA-MB-157 cells were seeded for transfection in 96-well plates at 40,000 cells per well 24h prior to transfection. Cells were then transfected in OPTI-MEM (Cat# 31985-070, Thermofisher Scientific, Waltham, USA) without serum, with varying concentrations of miRNA inhibitors/mimics or plasmids as appropriate, using Lipofectamine 2000 (Cat#11668019, Thermofisher Scientific, Waltham, USA). Prior to addition to wells, miRNA mimics or inhibitors, or plasmids, were combined with OPTI-MEM and Lipofectamine 2000 for 20min at room temperature; 2µl of Lipofectamine 2000 was used per 24-well in 100µl of OPTI-MEM or 0.5µl of Lipofectamine 2000 was used per 96-well in 50µl of OPTI-MEM. Transfection mixtures were added to cells, with medium already on the cells, for 6h and cells were incubated under standard culture conditions. After this 6h, medium was then replaced with full fresh medium and cells were incubated in standard culture conditions at least overnight before further manipulation (for example, trypsinisation for re-plating in co-culture experiments). MiRNA mimics or inhibitors and their associated controls were used at concentrations of between 500pM to 500nM in the transfection mix (mimics/inhibitors and controls at the same concentration in each individual experiment), while ISRE/GAS reporter plasmids were used at $1\mu g/ml.$

Target	Dharmacon catalogue number	
Control mimic	CN-001000-01-05	
Control inhibitor	IN-001005-01-05	
miR 27a-3p mimic	C-300502-03-0002	
miR 27a-3p inhibitor	IH-300502-05-0002	
miR 155-5p mimic	C-300647-05-0002	
miR 155-5p inhibitor	IH-300647-06-0002	

Table 2.1: MiRNA mimics and inhibitors used for transfection

2.3 Chemotherapy agents, chemical inhibitors, recombinant interferons, blocking antibodies

Epirubicin hydrochloride (Cat# E9406, Sigma, St Louis, USA) was dissolved in DMSO at a stock concentration of 100mM, and was used at a wide range of doses, depending on cell line and end-point assay, (short-term survival or colony forming assays) of 10nM to 79µM. Docetaxel (Cat#01885-5MG-F, Sigma, St Louia, USA) was dissolved in DMSO at a stock concentration of 10mM, and was used at 350pM for colony forming assays. Ruxolitinib (Cat#S1378, Selleckchem, Munich, Germany) was dissolved in DMSO at a stock of 10mM and was used at 10nM for colony forming assays. Recombinant IFN- α and IFN- γ were kind gifts from Matthew Holmes and Fiona Errington-Mais (University of Leeds) and were manufactured by Peprotech (Cat # 300-02A and 300-02 respectively, Rocky Hill, USA). Anti-Human Interferon Alpha/Beta receptor chain 2 (clone MMHAR-2, Cat# 21385-1, PBL Assay Science, Piscataway, USA) or anti-human Interferon gamma R1 neutralising antibodies (CD119, Cat# AF673, R&D Systems, Minneapolis, USA) were added to cultures with a final concentration of 1µg/ml or 5µg/ml respectively. Appropriate isotype controls for both the IFN- α and IFN- γ blocking antibodies were used respectively: mouse IgG_{2A} isotype control (Cat#MAB003, R&D Systems, Minneapolis, USA) and goat IgG control (Cat# AB-108-C, R&D Systems, Minneapolis, USA) at the same concentrations.

2.4 Dual luciferase reporter assays

MDA-MB-231 cells that had been transfected with the ISRE or GAS reporter plasmids were then trypsinised and re-plated in culture/co-culture with varying proportions of NF1 or CAF1 cells at 0, 8, 20 or 55% fibroblasts within a total of 20,000 cells per well in 96-well plates. Cultures were incubated with or without epirubicin as described in the appropriate figure legend. Cells were then washed twice with PBS and lysed using 50µl of 1x passive lysis buffer per well (Cat# E194A, Promega, Madison, USA) and left on a rocker for 15min at room temp. Analysis of luciferase activity followed the protocol in section 2.5.1, however *renilla* activity was also measured with 20µl Stop and Glo reagent (Cat#E640A and E641A, Promega, Madison, USA) added to each well after the LARII (Cat# E151A and E195A, Promega, Madison, USA) reading was taken. The plate was shaken for 2s, delay 2s and read for 5s.

2.5 Luciferase-based chemo-survival assays

2.5.1 Dose determination

For simple mono-culture experiments, MDA-MB-231-luc or MDA-MB-468-luc were plated in 96-well plates at 20,000 cells per well and were cultured for 24h as normal. Cells were then treated in triplicate with epirubicin hydrochloride at doses ranging from 1pM to 500µM, or without epirubicin treatment (control), by replacing the media with 50µl of dosed media. Following culture for 24h, cells were washed twice with PBS and lysed using 100µl of 1x passive lysis buffer (Cat# E194A, Promega, Madison USA). Plates were left on a rocker for 15min at room temperature for the cells to lyse. 10µl of lysate was added in triplicate to wells of a new 96-well opaque wall plate, and were analysed for luciferase activity using the Berthold technologies Mithras LB 940 (Bad Wildbad, Germany) 96-well plate reader. The following luminescence protocol was used: 20µl LARII (Cat# E151A and E195A, Promega, Madison, USA) was added to each well, shaken for 2s, delay 2s and read for 5s. For calculating cell survival,

mean treated readings were expressed relative to the no treatment (control) average. IC25, 50 and IC75 doses were estimated from dose response curves.

2.5.2 Co-culture assays

For co-culture experiments, MDA-MB-231-luc or MDA-MB-468-luc cells were plated in 96-well plates in triplicate with varying proportions of NF1, NF2, or CAF1 cells (from 0% fibroblasts to 55% fibroblasts) within a total of 20,000 cells per well. After 24h of culture, cells were treated with approximate IC10, 50 or 75 concentrations of epirubicin hydrochloride (as determined by mono-culture dose-response assays (see section 2.5.1), or with vehicle only, for a further 24h. For 24h treatments only, medium was then removed and cells washed twice with PBS and lysed in 1x passive lysis buffer (Cat# E194A, Promega, Madison, USA) as above. For 72h incubations, after 24h of drug treatment, medium was replaced with drug-free medium for a further 48h before washing and lysis as above. Luciferase activity within lysates was assessed using the Berthold plate reader as above. Readings from epirubicin-treated cultures were normalised to readings from matched cultures with the same fibroblast proportion and treated with vehicle control. Normalisation this way allowed focus on differences in response to epirubicin, and to avoid the confounding factors of differences in the total numbers of epithelial cells seeded and the potential influences of fibroblasts on epithelial growth.

2.6 MTT assays

Cells were seeded/transfected in 96-well plates as appropriate for individual experiments. 25µl of 5mg/ml MTT 3-(4,5-dimethylthiazol-2-yl)-2,5diphenyltetrazolium bromide (Cat# M6494, Thermofisher Scientific, Waltham, USA) was added to each well for 4h in the dark at 37^oC in 5% CO₂/air, and then removed. 50µl of propan-1-ol (Cat# 6775-25, Macron fine chemicals, Center Valley, USA) was added to each well and plates were rocked until all the precipitate had dissolved. Absorbances were then measured at 570nm using the Berthold 96-well plate reader.

2.7 Mono-culture colony forming assays

For initial chemotherapy dose-finding experiments and for recombinant interferon colony formation experiments, mono-culture colony forming assays were performed. Cells (MDA-MB-231-GFP/luc, MDA-MB-468-GFP, or MDA-MB-157) were seeded in 6-well plates at 200,000 cells per well and allowed to adhere overnight. Cells were treated with chemotherapy agents (see section 2.3) or recombinant interferons as appropriate for up to 48h. The final concentrations of IFN- α ranged from 500-5000pg/ml and IFN- γ ranged from 500-3000pg/ml. Cells were then trypsinised, counted using a haemocytometer and plated in duplicate 10cm tissue culture dishes (Cat# CLS430167, Corning, Sigma, St Louis, USA) at 500 cells per plate in fresh medium lacking chemotherapeutics or recombinant interferons. Plates were incubated for 14 days undisturbed (in an isolated incubator dedicated to these assays with minimal opening/closing – critical to minimise cells from individual colonies becoming dispersed into multiple colonies). For MDA-MB-231-GFP/luc and MDA-MB-468-GFP cells, cells were then fixed/stained using 1ml of 5mg/ml Crystal Violet (Cat# V5265, Sigma, St Louis, USA) in 50% methanol, 20% ethanol and 30% water for 30s, followed by 3 washes in distilled water. For MDA-MB-157 cells, cells were fixed/stained twice in succession by two full repeats of 2ml of 12.5mg/ml Crystal Violet in 50% methanol, 20% ethanol and 30% water, for 5min and followed by 3 washes in distilled water. Isolated colonies estimated to be of more than 40 cells were counted manually.

The reproducibility and objectivity of these manual colony counts was confirmed as follows. The author and a fellow researcher (Melina Teske, University of Leeds) independently counted colonies on 16 plates, representing a range of different number of colonies. The two independent values for the number of colonies on each plate were compared, and the correlation between the sets of data was calculated (Figure 2.1). The correlation coefficient of 0.949 indicates near perfect agreement between the independent scores, demonstrating excellent reproducibility of these data.



Figure 2.1: Comparison of 2 independent counts of colony forming assay plates of MDA-MB-468 cells

Colonies on 16 separate plates of MDA-MB-468 cells with differing numbers of colonies following recombinant interferon and/or epirubicin treatments were counted independently by the author, RB (count 1) or a colleague, Melina Teske (count 2). Counts for each plate were plotted independently (left panel) or relative to each other (right panel).

2.8 Co-culture colony forming assays

For co-culture experiments, MDA-MB-231-GFP/luc, MDA-MB-468-GFP, or MDA-MB-157 cells were seeded with varying proportions of NF1, CAF1, CAF1-GFP, pNF or pCAF cells at 0, 8, 20 or 55% fibroblasts within the same total number of cells (1x10⁶ cells in a T25 flask [Cat# CLS430639, Corning, Sigma, St Louis, USA] or 400,000 in each well for 6-well plates). MDA-MB-231-GFP/luc and MDA-MB-468-GFP were co-cultured with NF1 or CAF1 cells (which are GFP negative), but the GFP negative MDA-MB-157 cells were cultured with CAF1-GFP (GFP positive). Epithelial cells may have previously been transiently transfected immediately prior to seeding for these co-cultures, as previously described in section 2.2. Cultures were treated with a variety of chemotherapy agents, chemical inhibitors, or antibodies as appropriate (see section 2.3). After up to 48h, cultures (including the 0% fibroblast culture, which is an epithelial mono-culture) were treated for fluorescence-activated cell sorting (FACS), and epithelial cells were re-plated to form colonies as described for section 2.7.

FACS was used to separate components of cultures for subsequent downstream analyses of the separate cell types (by colony forming assay, or qPCR for example). Importantly, mono-cultures were also sorted in experiments where mono-cultures were compared with co-cultures, to ensure that the cells were treated similarly in both cases. Cultures were trypsinised and washed twice with PBS on ice, and resuspended in 1ml RPMI phenol red free media (Cat# 11835-030, Gibco, Thermofisher Scientific, Waltham USA). Cells were then strained using a 70µM nylon cell strainer (Cat#352350, Corning, Sigma, St Louis, USA) before sorting. The BD Influx 6-way cell sorter was used, identifying GFP positive cells using the 488nm blue laser, and gating on live cells using FSC/SSC. Only single cells were collected. Representative flow plots are shown for the cell combinations separated: MDA-MB-231-GFP/luc cells combined with fibroblasts (GFP negative), MDA-MB-468-GFP combined with fibroblasts (GFP negative), and CAF1-GFP cells combined with MDA-MB-157 cells (Figures 2.2, 2.3 and 2.4 respectively). Up to 100,000 cells of the desired population were collected on ice for re-plating (section 2.7) or other down-stream analysis (section 2.12).



Figure 2.2: FACS cell sorting plots for MDA-MB-231 GFP/luc cells co-cultured with 20% CAF1 cells

Viable cells were first gated (FSC vs SSC left panel) followed by gating on GFP expression: GFP negative CAF1 cells (red peak in right panel) and GFP positive MDA-MB-231 cells (green peak in right panel). Cells were sorted from the appropriate peaks on the right plot.



Figure 2.3: FACS cell sorting plots for MDA-MB-468 GFP cells co-cultured with 20% CAF1 cells

Viable cells were first gated (FSC vs SSC left panel) followed by gating on GFP expression: GFP negative CAF1 cells (red peak in right panel) and GFP positive MDA-MB-231 cells (green peak in right panel). Cells were sorted from the appropriate gated peak on the right plot.



Figure 2.4: FACS cell sorting plots for MDA-MB-157 cells co-cultured with 20% CAF1-GFP cells

Viable cells were first gated (FSC vs SSC left panel) followed by gating on GFP expression: GFP negative MDA-MB-157 cells (red peak in right panel) and GFP positive CAF1-GFP (green peak in right panel). Cells were sorted from the appropriate gated peak on the right plot.
2.9 Preparation of fibroblast conditioned media, and use in colony forming chemo-survival assays

In order to prepare fibroblast-conditioned media (with and without epirubicin), T150 tissue culture flasks of NF1 and CAF1 cells were seeded so as to be just sub-confluent. Flask were treated with 10nM epirubucin hydrochloride for 24h (induced conditioned media) or left without drug (conditioned media). Medium was collected from these flasks, centrifuged at 400g for 5min, and the supernatent was used on epithelial cells immediately without storage, meaning that multiple flasks of fibroblasts had to be established and treated at different times in order to supply fresh conditioned media when required. Epithelial cells (MDA-MB-231-GFP/luc, MDA-MB-157 or MDA-MB-468-GFP) were seeded for 8h in standard culture conditions at 1x10⁶ cells in T25 tissue culture flasks in either normal fresh medium or a mix of 50% conditioned medium/50% normal fresh medium using conditioned medium from either NF1 or CAF1 cells. Medium was then replaced with fresh medium or 50% conditioned medium as appropriate and cells were cultured for a further 16h. Cells were then treated with 10nM epirubicin hydrochloride in either 0% or 50% conditioned media from NF1 or CAF1 cells or 50% induced conditioned media from NF1 or CAF1 cells. Cells were left for 8h at 37°C 5% CO₂ with media again replaced for a further 16h. Cells were then trypsinised, counted and re-plated to allow colonies to form exactly as described in section 2.7.

2.10 Anoikis assays

MDA-MB-231-GFP/luc were seeded with NF1 or CAF1 cells in T75 flasks (Cat# CLS430641, Corning, Sigma, St Louis, USA) with varying proportions of fibroblasts, including 0, 8, 20 or 55% fibroblasts at a total cell number per flask of 2x10⁶. Cultures were incubated for 24h, followed by a media change and a further incubation for 24h. Cultures were then trypsinised, washed twice with ice cold PBS and GFP-positive MDA-MB-231-GFP/luc cells were collected by FACS exactly as described in section 2.8; 1.5x10⁵ MDA-MB-231-GFP/luc cells were collected. The remainder of the protocol followed the CytoSelect 96-well anoikis assay manufacturer's protocol (Cat#CBA-081, Cell Biolabs, San Diego, USA). All MDA-MB-231-GFP/luc cells collected from FACS were spun at 400g for 5min in a centrifuge and resuspended in 100µl of media and added to 96well anchorage resistant plates (part 108101 from kit named above). Cells were incubated for 36h under normal culture conditions. In order to quantify surviving cells, 10µl of the MTT reagent (part 113502), was added to each well for 3h, and the plate was incubated as for normal culture. Following this incubation, 100µl of detergent solution (part 108004) was added to each well and plates were incubated for 3h at room temperature in the dark. The 150µl of solution in the wells was transferred to a 96-well tissue culture plate and the absorbance of each well was measured at 570nm on the Mithras LB 940 96-well plate reader (Berthold Technologies).

2.11 Analysis of transcriptomes for MDA-MB-231 cells cultured with and without CAF1 cells

MDA-MB-231-GFP/luc cells were seeded alone (0% fibroblasts) or with 20% CAF1 cells in T75 tissue culture flasks, giving a total cell number of 2x10⁶ in both cases. Cultures were incubated (24h), treated with 10nM epirubicin (24h), and then treated for FACS to purify GFP-positive MDA-MB-231-GFP/luc cells as described above (section 2.8). 900,000 MDA-MB-231-GFP/luc cells were collected. Sorted MDA-MB-231-GFP/luc cells were then collected by centrifugation at 400g for 5min, washed with PBS and collected again by centrifugation at 400g for 5min. RNA was extracted using the ReliaPrep RNA miniprep system (Cat#Z6012, Promega, Madison, USA) following the manufacturer's protocol, and suspended in 30µl of nuclease free water (part#P119E). RNA concentrations were measured using the Nanodrop (ND-1000 spectrophotometer). Three separate biological repeats of this experiment were completed in a three week period to provide robust biological replicates. The same RNA samples were used for transcriptome analyses of both miRNAs and mRNAs (section 2.11.1 and 2.11.2).

2.11.1 mRNA transcriptome analyses

Samples were sent to the Sheffield Microarray/Genomics core facility at the University of Sheffield to be analysed on Affymetrix Clariom D microarrays

(Santa Clara, USA) by Dr Paul Heath (University of Sheffield). Initial data quality examination was then performed using Affymetrix Expression Console software and probe level summarisation files were then used for further analysis using the Affymetrix transcriptome analysis console v3.0. Using the transcriptome analysis console, the three biological repeats were grouped into 0% CAF and 20% CAF samples and a paired ANOVA was performed for each individual mRNA probe. Thresholds were put in place where fold changes between the 0% and 20% groups were greater than +/- 2 with p value from the ANOVA less than 0.05. The list of genes that fit these criteria were then analysed in ToppGene (https://toppgene.cchmc.org, Chen et al, 2009) to identify pathways of interest. For ToppGene software, the ToppFun programme was used and the full gene list was added with the entry type Ensembl ID.

2.11.2 miRNA analyses

The three separate samples representing independent biological replicates for 0% CAF or 20% CAF were pooled for each group. These pools were diluted to equal RNA concentrations (as determined by Nanodrop) of 136ng/µl. Expression of 377 miRNAs was determined using TaqMan low-density arrays: human miRNA array A (Thermofisher Scientific, Waltham, USA). RNA was reverse transcribed using MegaPlex RT pool A primers using the TaqMan miRNA reverse transcription kit (Cat# 4366597, Thermofisher Scientific, Waltham, USA): the RT reaction mix was made up of 0.8µl of MegaPlex RT pool A primer, 0.2µl of 100mM dNTPs, 1.5µl of 50U/µl Multiscribe Reverse transcriptase, 0.8µl 10X RT buffer, 0.9µl of 25nM MgCl₂, 0.1µl of 20U/µl RNase Inhibitor, 0.2µl of nuclease free water per sample. 4.5µl of the RT reaction mix was mixed with 3µl of RNA per sample and incubated on ice for 5min. The tubes were then placed on the Dyad Peltier Thermocycler (Bio-rad, Hercules, USA) following the thermal-cycling conditions shown in Table 2.2.

Stage	Temp	Time
Cycle (40 Cycles)	16°C	2min
	42°C	1min
	50°C	1sec
Hold	85°C	5min
Hold	4°C	∞

Table 2.2: Thermal cycling protocol required for Taqman MegaPlex miRNA cards reverse transcription

Following the RT, 6µl of the RT product was combined with 450µl of Taqman Universal PCR master mix (Cat# 4304437, Thermofisher Sceintific, Waltham, USA) and 444µl of nuclease free water per sample. 100µl of PCR reaction Mix was added into each port of a Taqman human miRNA array A plate and reactions were performed on the 7900 HT real time PCR machine using a standard curve (AQ) and 384 well TaqMan Low density array. These data were normalised in two alternative ways: to RNU48, a snoRNA commonly used as an invariant normaliser (Gee et al, 2011), or to the array mean including every CT value less than 40 (Park et al, 2003). Undetermined CTs were set to 40 to allow an estimation of minimum fold change when compared to the detected values. Fold changes were determined using δδct method (Livak et al, 2001).

2.12 RNA extractions and qPCR

RNA was extracted using the ReliaPrep RNA miniprep kit (Promega, Madison, USA) following the manufacturer's protocol.

2.12.1 miRNA qPCR

TaqMan miRNA assays were used. A minimum of 50ng of RNA was used per sample, with equal amounts of RNA used for all samples in the same analysis. RNA volumes were made up to 10µl using nuclease free water (giving 5µl each for detection of target and normaliser). Reverse transcription reactions were carried out in a total volume of 15µl per sample comprising of 5µl RNA, 3µl target or control primer and 7µl RT master mix. The RT master mix, using the

TaqMan miRNA reverse transcription kit (Cat#4366596, Thermofisher Scientific, Waltham, USA), was comprised of 0.15µl of 100mM dNTPs with dTTP, 1µl of 50U/µl Multiscribe Reverse transcriptase, 1.5µl 10X RT buffer, 0.19µl of 20U/µl RNase Inhibitor, 4.16µl of nuclease free water per sample. Samples were incubated on ice for 5min and the tubes were placed on the Dyad peltier thermocycler using the thermal-cycling conditions shown in Table 2.3.

Stage	Temp	Time
Hold	16°C	30min
Hold	42°C	30min
Hold	85°C	5min
Hold	4°C	∞

Table 2.3: Thermal cycling protocol required for TaqMan miRNA assays reverse transcription

For real-time PCRs, total reaction volumes were 10µl. This was made up of 0.5µl target or control primer, 0.67µl RT product (which had been diluted 1:3 with nuclease free water), 5µl TagMan 2x Universal PCR master mix (Cat# 4304437, Thermofisher Scientific, Waltham, USA) and 3.83µl of nuclease free water. This was added to a microAmp optical 96 well reaction plates (Cat # 4316813, Thermofisher Scientific, Waltham, USA) and qPCR analysis was performed using the QuantStudio 5 system on the TagMan chemistry setting. Target primers included miR-27a-5p (Cat# 4427975 ID 002445, Thermofisher Scientific, Waltham, USA), miR-27a-3p (Cat# 4427975 ID 000408, Thermofisher Scientific, Waltham, USA), miR-32-5p (Cat# 4427975 ID 002109, Thermofisher Scientific, Waltham, USA), miR-155-5p (Cat# 4427975 ID 002623, Thermofisher Scientific, Waltham, USA), miR-422a (Cat# 4427975 ID 002297, Thermofisher Scientific, Waltham, USA) and miR-454-3p (Cat# 4427975 ID 002323, Thermofisher Scientific, Waltham, USA). For each individual analysis, RNU48 (Cat# 4427975 ID 001006, Thermofisher Scientific, Waltham, USA) was performed in parallel and was used as the normaliser; relative expression levels of target miRNAs were calculated by the $\delta\delta$ ct method (Livak et al, 2001).

2.12.2 mRNA qPCR

The Promega GoTaq 2-Step RT-qPCR system (Cat# A5000, Promega, Madison, USA) was used. A minimum of 50ng in 4µl per sample was used. 1µl of random primers was added. The RNA/primer mix was preheated to 70°C for 5min then immediately chilled to 4°C for at least 5min. The reverse transcription reaction mix (Promega, Madison, USA) was then added to the RNA primer mix, with the following volumes per sample; 5µl nuclease free water, 4µl GoScript 5x reaction buffer, 4µl of 25M MgCl₂, 1µl PCR nucleotide mix for final concentration of 0.5mM and 1µl GoScript reverse transcriptase. The tubes were then placed on the Dyad peltier thermocycler and incubated at the thermal cycling temperatures shown in Table 2.4.

Stage	Temp	Time
Hold	25°C	5min
Hold	42°C	1hr
Hold	70°C	15min
Hold	4°C	∞

<u>Table 2.4:</u> Thermal cycling protocol required for SYBR green mRNA assays reverse transcription

The resulting cDNA was diluted with 40µl nuclease free water prior to real-time PCR. For the real-time PCR, 5µl of the GoTaq qPCR master mix was used per sample with 0.2µl of 40X target primer and 0.05µl CXR reference dye used per sample. Target primers used were OAS1 (Cat# 74007036, IDT, Coralville, USA), MX1 (Cat# 74007039, IDT, Coralville, USA), IFNA2 (Cat# 74849839, IDT, Coralville, USA), IFNB1 (Cat# 74849836, IDT, Coralville, USA), and IFNG (Cat#74849833, IDT, Coralville, USA), with ACTB (Cat# 74007033, IDT, Coralville, USA) used as a control normaliser (see Table 2.5 for primer sequences). Master mix and primer solution were combined with 5µl of cDNA per well in a microAmp optical 96 well reaction plate. qPCR was performed using the QuantStudio 5 system using the pre-programmed SYBR chemistry setting. Expression was determined relative to ACTB using the $\delta\delta$ ct method (Livak et al, 2001).

Target	Primer
OAS1	5'-GATGAGCTTGACATAGATTTGGG-3'
	5'-GGTGGAGTTCGATGTGCTG-3'
MY1	5'-CGAAACATCTGTGAAAGCAAGC-3'
	5'-CAGGCTTTGTGAATTACAGGAC-3'
IFNA2	5'-TTGACTTGCAGCTGAGCA-3'
	5'-CCCATTTCAACCAGTCTAGCA-3'
IFNB1	5'-GCCATCAGTCACTTAAACAGC-3'
	5'-GAAACTGAAGATCTCCTAGCCT-3'
IFNG	5'-CGACAGTTCAGCCATCACTT-3'
	5'-GCAACAAAAAGAAACGAGATGAC-3'
АСТВ	5'-CCTTGCACATGCCGGAG-3'
	5'-ACAGAGCCTCGCCTTTG-3'

Table 2.5: Primer sequences of target and control mRNA qPCR primers

2.13 Flow cytometric assessment of epirubicin loading or cell cycle

To assess epirubicin loading, cells (mono-cultures or co-cultures) were prepared for flow-cytometry by being removed from the plastic by trypsin, and being resuspended in PBS (see section 2.1). The Attune flow cytometer (Thermofisher, Waltham, USA) was used. The BL-1 channel was used to detect GFP expression, while the BL-3 channel was used to detect epirubicin. Analysis was gated on live cells on the basis of FSC/SSC, and then on either the GFPpositive MDA-MB-231-GFP/luc cells or the GFP-negative fibroblasts (representative flow-cytomtery dot plots are shown in Figure 2.5). Epirubicin loading was quantified as BL-3 median level from a minimum of 10,000 events. The specific cell populations are in the following sections; R2- GFP negative fibroblasts loaded with epirubicin, R3- GFP positive MDA-MB-231 cells loaded with epirubcin, R4- GFP negative fibroblasts not loaded with epirubicin, R5-GFP positive MDA-MB-231 cells not loaded with epirubicin. The following voltages for each channel and compensation matrix were used (Table 2.6).

	FSC	SSC	BL1	BL3
Voltage	2700	3400	1200	1400
Compensa	ation	BI 1	BL3	
matrix		BEI		
BL1		100	-1.5	
BL3		-3	100	





Figure 2.5: Flow cytometry gating strategy for epirubicin loading assessments MDA-MB-231-GFP/luc were co-cultured with CAF1 cells and treated with epirubicin. First, a gate was determined to select viable cells using FSC vs SSC (top left panel). Then cells were analysed based on BL-1 (GFP positive, x-axis) and BL-3 (epirubicin, y-axis) with the dot plot (top right panel) split into quarters. R2 represents GFP negative epirubicin loaded cells and R3 represents GFP positive epirubicin loaded cells. The median absorbance for BL-3 (epirubicin) in the separate quadrants was determined as shown in the table. For cell cycle analysis, cells (mono-cultures or co-cultures) were prepared for flow-cytometry by being removed from the plastic by trypsin, and being resuspended in 1ml complete medium with 1µl of Dye Cycle violet (Cat# V35003, Thermofisher Scientific, Waltham, USA). Samples were incubated for at least 30min at 37^oC in the dark. Samples were then analysed on the Attune flow cytometer. Analysis was gated on Dye cycle violet viable cells, on the basis of FSC/VL1 and single cells, on the basis of VL1-H vs VL1-A. GFP positive (MDA-MB-231-GFP/luc) or negative (CAF1 cells) were gated separately using VL1 channel vs BL1 channel. Finally, cell cycle profiles from the Dye cycle violet were extracted using VL1-H. For the analysis, the following compensation and settings were used.

	FSC	SSC	BL1	VL1
Voltage	2700	3600	1200	1300
Comper mat	nsation rix	BL1		VL1
BL	.1	100		-0.1
VL	.1	-0.6		100

Table 2.7: Voltages for channels and compensation matrix used for cell cycle analysis

Following the formation of the cell cycle profile, these data were analysed using ModFit LT (Verity Software) to determine the percentage of cells that were in each stage of the cell cycle (G1, S, or G2/M).





Chapter 3- Cancer-associated fibroblasts protect some triple negative breast cancer cell types from chemotherapeutics

3.1 Abstract

Cytotoxic chemotherapy is the only routine systemic treatment for primary triple negative breast cancer (TNBC). Resistance to chemotherapy in TNBC is a clinical problem reflected in the relatively high rate of early recurrences. Cancer-associated fibroblasts (CAFs) are the most abundant cell type in breast tumour stroma, and have been shown to influence behaviour of tumour cells. My aims were to determine whether the presence or amount of CAFs or normal fibroblasts (NFs) modifies responses of TNBC cells to chemotherapy.

I have shown that the short-term survival of MDA-MB-231 cells after treatment with the anthracycline chemotherapeutic epirubicin was significantly enhanced by the presence of immortalized human breast CAFs, and that this protection was dose-dependent in terms of the proportion of CAFs in the culture. This survival advantage was not observed in MDA-MB-468 cells, which represent a different sub-class of TNBC – the claudin-high cancers. Using clonogenic survival as the end point, claudin-low TNBC cell lines MDA-MB-231 or MDA-MB-157 cells were both significantly protected from epirubicin-induced death by immortalized CAFs, while claudin-high MDA-MB-468 cells were again not protected. Immortalised breast NFs did not protect any cell lines from the effects of epirubicin in this assay. Clonogenic assays were also performed using MDA-MB-231 cells in combination with a matched pair of primary breast CAFs or NFs. As previously, CAFs protected the cancer cells from epirubicin, while NFs did not. Immortalised CAFs also protected MDA-MB-231 cells from the taxane chemotherapeutic docetaxel. Interestingly, conditioned media from CAFs did not offer significant protection from epirubicin in MDA-MB-231, MDA-MB-157 or MDA-MB-468 cells, suggesting that physical contact or bi-directional signaling between the CAFs and epithelial cells is required to induce protection.

I concluded that CAFs, but not NFs, were capable of protecting some types of TNBC cells from chemotherapy, and that the mechanisms involved were worthy of further study.

3.2 Introduction

For TNBC patients, there are no routinely used targeted treatments currently available, so chemotherapy is the only systemic therapeutic option (Chittaranjan et al, 2014). Chemotherapy can be given prior to surgery (neoadjuvant) or post-surgery (adjuvant), with TNBC patients showing relatively good responses in comparison to other breast cancer groups in the neoadjuvant setting (Rouzier et al, 2005). Up to 50% of TNBC patients can achieve pCR, however, responses in the remainder of patients are highly variable. Some patients' tumours will decrease in size but will not achieve a pathological complete response, while others may even continue to grow. It is these patients, those that do not respond to chemotherapy, that are more likely to have a poorer prognosis, with relapses common within 1-3 years post-treatment (Cortazar et al, 2014). Therefore, chemoresistance in these patients is a clinical problem and an understanding of chemoresistance would be useful in order to design improved therapies, as there are currently limited alternative treatment options available.

Tumour stroma is usually defined as comprising the non-cancer cell components of tumours, such as pericytes, adipocytes, immune cells, blood and lymph vessel cells, and fibroblasts, along with the extracellular matrix. Typically, the most abundant cell type present in the tumour stroma is the cancer-associated fibroblasts (CAFs) (Madar et al, 2013). However, the number of fibroblasts present can vary greatly from patient to patient. Although studies focusing solely on fibroblasts are lacking, it has previously been shown that TNBC patients with a higher proportion of stroma (>50%) have a poorer prognosis than patients with a lower proportion of stroma (<50%) (Moorman et al, 2012), with – surprisingly - opposite results reported in ER positive breast cancers (Downey et al, 2014). This suggests that stromal elements may have roles to play in defining treatment responses.

CAFs have been shown to have influences on tumour progression, including at the specific levels of tumour growth, angiogenesis and metastasis (Gascard et al, 2016). In addition, it has been reported that CAFs may have a protective role for the epithelial cancer cells that enables drug resistance, through remodeling of collagen I within the extra-cellular matrix thereby providing a physical barrier for chemotherapeutic perfusion of the cancer cells (Netti et al, 2000). In addition to structural changes, secretory factors released by CAFs to induce drug resistance in breast cancer have also been explored. For example, CAFs have been shown to be responsible for de-sensitization of ER positive breast cancer to the anti-estrogen fulvestrant (Leyh et al, 2015). In TNBC, CAFs have been shown to release to HGF, which activates Met signaling and can induce resistance to EGFR inhibitors (Mueller et al, 2012). However, there is no firm consensus concerning the main pathways by which CAFs influence therapy-response of breast cancer cells, particularly within the TNBC subtype and with respect to cytotoxic chemotherapeutics.

In this chapter, I have explored whether the presence and differing amounts of breast fibroblasts (NFs or CAFs) have an impact on chemoresistance in TNBC cells. The chemotherapeutic agent that I have used mainly is the anthracycline epirubicin. Epirubicin is routinely used in treatment of TNBC, usually in combination with other chemotherapeutic agents, such as cyclophosphamide, and sometimes in sequence with taxanes. My aim was to provide insights into whether different fibroblast proportions confer changes in chemo-response and therefore, potentially, outcomes after chemotherapy.

3.3 Results

3.3.1 Breast fibroblasts can protect TNBC cell lines from the chemotherapeutic epirubicin

My first aim was to determine whether the presence of normal breast fibroblasts, or breast CAFs was able to influence the immediate sensitivity of TNBC epithelial cells to cytotoxic chemotherapy. Therefore, I developed a shortterm, co-culture survival assay where luciferase expressing TNBC cells (MDA-MB-231-GFP/luc or MDA-MB-468-luc cells) were co-cultured with either immortalised breast NFs or CAFs in varying proportions, ranging from a majority of fibroblasts in the culture (seeded as 55% fibroblasts: 45% epithelial cells) through to a small minority of fibroblasts (seeded as 8% fibroblasts) or only epithelial cells (0% fibroblasts), with the same total cell number seeded in each case in order to achieve comparable overall confluencies. Two different breast NF lines were used (NF1 and NF2), while only one CAF line was used (CAF1). Cultures were treated for 24h with the chemotherapeutic epirubicin at 3 different doses approximating to the IC10, IC50 and IC75 doses, which had previously been estimated for each epithelial cell line when treated alone, or were treated with vehicle control (Appendix Figure 1). Survival of epithelial cells only was assessed using luciferase assays and is expressed relative to the untreated cultures of the same fibroblast: epithelial ratio (Figure 3.1). This normalisation method enabled a comparison of epithelial survival following treatment in the cultures of differing ratios of cells, and excluded the differences in actual numbers of epithelial cells plated to create these differing ratios.

Epirubicin dose-dependently reduced survival of the epithelial cells, which was evident at the higher doses of epirubicin, although less consistently for the lower doses. In the case of MDA-MB-231 cells and MDA-MB-468 (Figure 3.1A and 3.1B), the introduction of increasing proportions of breast NFs (NF1 or NF2) or CAFs did not have any signifcant impact on the survival of the epithelial cells, with no significant differences observed between the 0% fibroblast cultures and any other proportion. Although not significant (p values 0.571, 0.881 and 0.956), CAF1 cells appeared to provide some protection to MDA-MB-231 against all epirubicin doses at the lower proportions of CAF1 cells (4%,8% and 20%). It is also interesting to note that at some doses, the ability of the epithelial cells to survive in the presence of CAF1 cells increased above the level of the matched untreated culture (ie above 1 on the y-axis). Therefore, CAF1 cells appeared to enable an epirubicin-dependent growth stimulatory effect.

Next, I modified this assay to be more sensitive to the toxic effects of epirubicin that might not be evident immediately after such a short treatment with the drug. In this modified assay, following the 24h treatment with epirubicin, drug was removed and cells were cultured for an additional 48h before luciferase measurement (Figure 3.2). My hypothesis was that this extended culture time might allow cells that were severely damaged but still assessed as alive after 24h to break-down fully, and also might allow more time for fibroblast-dependent protection.

As expected, the same doses of epirubicin proved to be more toxic to both epithelial cell lines after this extended incubation (compare the positions on the y-axis in Figure 3.2 to Figure 3.1). More interestingly, significant fibroblast-dependent protection of epithelial cells was seen. This was most striking for MDA-MB-231 cells with CAF1 cells (Figure 3.2A, p<0.01), and from CAF1s with MDA-MD-468 cells (Figure 3.2B, p<0.01). In the case of CAF1 cells and MDA-MB-231 cells, 55% fibroblasts provided *complete* protection from the effects of the lowest dose of epirubicin.



Figure 3.1: CAF1s enhance chemo-survival in MDA-MB-231 cells with various proportions of fibroblasts after 24h epirubicin treatment

MDA-MB-231-GFP/luc (A) or MDA-MB-468-luc (B) luciferase-positive cells were plated in mono-culture, or in co-culture with two alternative immortalised normal breast fibroblast lines (NF1 or NF2) or the immortalised breast CAF line (CAF1) in the proportions shown. Cultures were treated with their respective estimated IC10, 50 and 75 epirubicin doses for 24h. Cells were lysed, and luciferase assays were performed. Values were normalised to untreated cultures of the same fibroblast proportion. Data represent the means (+/- SE) of 3 independent experimental repeats except for MDA-MB-468 with NF (2 replicates only), while each independent repeat itself comprised 3 replicate wells. Linear regression analysis was carried out with significant differences in the overall trend across the fibroblast proportions for the appropriate doses calculated. No significant differences were observed.





3.3.2 Immortalised CAF1s protect claudin-low TNBC cells (MDA-MB-157 and MDA-MB-231), but not claudin-high MDA-MB-468 cells from epirubicin

Having identified that breast fibroblasts can enhance the survival of TNBC cells using a short-term assay, my next aim was to extend this observation to examine the influence of fibroblasts on epithelial survival using an alternative assay. This was done using clonogenic survival assays, which are more sensitive to lesser degrees of chemotherapy-induced damage, as for cells to count as "having survived" they must be capable of repeated cell divisions.

A similar principle was used as previously for the short-term chemo-survival assays, whereby triple negative cancer cell lines (MDA-MB-231-GFP/luc or MDA-MB-468-GFP) were cultured alone or with increasing proportions of NF1 or CAF1 cells. Cultures were established, and epirubicin or vehicle control was added. GFP-positive epithelial cells were separated from fibroblasts using fluorescence activated cell sorting (FACS) and re-plated to assess clonogenic potential. Cultures without fibroblasts (i.e. 0% fibroblast / 100% epithelial) were also sorted to control for any FACS-induced technical artifacts and to allow accurate comparisons. 500 cells were plated, cultured for 2 weeks, fixed and stained before individual colonies were counted (Figure 3.3). Data are expressed as numbers of colonies (Figure 3.3A and B "Colony counts"), and as colony numbers after epirubicin treatment *relative* to matched cultures that were not treated with epirubicin (Figure 3.3 A and B "Relative to no chemo"), to allow focus on the influence of fibroblasts on epithelial survival after chemotherapy treatment.

Firstly, an unexpected observation was noted in the colony number data in the absence of epirubicin treatment, which was not the initial aim of this set of experiments. Although not significant (p value 0.087), clonogenic plating efficiency of MDA-MB-231 cells was improved as the proportion of CAF1 cells increased, while - interestingly - NF1 cells significantly decreased clonogenicity of MDA-MB-231 cells (Fig 3.3A; 'colony counts' compare pink and orange bars p<0.05). CAF1 cells did not confer this increased clonogenic ability on MDA-MB-468 cells (Figure 3.3B 'colony counts').

Next, as expected, epirubicin reduced clonogenic survival by between 48 and 52% in both MDA-MB-231 and MDA-MB-468 cells in the absence of fibroblasts (Fig 3.3A and 3.3B 'relative to no chemo', 0% bars: note position on y axis). CAF1 cells, but not NF1 cells, protected MDA-MB-231s from this epirubicinmediated impairment to colony formation in a dose-dependent manner (Fig 3.3A purple bars; p<0.05). However, CAFs did not exert protection on MDA-MB-468 cells (Figure 3.3B, dark blue bars). At the highest proportion of CAF1s (55%), MDA-MB-231 cells were very substantially protected from epirubicin; survival was 83% relative to the MDA-MB-231 cells cultured with 55% CAF1s that were not treated with epirubicin.

It is important to note that although the MDA-MB-231 and MDA-MB-468 cell lines are both classed as triple negative cell lines, they can be sub-classified into claudin-low (MDA-MB-231) and claudin high (MDA-MB-468) (Ricardo et al, 2012, Dias et al, 2017), and these sub-classes may respond and act differently (see section 1.4.2 for an introduction to sub-classes of TNBCs). In order to examine whether the effect of CAF1s on the MDA-MB-231 cells was potentially related to the claudin-low phenotype, the experiment was repeated with a second claudin-low TNBC cell line, MDA-MB-157. Some methodological changes were required as GFP-positive MDA-MB-157 cells were not available. Therefore, GFP-negative MDA-MB-157 cells were used with GFP-positive fibroblasts, with FACS again used to separate the cell types. GFP-positive CAF1s, but not NFs, were available therefore this experiment was performed only with GFP-CAF1. Otherwise, the experiment was performed and data were expressed as previously (Figure 3.4A).

As previously, increasing proportions of CAF1s enhanced the clonogenic plating efficiency of the cancer cells in the absence of epirubicin, however this was not found to be statistically significant (compare light green bars in Figure 3.4 A "Colony counts", p value 0.337). The findings in MDA-MB-157 cells with respect to chemotherapy response mirrored those found with MDA-MB-231 cells. Epirubicin treatment reduced clonogenic survival in the purely epithelial culture, 0% fibroblasts, to 32% compared to untreated. As CAF1s were

introduced, MDA-MB-157 survival increased but the overall trend in survival was deemed not significant by linear regression (p value 0.242). However comparison of survival of 0% vs 55% cultures, using Mann-Whitney, was significant in isolation (p<0.05) (Figure 3.4 A "Relative to no chemo", compare across the dark green bars).



В



Colony counts Normal fibroblasts Cancer fibroblasts Normal fibroblasts 150 1.00 Number of colonies 00 00 0.75 Relative survival 468 0.50 0.25 0.00 0 0 8 20 55 0 8 20 55 0 8 2055 0 8 2055 Fibroblast % 0 20 8+

Relative to no chemo

Cancer fibroblasts



Cultures of MDA-MB-231-GFP/luc or -468-GFP cells were established with various proportions of either immortalised normal breast fibroblasts (NF1) or immortalised breast CAFs (CAF1). Epithelial cells had been labelled with GFP, so the cell types could be differentiated by flow-cytometry. Cultures were treated with 10nM epirubicin or untreated for 24h. Cultures were then treated by FACS to prepare pure epithelial populations. 500 epithelial cells were seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are normalised to the no treatment value for each individual fibroblast proportion and represent the means (+/- SD) of 3 independent experimental repeats. Linear regression analysis was carried out on no treatment for colony counts and normalised treated data with significant differences in the overall trend across the fibroblast proportions calculated. P values * <0.05



Figure 3.4: CAF1s protect MDA-MB-157 cells from epirubicin Cultures of MDA-MB-157 cells were established with various proportions of GFP positive immortalised breast CAFs (CAF1-GFP). Fibroblasts had been labelled with GFP, so the cell types could be differentiated by flow-cytometry. Cultures were treated with 25nM epirubicin or untreated for 24h. Cultures were then treated by FACS to prepare pure epithelial populations. 500 epithelial cells were seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are normalised to the no treatment value for each individual fibroblast proportion and represent the means (+/- SD) of 3 independent experimental repeats. Linear regression analysis was carried out on no treatment for colony counts and normalised treated data with significant differences in the overall trend across the fibroblast proportions calculated. No significant differences were observed.

3.3.3 Primary breast CAFs, but not primary NFs, protect the claudin-low TNBC cell line MDA-MB-231 from epirubicin

To assess further whether CAF-induced protection from epirubicin was generalizable, I next tested whether primary breast fibroblasts could enable protection from epirubicin. A matched pair of primary NFs (pNF) and primary breast CAF (pCAF) cultures was obtained: NFs were extracted from a breast cancer resection specimen at least 1cm outside the tumour margin, and CAFs were taken from within the tumour mass itself. The influence of this matched pair of primary fibroblasts on clonogenic survival after epirubicin treatment was assessed in MDA-MB-231 cells exactly as previously. MDA-MB-231-GFP/luc cells were used, to allow fluorescent cell sorting, while the experiment was not performed with MDA-MB-157 cells as neither GFP-positive primary fibroblasts, nor GFP-positive MDA-MB-157 cells were available.

In the absence of epirubicin treatment, fibroblasts again induced a dosedependent increase in clonogenic plating efficiency (Figure 3.5A 'colony counts', bright orange and pink bars). This was evident both with pNFs and with pCAFs, although the pCAFs achieved near maximal increases in survival at a lower percentage (compare 8% bars), suggesting their influence was stronger. After normalization to allow focus on responses to epirubicin, all proportions of pCAFs increased protection of MDA-MB-231 cells, but pNFs did not (Figure 3.5 'relative to no chemo').



Figure 3.5: pCAFs, but not matched pNFs, protect MDA-MB-231 cells from epirubicin

Cultures of MDA-MB-231-GFP/luc cells were established with various proportions of either primary normal breast fibroblasts (pNF) or primary breast CAFs (pCAF). Epithelial cells had been labelled with GFP, so the cell types could be differentiated by flow-cytometry. Cultures were treated with 10nM epirubicin or untreated for 24h. Cultures were then treated by FACS to prepare pure epithelial populations. 500 epithelial cells were seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are normalised to the no treatment value for each individual fibroblast proportion and represent the means (+/- SD) of 1 independent experimental repeat.

<u>3.3.4 Immortalised CAF1 cells protect MDA-MB-231s from an alternative</u> <u>chemotherapeutic, docetaxel</u>

Next, I wanted to investigate if CAFs could also confer protection from another, functionally-unrelated, chemotherapeutic. I chose to examine the taxane, docetaxel, which is – like epirubicin – routinely used in treatment of TNBC. The same clonogenic survival methodology was used as above, with only MDA-MB-231 and immortalized CAF1s as the cells under test (Figure 3.6), since these had previously demonstrated strong CAF-dependent protection from epirubicin.

Docetaxel treatment reduced clonogenic survival to 22% in the 0% fibroblasts MDA-MB-231 cultures ("Relative to no chemo" plot, Figure 3.6), while the presence of CAF1 cells dose-dependently increased survival (p<0.05) after docetaxel treatment to a maximum of 61% survival with the highest proportion of CAF1 cells relative to the untreated.



Figure 3.6 CAF1s protect MDA-MB-231 cells from docetaxel Cultures of MDA-MB-231-GFP/luc cells were established with various proportions of immortalised breast CAFs (CAF1). Epithelial cells had been labelled with GFP, so the cell types could be differentiated by flow-cytometry. Cultures were treated with 350pM docetaxel or untreated for 24h. Cultures were then treated by FACS to prepare pure epithelial populations. 500 epithelial cells were seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are normalised to the no treatment value for each individual fibroblast proportion and represent the means (+/- SD) of 3 independent experimental repeats. Linear regression analysis was carried out on no treatment for colony counts and normalised treated data with significant differences in the overall trend across the fibroblast proportions calculated. P values * <0.05

3.3.5 Fibroblast-dependent protection of claudin-low TNBC cell lines MDA-MB-157 and MDA-MB-231 is not reproduced by fibroblast conditioned medium

Next, I wanted to investigate mechanisms by which CAFs were able to protect selected epithelial breast cancer lines from chemotherapy. My first hypothesis was that the fibroblasts release a specific secreted factor(s) from CAFs, but not from NFs, that acts on the target epithelial cells. In order to test this, I collected conditioned medium from fibroblast mono-cultures (either NFs, NF1, or CAFs, CAF1) and assessed its ability to stimulate chemoresistance in epithelial cultures.

Clonogenic survival assays were again performed using either MDA-MB-231, MDA-MB-157 or MDA-MB-468 cells. In these experiments, mono-cultures of these epithelial cells were treated with conditioned media taken from fibroblast cultures to simulate the presence of fibroblasts (labelled "M" on the Figures), or were left untreated to simulate the 0% fibroblast cultures. A further layer of complexity was also added to the experiment to take account of the fact that it was possible that fibroblasts were only induced to release the relevant secreted factor(s) by epirubicin treatment itself. Therefore, conditioned medium was also collected from fibroblast mono-cultures that had been pre-treated with epirubicin for 24h, and had then been swapped into epirubicin-free medium for conditioning of this fresh medium to take place; I have referred to this as "induced" conditioned medium ("iM").

In all cell lines used, MDA-MB-231, MDA-MB-157 and MDA-MB-468 no significant differences were observed on treatment with either NF1 or CAF1 conditioned media in either clonogenic plating efficiency (Figure 3.7 A, B, C left panel) or survival after epirubicin treatment (Figure 3.7 A, B, C right panel).





Cultures of MDA-MB-231-GFP/luc, 468-GFP and -157 cells were established in media containing 50% fibroblast conditioned medium that had been taken from NF1 or CAF1 after 24h of culture (either without, "M", or after pre-treatment of fibroblasts with epirubicin, "iM"). Epithelial cultures were then treated with 10nM epirubicin, with conditioned media replenished for 24h. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted (left panel). Data were also normalised to the no epirubicin treatment values for each

conditioned medium treatment. Data represent the means (+/- SE) of 3 (-231) or 2 (-468 and -157) independent experimental repeat. Two-tailed Mann-Whitney U tests were carried out to no conditioned with no significant differences observed.

3.4 Discussion

The aim of the work in this chapter was to identify whether the presence of fibroblasts (NFs or CAFs) had an impact on chemo-response of TNBC cells, and to determine if any impact was mediated by a secreted factor potentially contained within conditioned medium.

3.4.1 Methodology for assessing chemo-response

In order to study chemo-response *in vitro*, the ability to measure cancer cell survival following chemotherapeutic treatment was required. Over the years, various techniques to determine cell survival have been developed to assess not only chemotherapy response but also targeted drug and radiation responses (Menyhart et al, 2016). However, the particular circumstances of the research question I aimed to answer – ie the influence of one cell type on another in co-culture, defined that I needed to use slightly adapted assays.

The most common and well-established assay for assessing chemo-response *in vitro* are MTT assays. Importantly, MTTs can measure drug sensitivity in multiple different cancer cell lines (Alley et al, 1988). MTT assays are commonly used as they are cheap, convenient, quick and reliable at assessing survival (Kratze et al, 1996). Although use of MTT assays would enable assessment of chemo-response, MTT assays could not be used in the work described here for this purpose as co-culture experiments were carried out and the aim was to assess response of only the breast cancer cell lines through the background of differing amounts of fibroblasts.

Therefore, in order to assess survival of purely the epithelial cells in co-culture instead of the MTT readout, stable luciferase expression in the epithelial cells was exploited as a readout method. While uncommon *in vitro*, this methodology is commonly used for assessment of cell numbers, or cell survival after

treatments using *in vivo* models. Luciferase positive human breast cells have frequently been used in monitoring tumour development in localised and disseminated disease through bioluminescent imaging using IVIS imaging systems (Kalra et al, 2011). Cellular luciferase expression has also been used to quantify total cell numbers in novel *in vitro* 3D tissue engineered breast cancer models (Goliwas et al, 2017). Therefore, the use of luciferase as a read out was a viable option for assessment of purely epithelial cells in a mixed coculture of cells.

However, short-term assays commonly only last a few days due to cultures becoming confluent, therefore the use of a longer-term assay was also appropriate to determine the cellular fate of the cells in terms of potentially continuing to proliferate; therefore, I also used colony formation/clonogenic assays. Clonogenic assays are based on the ability of single cells to divide and continue to grow into individual colonies. Clonogenic assay show the reproductive ability following cytotoxic treatment and, therefore, the effectiveness of the treatment (Crowley et al, 2016). Clonogenic assays are deemed more representative of what occurs clinically; firstly, the ability of a tumour to respond to treatment and secondly the potential of relapse and formation distant metastases from the primary tumour (Franken et al, 2006).

Interestingly, and in support of my preferred use of clonogenic assays here, responses of breast tumours to chemotherapy have been successfully determined from patient samples using clonogenic assays (Jones et al, 1985). Among breast cancer patients who were deemed *in vitro* sensitive to treatment, 59% of patients responded clinically to chemotherapeutics, with resistance correctly predicted in 100% patients who were *in vitro* resistant to treatment. Thus clonogenic assays were found to reproduce clinical findings with respect to chemo-response impressively effectively (Jones et al, 1985). The clinical relevance of the long-term clonogenic assays in addition to the ability to assess response, cellular fate and cell regrowth following treatment were key features in choosing clonogenic assays for use in this chapter to assess chemoresistance.

For my clonogenic experiments, I needed to sort the epithelial cancer cells, the cells of interest in terms of chemo-response, away from the co-cultured fibroblasts, which I achieved using FACS. As far as I am aware, this combination of co-cultures, FACS, and clonogenic assays have not previously been published. The only related assay I have found is where breast cancer cells were seeded for clonogenic assays onto monolayers of fibroblasts (Samoszuk et al, 2005), which support their subsequent proliferation, but their initial response to chemotherapeutics was not assessed.

One of the difficulties for both of the assays I performed was deciding how best to show the treatment effects. In particular, since different numbers of epithelial cells were used in each condition under test, it was necessary to avoid this confounding experimental artifact to allow focus on the influence of the CAFs on epithelial survival. In many cases, in both short-term assays and long-term clonogenic assays, data are routinely normalised to the no treatment sample rather than using absolute values (Maycotte et al, 2012). In the case of my experiments, separate no treatment data were available for each individual fibroblast proportion under test, which led to more complex normalisation of each individual experimental value to its matched no treatment sample. The value of this method of normalisation was particularly evident in the clonogenic assays, where the increasing proportions of CAFs had an impact on the clonogenic plating efficiency of the cells in the absence of chemotherapy agents (visible in the colony counts), but also additionally on chemo-response (visible in the colony counts, but more clearly in data that were normalised relative to untreated samples).

3.4.2 Conditioned media clonogenic assays – a soluble mediator of cross-talk between CAFs and cancer cells?

Conditioned medium has frequently been used experimentally to show how secretory factors can impact on growth (Cullen et al, 1989), migration and invasion (Walter et al, 2009) of breast cancer cells. Therefore, I attempted to use the same approach to identify if CAFs were secreting a soluble factor that was driving the chemoresistance I had seen in the TNBC cells (section 3.2.5).

Previously, conditioned media from CAFs has been shown to increase migration and invasion of MDA-MB-231 cells significantly, through the chemokine CXCL12 (Dvorak et al, 2018). CAFs have also been shown to induce phenotypic transformation of breast cancer cells through use of CAF conditioned media on MCF-7 cells (Wang et al, 2018). Conditioned media from NFs and CAFs have also been shown to increase growth in SKBR-3, T47D and MDA-MB-468 cells using MTT assays (Merlino et al, 2017). It is therefore evident that CAFs could impact on breast cancer progression through conditioned media. However, here in this chapter CAF conditioned media had no significant impact on chemoresistance in MDA-MB-231, -157 or -468 TNBC cells (Figure 3.7).

The simplest interpretation of these results is that a secreted factor is not responsible for the CAF-induced chemoresistance in section 3.3.2, however there is a risk that the result is a false negative. It is possible that the secretory factor is a labile factor that is not well maintained within the conditioned medium, or that the concentration of the secretory factor in the conditioned media is at a low concentration. In order to minimise the risk of these false negatives, I ensured that my conditioned medium was prepared fresh for every experiment (rather than being stored), and I reduced the medium volume on the conditioning cells in an effort to concentrate any secreted factors. A further possibility is that the CAFs may need to be induced to produce the proposed secretory factor, perhaps by the epirubicin treatment itself, or by the epithelial cells, representing bi-directional cross-talk. With respect to the hypothesis that the epirubicin treatment could induce the CAFs to start secreting the relevant factor(s), there are published precedents for this. Treatment induced changes in CAF secretory molecules and subsequent effects on cancer cells have been shown in colorectal cancer, where chemotherapy activated CAFs to increase secretion of IL-17A that in turn led to remodelling of the tumour microenvironment to self-renew colorectal cancer-initiating cells (Lotti et al., 2013). In breast cancer, chemotherapy has been shown to induce CAFs to secrete IL-6 inducing a pro-inflammatory microenvironment, which subsequently activated interferon-mediated signaling as well as stemness

pathways (Sonic Hedgehog signaling) in breast cancer cells (Peiris-Pagès et al, 2015). In addition, chemotherapy induced damage of the microenvironment and CAFs in breast cancer led to increased levels of WNT16B, which drove breast cancer resistance to docetaxel / mitoxantrone combination therapy (Sun et al, 2012). To take into account the potential influence of epirubicin directly on the CAFs themselves, in my experiments (section 3.2.5) I also pre-incubated CAFs with epirubicin and collected "induced" conditioned medium following this treatment. Interestingly induced CAF induced conditioned media didn't behave significantly differently from the non-induced conditioned media, suggesting that epirubicin treatment alone did not induce CAFs to secrete a relevant factor. However, making induced conditioned media was slightly problematic as it was essential to pre-treat CAFs with epirubicin, then wash out epirubicin and collect induced conditioned media following this treatment, not during the treatment. If media was collected during epirubicin treatment, induced conditioned media could have added an additional substantial dose of epirubicin to cells and led to further cell death. However, it is possible sustained chemotherapy treatment is required to induce CAFs correctly but by collecting media following treatment, this could have been missed.

Therefore, because of these unavoidable experimental weaknesses with the design of the conditioned medium experiments undertaken, it is not possible to rule out a role for a simple secreted factor. Nevetheless, conditioned media data also supports the model of a more complex interaction between the CAFs and epithelial cells, where bi-directional signaling or physical contact between the cell types is required.

3.4.3 CAFs induce enhanced clonogenicity

One unexpected finding, which was not an initial aim of my experiments, was the discovery that CAFs, but not NFs, induced increased clonogenic survival in cancer cells in the absence of chemotherapy treatment. This effect was often normalised away to allow focus on chemo-response (for example, in Figure 3.3 compare the "Relative to no chemo" and the "Colony counts" plots). However, it is worthy of consideration in its own right. During these clonogenic assays, the cells are detached from their substrate and are "rounded up"; cells remain suspended in solution until they are re-plated and are able to reattach. This can be a traumatic insult to many cell types – particularly epithelial cells – and can induce apoptotic cell death, termed anoikis in this context (Ishikawa et al, 2015). One interpretation of my observation regarding CAF-induced increased clonogenicity is that CAFs had induced anoikis-resistance in the co-cultured epithelial cancer cells. In terms of cancer biology, resistance to this type of induced apoptosis is of great interest, as some resistance is required to survive metastatic dissemination via blood or lymph (Cao et al, 2017). Therefore, an interesting hypothesis that could follow on would be that epithelial cancer cells from fibroblast-rich tumours might survive longer in the blood and could therefore be more metastatic. The potential role of CAFs in modifying anoikis in epithelial cells is explored experimentally in Chapter 6.

3.5 Conclusions and aspects of future work

I have demonstrated that CAFs, but not NFs, are capable of inducing enhanced clonogenic survival, and relative chemoresistance in selected TNBC cell lines. The protection from chemotherapy is not reproduced by conditioned medium taken from fibroblasts, indicating that either the relevant secreted factors are too labile or low in concentration to be transferred in this context, or that physical contact or bi-directional signaling is required between the two cell types. In the next chapter, I have aimed to explore the potential mechanisms of increased chemoresistance in CAF-protected epithelial cells, using transcriptomic approaches, with a view to identifying the signaling mechanism between the cell types and/or the effectors of the resistance itself.

Chapter 4- Cancer-associated fibroblasts induce up-regulation of interferon signaling in breast epithelial cancer cells, leading to chemoresistance

4.1 Abstract

CAF-dependent mechanisms of chemoresistance are poorly understood, with no firm consensus on the signaling pathways involved. Identifying potential signaling pathways driving chemoresistance in TNBC could pave the way for exploitation of these pathways to improve chemotherapy responses. In the previous chapter, I demonstrated that breast CAFs, but not NFs, can induce chemoresistrance in the TNBC cell lines MDA-MB-231 and MDA-MB-157.

I next used array technologies to investigate the changes in the transcriptome of MDA-MB-231 cells induced by the presence of CAFs during treatment with the chemotherapeutic epirubicin. A signature of up-regulation of interferon signaling was observed, as reflected in the up-regulation of multiple components and targets of the pathway, including MX1, OAS1 and miR-155. CAFs, but not NFs, were capable of inducing interferon signaling in MDA-MB-231 cells. Also, CAF-induced up-regulation of interferon signaling in breast cancer epithelial cells only occurred in the cell lines that were protected from chemotherapy by CAFs (MDA-MB-231 and MDA-MB-157 cells), but not in a line in which protection was not induced (MDA-MB-468 cells). In the absence of fibroblasts, recombinant interferons were sufficient to induce chemoresistance in MDA-MB-231 and MDA-MB-157 cells, but not in MDA-MB-468 cells. The expression profile of CAFs when co-cultured with breast cancer epithelial cells and treated with epirubicin implicated interferon-β as the activator of the pathway.

I concluded that identification of interferon signaling, as a key driver of CAFinduced chemoresistance in TNBC, could provide future scope for development of treatments to improve patient responses to chemotherapy.

4.2 Introduction

In Chapter 3, I showed that CAFs can stimulate chemoresistance in MDA-MB-231 and MDA-MB-157 TNBC cells, but not in the MDA-MB-468 TNBC cell line. My next interest was to try to understand how CAFs enable this chemoresistance.

Various potential mechanisms of chemoresistance have been identified. Of particular interests is the fact that stromal components have previously been implicated in chemoresistance, since this is similar to my observation of stromal fibroblasts inducing chemoresistance. The published data, however, appear to support chemoresistance through a physical barrier to chemotherapy penetration of the tumour that is enabled by production of collagen and increased interstitial pressure as well as increased ECM stiffness (Netti et al, 2000). Type I collagen production, that is commonly produced by fibroblasts, can also directly induce chemoresistance in cancer cells through integrin signaling (Armstrong et al, 2004). It is not obvious that these could be the mechanisms occurring in the context of my experiments, as it seems unlikely that CAFs could provide a physical barrier to drug delivery, or secrete substantial ECM proteins in standard two-dimentional co-culture on plastic.

It is also possible to speculate about other potential mechanisms by which CAFs could directly impact on the chemotherapy dose reaching the epithelial cells. For example, it is plausible that fibroblasts themselves could be responsible for inactivation of the chemotherapy agents. The majority of drug inactivation is driven by phase I and/or phase II enzymes, which are present in the liver, intestines and tumour cells (Pan et al, 2015). An important phase I group of enzymes is the human cytochrome P450s (CYPs), and these have been seen to detoxify anticancer drugs (Fujita, 2006). CYP3A4 is responsible for metabolizing targeted breast cancer treatments such as tamoxifen as well as chemotherapeutic agents like taxanes (Miyoshi et al, 2002). Mutations in CYPs have also been shown to be associated with poor responses to treatment in breast cancer. Changes in CYP2B6 and CYP2D6 have lead to poor responses to cyclophosphamide chemotherapy and tamoxifen respectively (Patel, 2015). A phase II set of enzymes, glutathione s-transferases (GSTs) are responsible for metabolizing a large number of chemotherapeutic drugs including cisplatin and cyclophosphamide, which can be used in breast cancer treatment (Sau et al, 2010, Allocati et al, 2018). However, data implicating CAFs in direct inactivation of chemotherapy agents are lacking.

I have taken the view that it may be more likely that CAFs induce changes in gene expression within the epithelial cells, perhaps by paracrine signaling, that allow the epithelial cells themselves to become more resistant. A number of activated signaling pathways within epithelial cells have been associated with chemoresistance, such as the EGFR pathway (Kuroda et al, 2010), which in turn can activate the MAPK (Zhang et al, 2014) and PI3K/AKT pathways (Page et al, 2000), or the NF-κB signaling pathway (Weldon et al, 2001). However, activation of these in breast cancer cells has not been attributed to the presence of CAFs, therefore these do not appear to present likely candidate pathways in the context of my findings. Various effectors of chemoresistance, such as xenobiotic drug pumps or enzymes capable of inactivating chemotherapy agents (Zheng et al, 2017), are relatively well established within the literature, but there is little evidence that the presence of CAFs can lead to up-regulation of their expression or activity leading to chemoresistance.

In the absence of clear and definitive candidate pathways that have been shown to enable CAFs to induce chemoresistance, I chose to examine changes in the gene expression profile that CAFs induce in co-cultured breast epithelial cells in the hope that these would provide clues about the key mechanisms of CAF-induced chemoresistance in this system. I focused on both mRNA and miRNA expression changes in order to gain a comprehensive insight, and have tested the functional role in chemoresistance of candidates identified from these screens.

4.3 Results

4.3.1 CAF1s induce changes to the mRNA transcriptome of MDA-MB-231 cells during epirubicin treatment

My first aim was to identify the gene expression changes in TNBC cells induced by the presence of CAFs that could be responsible for CAF-induced chemoresistance.

To achieve this, MDA-MB-231-GFP/luc breast cancer cells were cultured on their own (0% fibroblasts) or were co-cultured with CAF1 cells (seeding the fibroblasts as 20% of the initial culture) exactly as previously described (section 3.3.2). The 20% CAF proportion was chosen, as opposed to 8% or 55%, as this induced strong and reproducible chemoresistance in the epithelial cells and was most representative of the proportion of CAFs commonly seen in breast tumours (unpublished data from the Hughes group). Cultures were treated with epirubicin for 24h, and the epithelial cancer cells were isolated by FACS (on the basis that MDA-MB-231-GFP/luc cells expressed GFP) from both the epithelial alone (0% fibroblasts) and 20% fibroblast cultures as previously. RNA was then extracted from the MDA-MB-231-GFP/luc cells. This entire experiment was performed three times over three separate weeks (importantly, not simply using three separate concurrent parallel cultures) to provide robust biologically separate repeats. Expression profiling was performed on the three pairs of samples using Affymetrix Clariom D microarrays, and data were normalized to array medians.

Following successful completion of the Affymetrix clariom D microarray analyses, supervised hierarchical clustering was performed to determine what differences there were between epithelial expression in the 0% and the 20% cultures (Figure 4.1). It was clear that between the 0% and 20% groups there were differing expression levels for many genes with triplicates within each group also similar, thereby allowing robust comparison between the groups. Expression data from the 3 samples for 0% or 20% cultures were grouped and paired ANOVA tests were performed to identify significant changes in gene expression of either up- or down-regulation by at least 2-fold. The numbers of genes showing such changes are summarized in Table 4.1, and their names and fold-changes are listed in appendix Table 1.



Figure 4.1: Supervised hierarchical clustering of expression profiles from biological triplicate samples of MDA-MB-231 cells when in culture alone (0%) or with CAF1 cells (20%) following epirubicin treatment MDA-MB-231-GFP/luc cells were cultured alone (0%) or with CAF1 cells (20%) and were treated with 10nM epirubicin. MDA-MB-231-GFP/luc cells were purified by FACS on the basis of positive GFP expression and RNA was prepared. Three separate biological repeats were performed giving three pairs of samples. Gene expression was assessed using Affymetrix Clariom D microarrays, and comparisons were made between the 0% and 20% using the Affymetrix transcriptome analysis console (paired ANOVA tests with fold changes greater than +/- 2; supervised hierarchical clustering analyses).
Total number of gene probes	135750	
Differentially expressed genes	184	
Genes expressed more highly in	127	
20% compared to 0%		
Genes expressed at lower levels in	57	
20% compared to 0%	0.	

Table 4.1: Summary of gene expression changes in MDA-MB-231 cells between the 0% fibroblast and 20% fibroblast cultures MDA-MB-231-GFP/luc cells were cultured alone (0%) or with CAF1 cells (20%) and were treated with 10nM epirubicin. MDA-MB-231-GFP/luc cells were purified by FACS on the basis of positive GFP expression and RNA was prepared. Three separate biological repeats were performed giving three pairs of samples. Gene expression was assessed using Affymetrix Clariom D microarrays, and comparisons were made between the 0% and 20% samples (paired ANOVA tests with fold changes greater than +/- 2).

4.3.2 CAF1s induce changes to the miRNA transcriptome of MDA-MB-231 cells during epirubicin treatment

The same RNA samples as described above (section 4.3.1) were also used for analysis of miRNA expression profiles in order to identify miRNAs that were differentially expressed in the presence of CAFs. For this analysis, the three separate biological replicates for the two groups were pooled into single samples representing MDA-MB-231 cells with 0% fibroblasts or with 20% CAF1 cells. Expression was assessed using Taqman miRNA array cards, testing levels of 377 individual RNA species. Expression levels were normalized to expression of the small nucleolar RNA, RNU48, which is commonly used as an appropriate normalizer (Gee et al, 2011), and were compared between the 0% and 20% cultures. Differences in miRNA expression were categorised into four groups: present in the 0% culture but not in the 20% culture, present in the 20% culture but not the 0% culture, and increased or decreased relative expression from 0% to 20%. The numbers of miRNAs in each of these categories are shown in Table 4.2 ("Number of differentially expressed miRNAs" column), the total being 145. In order to reduce the number of candidate miRNAs of interest to a number that was experimentally tractable, thresholds were incorporated to define minimum fold-changes and maximum cT values. For all four groups categorised, I imposed a maximum threshold for qPCR cT values of 35, thereby reducing the chances of including miRNAs detected in error or expressed at very low levels. For miRNAs in the increased (from 0% to 20%) or decreased (from 0% to 20%) expression groups, I set a further arbitrary threshold of requiring a minimum of 2-fold change to remain of interest. The numbers of miRNAs remaining of interest after incorporating these thresholds are shown in Table 4.2 ("…after thresholds" column), with the total being 45; their identities are listed in appendix Tables 2 and 3.

Category	Number of differentially expressed miRNAs	Number of differentially expressed miRNAs after thresholds
Present in 0% but not 20%	14	3
Present in 20% but not 0%	45	18
Increase from 0% to 20%	54	13
Decrease from 0% to 20%	32	11

Table 4.2: Numbers of miRNAs differentially expressed between MDA-MB-231 cells cultured alone (0%) or with CAF1 cells (20%) during epirubicin treatment RNA was prepared, as described for Figure 4.1, from triplicate cultures of MDA-MB-231-GFP/luc cells alone or co-cultures with CAF1 cells. Triplicate samples were combined, and miRNA expression was profiled using TaqMan qPCR arrays. The numbers of miRNAs showing differential expression in the categories described are listed, either including all data and any degree of differential expression, or setting thresholds for detection level (maximum cT of 35 for all four groups) and fold difference (2-fold for increase and decrease groups).

Following the thresholds that had been put in place to reduce the number of differentially expressed miRNAs, literature searching was also carried out. For literature searching, each miRNAs expression in breast cancer was searched in addition to whether the miRNA had been previously been implicated in cancer progression or chemoresistance. This resulted in 5 miRNAs for further study (see Table 4.3, which names the miRNAs, categorises them, and gives a fold-change where appropriate).

MiRNA and group	Fold change
miR-32 (present in 0% not 20%)	-
miR-27a (present in 20% not 0%)	-
miR-155 (increase from 0% to 20%)	Fold change 108.93
miR-422a (increase from 0% to 20%)	Fold change 51.52
miR-454 (increase from 0% to 20%)	Fold change 2.63

Table 4.3: Five miRNAs that were differentially expressed between MDA-MB-231 cells cultured alone (0%) or with CAF1s (20%) were selected for further study

5 miRNAs were selected for further study as described in the main text. For miRNAs expression in only one sample, the sample in which the miRNA was present is stated. For miRNAs expressed in both samples, the extent and direction of differential expression is stated.

These final 5 miRNAs were then validated using qPCR to support the up- or down-regulation observed in the microarray experiments. The same pooled RNA from the 3 individual biological repeats (as previously) was used (Figure 4.2). Data for miR-27a, miR-155 and miR-422a broadly reproduced the findings from the microarray – with the microRNAs up-regulated in the presence of CAF1s. MiR-32 expression was not reliably detected by qPCR, and these data are excluded from the figure. For miR-454, down-regulation in the presence of CAF1s was seen in the qPCR data, which did not match the microarray observations. After these analyses, I concluded that miR-27a and miR-155 were the strongest candidates as regulators of chemoresistance, as they were dysregulated in the presence of CAFs during epirubicin treatment to the greatest extent and were consistent between the microarray and validation qPCR.





RNA was prepared, as described for Figure 4.1, from triplicate cultures of MDA-MB-231-GFP/luc cells alone or from co-cultures with CAF1 cells. Triplicate RNA samples were combined, and miRNA expression was assessed using TaqMan qPCR assays (normalised to RNU48). Data represent mean fold changes (+/-SD for technical triplicate analysis of one biological replicate). 4.3.3 MiR-155 and miR-27a do not have consistent, significant influences on survival of MDA-MB-231 cells in the absence or presence of epirubicin treatment, although miR-155 demonstrated a potentially relevant trend

Having identified miRNAs that were differentially expressed in cancer cells in the presence of CAFs (section 4.3.2), I was next interested to test whether the dysregulation of these miRNAs potentially contributed to the CAF-dependent chemo-protection of epithelial cells observed.

MiRNA mimics or inhibitors for miR-27a and miR-155 were transfected into MDA-MB-231-GFP/luc cells in mono-culture to allow manipulation of their expression levels and therefore testing of whether changes in the levels of these miRNAs were sufficient to alter chemo-response. After optimisation of transfections, appropriate dose of miRNA mimics and inhibitors for each of miR-27a and miR-155 were identified, and their abilities to cause successful over-expression or knockdown of miRNA expression respectively were validated by qPCR (Figure 4.3).



Figure 4.3: MiR-155 and miR-27a mimics and inhibitors successfully overexpress or knockdown appropriate miRNA

MDA-MB-231-GFP/luc cells were transfected in triplicate with different doses of miR-27a and miR-155 mimics or inhibitors as well as control mimic or inhibitor and left for 24h. miR2-27a mimics and inhibitors concentrations were 500nM and 50nM respectively with miR-155 mimics and inhibitors 500pM and 10nM RNA was then extracted from triplicate wells and combined. MiRNA levels were assessed using TaqMan qPCR assays (normalized to RNU48). Data represent mean fold change (+/- SD for technical triplicates, with biological n=1).

To investigate chemo-responses, cells were transfected with targeted or control mimics or inhibitors, and then incubated without a chemotherapy agent, or with 10μ M or 50μ M epirubicin for 24h. Cells were then cultured for a further 24h in fresh medium without drug, before MTT assays were performed to quantify viable cells remaining in the culture (Figure 4.4).

To determine the influence of these miRNAs on survival and proliferation in the absence of chemotherapy, raw MTT assay values from the untreated transfections were plotted (Figure 4.4, left panel). The mimics and inhibitors, for both miR-27a and miR-155 had only very small, and non-significant, influences on survival/proliferation. In the context of epirubicin treatment, miR-27a mimics or inhibitors (Figure 4.4A and B right panel) showed no significant effects on chemotherapy-response. Similarly, miR-155 mimics and inhibitors showed no significant effects (Figure 4.4C and D right panel).

However, these data were worthy of closer inspection. Cells transfected with miR-155 mimics showed a marginal increase in relative cell survival at both doses of epirubicin (Figure 4.4C right panel, p values 0.310 and 0.093), while those transfected with the miR-155 inhibitor showed a marginal decrease in relative cell survival at least at one dose (Figure 4.4D right panel, p values 0.937 and 0.132). These marginal changes were consistent with the hypothesis that increased miR-155 expression, as seen in the CAF-induced chemo-protected cancer cells, could contribute to chemo-resistance. Therefore, miR-155 remained of interest as a potential mediator of CAF-induced chemoresistance, even though the results from these assays were not significant.



Figure 4.4: MiR-155 and miR-27a do not have consistent, significant influences on survival of MDA-MB-231 cells in the absence or presence of epirubicin treatment using short-term assays

MDA-MB-231-GFP/luc cells were transfected with miR-27a mimics (500nM, A), miR-27a inhibitors (50nM, B), miR-155 mimics (500pM, C), miR-155 inhibitors (10nM, D) or matched mimic/inhibitor controls (concentrations matched to targeted mimics/inhibitors, A-D). 24h later, transfected cells were treated with 10 or 50µM epirubicin or were untreated for 24h. MTT assays were performed. Plots on the left show raw absorbance values for cells transfected with targeted mimics/inhibitors or matched controls in the absence of epirubicin. Plots on the right show data relative to samples without epirubicin treatment. Data represent means (+/- SE) for three independent biological repeats. Two-tailed Mann-Whitney U tests were carried out with no significant differences observed between control or mimic/inhibitor transfected.

4.3.4. MiR-155 may have a role in chemo-response in MDA-MB-157 cells

Having shown that miR-155 was a potential interest in MDA-MB-231 cells, the assay was repeated in MDA-MB-157 cells. In the absence of epirubicin treatment, miR-155 mimics or inhibitors had no impact on cell survival (Figure 4.5A and B left panels). However, as for MDA-MB-231 cells, the miR-155 mimic and inhibitors had marginal, but non-significant, influences on chemo-response (Figure 4.5 A and B right panel). Consistently with data in MDA-MB-231 cells, miR-155 mimic increased chemoresistance (p values 0.818) and miR-155 inhibitors significantly sensitized the cells to chemotherapy (Figure 4.5B right panel p<0.01) therefore miR-155 remained potentially of interest.





MDA-MB-157 cells were transfected with miR-155 mimics (A), miR-155 inhibitors (B) or matched mimic/inhibitor controls (A-B). 24h later, transfected cells were treated with 10 μ M epirubicin or were untreated for 24h, followed by a further 24h culture in drug free media. MTT assays were performed. Plots on the left show raw absorbance values for cells transfected with targeted mimics/inhibitors or matched controls in the absence of epirubicin. Plots on the right show data relative to samples without epirubicin treatment. Data represent means (+/- SE) for n=3. Two-tailed Mann-Whitney U tests were carried out with no significant differences observed between control or mimic/inhibitor transfected.

4.3.5 Co-culture with CAF1s during epirubicin treatment induces dysregulation of two major biological pathways in MDA-MB-231 cells

Returning to the mRNA analyses using Affymetrix arrays, one way of gaining biological insights from large lists of differentially expressed genes is to analyse these genes for any changes (up-regulation or down-regulation) within specific pathways, thereby potentially defining dysregulated pathways. In my case, I was interested to identify pathways that could potentially play roles in the CAFinduced chemoresistance of MDA-MB-231 cells. Therefore, all the genes that were defined as differentially expressed between the 0% fibroblast MDA-MB-231 cultures and the 20% CAF1 MDA-MB-231 cultures (184 genes, Table 4.1) by the Affymetrix expression arrays, were analysed using a pathway enrichment platform, ToppGene (Chen et al, 2009). Analysis identified two notable and significantly dysregulated pathways (Table 4.4): interferon signaling and genes responsible for encoding components of the extracellular matrix. Differentially expressed genes from within these pathways are listed in Table 4.5 (interferon signaling) and Appendix Table 4 (extracellular matrix) with a full list of dysregulated pathways in Appendix table 5. For both pathways, all the identified pathway components were up-regulated in the co-culture with CAF1s (20%), as compared to the mono-culture (0%), suggesting that CAFs induce comprehensive up-regulation of these pathways.

Pathway	P Value	Number of genes	Number of genes
		from 184 input	in pathway
Interferon			
alpha/beta	5.127x10 ⁻¹³	12	69
signaling			
Ensemble of			
genes encoding			
extracellular	3.676x10 ⁻¹²	29	1029
matrix and		20	1020
extracellular			
matrix proteins			

Table 4.4: Molecular pathways enriched in the differentially expressed genes in MDA-MB-231 cells after co-culture with CAF1s, as identified using the ToppGene portal

Genes showing significant differential expression of greater than 2-fold between a mono-culture of MDA-MB-231-GFP/luc cells (0% fibroblasts) and the MDA-MB-231-GFP/luc cells within a co-culture with 20% CAF1 cells were analysed using ToppGene gene enrichment analysis. The top two significant pathways identified are listed, with the number of genes from the inputted list within that pathway, and the total number of genes identified as part of the specified pathway. All other dysregulated pathways from ToppGene analysis are listed in appendix table 5. The individual genes from the interferon signaling pathway are listed in Table 4.5.

Target	Fold change 1	Fold change 2	Fold change 3
IRF9	8.41	5.17	3.42
STAT2	4.3	4.6	2.06
OAS1	36.36	21.93	37.03
OAS2	15.52	4.71	5.95
MX1	46.38	32.23	15.66
MX2	18.43	13.66	10.79
IFI6	11.91	13.35	3.25
IFI44L	12.04	8.37	11.83
IFI27	12.92	7.87	6.2
IFITM1	5.56	4.71	3.09
USP18	7.13	9.15	5.15
XAF1	7.83	3.93	3.74

Table 4.5: Genes in the interferon signaling pathway that were significantly upregulated in MDA-MB-231 cells by co-culture with CAF1s during epirubicin treatment

See Table 4.4 legend for experimental details. Genes are listed with the fold changes associated with each of the three biological replicate analyses.

As discussed in section 4.2, it has already been reported that CAF-induced changes in the ECM may have roles in chemoresistance, therefore I chose to look further into the potential role of the interferon signaling pathway in chemoresistance as this appeared to have more novelty. In addition, *in vitro* work investigating the interferon signaling pathway would potentially be experimentally easier to manipulate and analyse since a range of chemical inhibitors are available and the pathway is comparatively simple compared to the very diverse signaling induced by the ECM and complexity involved in using 3D ECM scaffolds. In addition, the interferon pathway was chosen given that miR-155 (from the miRNA screen in section 4.3.2 and 4.3.3/4) has been reported as a positive regulator of the interferon pathway (Forster et al, 2015).

4.3.6 Interferon signaling is stimulated by CAF1s in epithelial cell types that are protected from epirubicin treatment by CAF1s, but not in cells that are not

To confirm my proposed CAF-dependent up-regulation of interferon signaling, I selected key genes from those identified as up-regulated to act as endogenous "reporter genes" for the pathway. OAS1 and MX1 were selected as these are very well-established interferon-responsive genes (Schoggins et al, 2011), and they had very high fold-changes (means >25) in the microarray data (Table 4.5). I also selected miR-155 that I demonstrated to be deregulated previously (section 4.3.2 and Figure 4.2), since it is both a positive regulator of the interferon pathway and is itself up-regulated by the pathway (Forster et al, 2015).

Next, I repeated the experiment where MDA-MB-231-GFP/luc cells were grown in mono-culture (0%) or in co-culture with 20% CAF1 for 24h. However, in this repeat, I treated cultures either with or without epirubicin to allow assessment of whether epirubicin impacts on the CAF-induced interferon signaling stimulation. As before, epithelial cells were purified by FACS, and RNA was extracted for qPCR expression analysis of OAS1, MX1 and miR-155 (Figure 4.6 top row). Furthermore, the same experiment was performed using two further cell lines: MDA-MB-157 cells (Figure 4.6 middle row), which had also demonstrated CAFdependent chemoresistance (see Fig 3.4), and MDA-MB-468 cells (Figure 4.6 bottom row), which had not shown chemoprotection from CAFs (see Figure 3.3).

All 3 reporter genes (OAS1, MX1 and miR-155) demonstrated CAF-induced upregulation in the two cell lines that were previously protected from chemotherapy by CAF1s (MDA-MB-231 and MDA-MB-157) (Figure 4.6 top and middle row), with expression of these endogenous reporter genes further increased by epirubicin treatment. However, epirubicin treatment was not itself sufficient to induce interferon signaling in the absence of CAFs. In contrast, MDA-MB-468 showed no sign of CAF-dependent activation of interferon signaling (Figure 4.6 bottow row), which can be observed by the much lower levels of expression compared to MDA-MB-231 and MDA-MB-157 cells under all conditions (note the very different y-axis scales between Figure 4.6 top row and bottom row). To conclude, CAF-dependent protection from epirubicin appeared to correlate with CAF-dependent up-regulation of the interferon pathway.



Figure 4.6: Interferon response genes are up-regulated in MDA-MB-231 and MDA-MB-157 cells by CAF1s, but not in MDA-MB-468 cells MDA-MB-231-GFP/luc, -468-GFP and -157 cells were cultured on their own or in combination with 20% CAF1s with and without 10nM epirubicin treatment. Cultures were then sorted by FACS to give pure epithelial populations. RNA was then extracted from the epithelial cells and qPCR used to determine relative expression of interferon response genes OAS1, MX1 and miR-155. mRNA qPCRs were normalised to ACTB and miR-155 to RNU48. Data represent the mean of technical triplicates (+/- SD) from one biological experiment apart from miR-155 analysis in MDA-MB-231 cells, which is from 3 biological experiments (+/- SE). Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value ***<0.001.

4.3.7 Recombinant IFNs are sufficient to induce chemoresistance in MDA-MB-231 and MDA-MB-157 cells, but not MDA-MB-468s

Having demonstrated that interferon activation correlated with CAF-induced chemoprotection, my next aim was to identify if an interferon signal was *sufficient* to induce this chemoresistance in TNBC cell lines in the absence of CAFs. In order to do this, recombinant IFN- α and IFN- γ , representing interferons that signal through the Type I or Type II receptors respectively, were added in a range of concentrations to mono-cultures of, initially, only MDA-MB-231-GFP/luc cells for 24h. Cells were then treated with 10nM epirubicin for 24h in the continued presence of recombinant IFNs. Clonogenic survival was assessed (Figure 4.7).

The first finding to note was that recombinant IFN- α or recombinant IFN- γ increased the plating efficiency of the cells in a dose-dependent manner in the absence of epirubicin, which was significant for IFN- γ (Figure 4.7 "Colony counts" data, p<0.01). This observation was similar to the co-culture experiments, where the presence of CAFs in cultures enhanced the clonogenicity of these cells (for example, see Figure 3.3). As previously, in order to determine effects on chemoresistance, colony counts after epirubicin treatment were normalized to their respective untreated values (Figure 4.7 "Relative to no chemo" data). This revealed a significant dose-dependent increase (p<0.0001) in survival after epirubicin treatment in MDA-MB-231 cells resulting from treatment with either recombinant IFN- α or IFN- γ .

Having determined that recombinant IFN- α and IFN- γ separately were sufficient to induce chemoresistance in MDA-MB-231 cells, I next investigated whether IFN- α and IFN- γ in combination would have an additive effect. In addition, I expanded the experiment into MDA-MB-157 and MDA-MB-468 cells (Figure 4.8). Again IFN- α or IFN- γ significantly increased clonogenic plating efficiency in both MDA-MB-231 and MDA-MB-157 (Figure 4.8 "Colony counts" data, p<0.05), which was further increased by the combination of IFN- α and IFN- γ . However, IFNs had no significant influence on MDA-MB-468 cells in this assay (Figure 4.8). It should be noted that MDA-MB-468 cells are the cell line that was previously found to be un-responsive to the presence of CAFs in this assay in terms of chemo-response (see Figure 3.3) and to CAFs in terms of upregulation of IFN-signaling (Figure 4.6)

As previously, chemotherapy-specific effects were determined by normalization. Introduction of IFN- α or IFN- γ on their own provided significant protection for MDA-MB-231 and MDA-MB-157 cells from epirubicin, which was further increased by the combination of IFN- α and IFN- γ (Figure 4.8 "Relative to no chemo, p<0.0001, p<0.01). However, there was no significant change in chemo-response in MDA-MB-468 cells (Figure 4.8 "Relative to no chemo").





MDA-MB-231-GFP/luc cells were cultured with IFN- α or IFN- γ separately at 3 different doses for 24h (500, 2500, 5000pg/ml and 500, 1500, 3000pg/ml respectively). Cells were then treated with 10nM epirubicin hydrochloride for 24h along with re-addition of fresh IFNs. Cells were then treated to determine clongenic survival as previously. Data are normalised to the no treatment value for each IFN condition and represent the means (+/- SE) of 3 independent experiments. One-way ANOVA was performed on no treatment data from colony counts and normalised treated data with significant differences between no IFN and doses of IFN calculated. P value ** <0.01 and **** <0.0001



Figure 4.8: Recombinant IFNs are sufficient to stimulate chemoresistance in MDA-MB-231 and MDA-MB-157 cells but not MDA-MB-468 cells MDA-MB-231-GFP/luc, -468-GFP and -157 cells were cultured with IFN- α or IFN- γ separately or combination for 24h (2500pg/ml and 1500pg/ml resepectively). Cells were then treated with 10nM or 25nM epirubicin hydrochloride for 24h with IFNs re-added. Cells were then treated to determine clongenic survival as previously. Data are normalised to the no treatment value for each IFN condition and represent the means (+/- SE) of 2 independent experiments. One-way ANOVA was performed on no treatment data from colony counts and normalised treated data with significant differences between no IFN and doses of IFN calculated. P value * <0.05, ** <0.01 and **** <0.0001

<u>.</u>

In an effort to understand why MDA-MB-468 cells failed to be protected by either CAF1 cells or recombinant IFNs, I used qPCR for the interferon reporter genes OAS1 and MX1 to assess whether the MDA-MB-468 cell line was capable of activating the interferon signaling pathway in response to an appropriate signal. MDA-MB-468 cells were treated with recombinant IFN- α or IFN- γ , or both, with or without epirubicin exactly as previously. RNA was then extracted and qPCR was performed (Figure 4.9). OAS1 and MX1 expression levels were not stimulated by treatment with recombinant IFNs suggesting that this cell line is simply unable to respond to an IFN stimulus.



Figure 4.9: MDA-MB-468 are unable to activate interferon signaling in the presence of recombinant IFNs

MDA-MB-468-GFP cells were cultured IFN- α or IFN- γ separately or combination for 24h (2500pg/ml and 1500pg/ml resepectively) with and without 10nM epirubicin treatment for 24h. RNA was then extracted from the epithelial cells and equal amounts of RNA were used for qPCR to determine relative expression of IFN-target genes OAS1 (left plot) and MX1 (right plot) that were normalised to ACTB. Data represent the mean (+/- SD) for n=1

4.3.8 CAF1s, but not NF1s, activate interferon signaling in co-culture with MDA-MB-231 cells

Both NFs and CAFs have vast secretomes, so I next wanted to identify whether activation of interferon signaling in co-cultured epithelial cells was a generic fibroblast function or if this attribute was specific to CAFs. In order to test this, I transfected interferon signaling luciferase reporter constructs into MDA-MB-231 cells and co-cultured these cells with either NF1 or CAF1 cells. As previously mentioned in the introduction (section 1.13.1 and 1.13.2), the interferon signaling pathway acts through two different transcription factor binding sites - the ISRE (Type I signaling) or the GAS element (Type I and II signaling) – and separate reporters for each were used. Following transfection with the reporters, MDA-MB-231 cells were either cultured alone, or in co-culture with increasing proportions of NF1s or CAF1s for 24h and 48h without treatment or with 10nM epirubicin treatment for 24h. Luciferase assays were performed to assess ISRE or GAS activity within the epithelial cells (Figure 4.10).

Interferon signaling activity at the ISRE and GAS elements was relatively low in MDA-MB-231 cells cultured alone at all timepoints, either without or with epirubicin treatment. Co-culture with normal fibroblasts had no significant influence on these basal levels. However, co-culture with CAFs produced a dramatic activation of both reporters, which increased with increasing proportion of CAFs.

ISRE reporter

GAS reporter



Figure 4.10: CAF1s activate ISRE and GAS activity in co-cultured MDA-MB-231 cells

MDA-MB-231 cells were transfected with ISRE or GAS reporter plasmids driving firefly luciferase expression, and a control plasmid (pRL-TK) with the HSV thymidine kinase promoter driving renilla luciferase expression. Transfected MDA-MB-231 cells were then cultured on their own or with different proportions of immortalised NF1s or CAF1s for 24h and 48h without treatment or with 10nM epirubicin treatment for 24h. Dual luciferase assays were performed, with firelfy readings normalised to renilla readings. Data represent the mean (+/- SD) for triplicate wells, for 1 biological replicate.

<u>4.3.9 IFN- β is secreted by CAF1s, and not NFs, following co-culture with MDA-</u>MB-231 cells, but not in CAF1 mono-cultures

Based on data in this chapter so far, my hypothesis was that CAFs, but not NFs, release interferon(s) as a paracrine signal, which stimulates interferon signaling in co-cultured breast cancer cells that are capable of responding to this signal (ie MDA-MB-231 and MDA-MB-157 cells, but not MDA-MB-468 cells). My next aim was to identify which interferon, of the many different subtypes (see section 1.11), might be responsible for this signaling and, therefore, for the subsequent chemoprotection observed in MDA-MB-231 and MDA-MB-157 cells.

Initially, ELISAs were used to attempt to quantify the levels of IFN- α or IFN- β . Media was taken from NF1s and CAF1s alone, MDA-MB-231 cells alone, and from 20% co-cultures of MDA-MB-231 with NF1 or CAF1s. All these cultures were treated with or without epirubicin. However, following analysis, levels of both IFN- α and IFN- β proved to be below the levels of accurate detection for this assay.

As an alternative, expression of IFN- α 2, IFN- β 1 and IFN- γ was assessed by qPCR in mono-cultures of immortalized NF1s, CAF1s and MDA-MB-231-GFP/luc cells as well as CAF1s and MDA-MB-231-GFP/luc cells isolated from 20% CAF1 co-cultures by FACS. In each case, these mono- or co-cultures were also treated with or without epirubicin.

Levels of IFN- α 2 and IFN- γ were below the level of reliable detection and therefore expression data were not available for these interferons. However, data for IFN- β 1 were more informative. There was little expression of IFN- β in NF1 mono-cultures with or without epirubicin treatment (Figure 4.11A left panel). IFN- β expression was similarly low in mono-cultures of CAF1s with and without epirubicin, However, when CAF1s were cultured with MDA-MB-231 cells, the expression of IFN- β in the CAF1s increased slightly without epirubicin, but very substantially with epirubicin treatment (Figure 4.11A, left panel). IFN- β 1 was consistently not detected in mono-cultures of MDA-MB-231 cells or in MDA-MB-231 cells sorted from the co-cultures with CAF1s (Figure 4.11A, right panel).

The same experiment described above was also carried out using a matched pair of primary NFs and CAFs taken from a TNBC patient (pNF and pCAF). Similarly to the result using the immortalised fibroblasts, IFN- β 1 expression was only detected in the pCAFs after they were co-cultured with MDA-MB-231 cells (Figure 4.11B, left panel), and was not detected in the MDA-MB-231 cells (Figure 4.11B, right panel).



Figure 4.11: IFN-β1 expression in CAF1s is up-regulated in the presence of MDA-MB-231 cells

Immortalised CAF1 (A left), or primary CAF (B left) were cultured on their own or with MDA-MB-231-GFP/luc cells (20% fibroblasts; 80% epithelial cells; "20%") for 24h. MDA-MB-231-GFP/luc cells (A and B right) were also cultured alone for 24h. Cultures were then either untreated (No T) or were treated with 10nM epirubicin (T). All cultures were treated for fluorescence-activated cell sorting, allowing separation of fibroblasts and MDA-MB-231-GFP/luc cells from the co-cultures on the basis of GFP expression. RNA was then extracted from mono- and co-cultured cells, and qPCR used to determine expression of IFN- β 1 relative to ACTB. Left plots represent expression in fibroblast populations from mono-cultures or co-cultures (denoted "20%"), while right plots represent expression in the MDA-MB-231 cells (mono-culture or "20%" co-culture). Data represent the mean of duplicate culture wells (+/- SD) for 1 biological replicate. ND denotes non detected values.

4.4 Discussion

The main aim in this chapter was to determine the potential molecular mechanisms responsible for the CAF-induced chemo-protection of TNBC cells that were observed in chapter 3. In order to achieve this, differences in expression profiles for both mRNAs and miRNAs were identified between MDA-MB-231-GFP/luc cells treated with epirubicin alone, or MDA-MB-231-GFP/luc cells treated in the presence of CAFs.

4.4.1 The choice of expression profiling methodology

A key consideration for expression profiling work, was the choice of expression profiling technology to be used, with the basic choice being between RNA sequencing (Stadler et al, 2009) or expression microarrays (Ness, 2007).

RNA sequencing is a relatively recent advance in gene expression profiling technology (Wang et al, 2009). At a very basic level, RNA is fragmented and each individual fragment is then sequenced. Sequence reads are then aligned to a reference genome, and expression of individual genes is estimated from the number of reads aligning to the gene (Kukurba and Montgomery, 2015). RNA sequencing can therefore provide analysis that is high coverage in terms of proportion of the transcriptome detected, high sensitivity (if sufficient read depth is available to detect rare transcripts) and high resolution in terms of splicing level information, in comparison to other gene expression profiling technologies (Hrdlickova et al, 2017). However, analysis of RNA-sequencing is known to involve complex bioinformatics and as yet standardized and generally accepted protocols for analysis have not been developed. In addition, RNA-sequencing can also be extremely expensive (Nagalakshmi et al, 2010). It was for these reasons that RNA sequencing was not used in this chapter for gene expression analysis.

Expression microarrays, on the other hand, are a well-established technology with reliable and relatively easy experimental and analysis pipelines. The technology enables the analysis of the RNA levels of thousands of genes simultaneously (Li et al, 2002), although a key point is that targets are predefined and limited, while RNA-sequencing has the potential to identify transcripts that were not previously known. The basic technology is that arrays contain single-stranded nucleic-acid probes that hybridise to target gene products and allow fluorescent detection proportional to expression (Kumar et al, 2012). Analysis of microarray data is relatively simple and user friendly with compatible software available for analysis with data exported into excel for further analysis (Cooper, 2001).

Microarrays have a long successful history of use in breast cancer research as demonstrated by their use in identifying differentially expressed genes between normal breast tissue and breast cancer (Perou et al, 1999) as well as identifying particular genes that predict pathological complete response and disease free survival in breast cancer patients who have different subtypes and stages of cancer (Perou et al, 2000, Okuma et al, 2016). In addition, gene expression analysis has also been undertaken in stromal cells, as opposed to the breast cancer cells. One study used microarrays to identify stromal gene expression changes in breast cancer progression, which could separate breast cancer patients into distinct groups displaying different clinical outcomes (Planche et al, 2011), with another study highlighting that molecular composition of tumour stroma in breast cancer was important for breast cancer prognosis (Winslow et al, 2016). Therefore, I was confident that the use of microarray technology would be suitable for identification of CAF-induced expression changes in my experiments.

In addition, analysis of miRNA expression levels was a further aim of this chapter. This can also be achieved using a wide-range of technologies, including RNA-sequencing, hybridization arrays, and qPCR arrays. qPCR arrays were chosen to determine miRNA expression as qPCR arrays are reliable with high sensitivity as well as being relatively easy experimentally. In addition, qPCR arrays could be carried out within the institute rather than being

outsourced. Use of qPCR arrays, again, had a successful relevant history in previous literature. MiRNA expression profiling identified up-regulated miRNAs in breast cancer patients by isolating circulating miRNAs. MiRNA expression was found to be higher in early stages of breast cancer, which could be useful for early detection of breast cancer. In addition, miRNA expression was slightly higher in TNBC and HER2 patients compared to ER positive patients (Hamam et al, 2016). The use of qPCR array was therefore a viable option for miRNA screening in this chapter.

4.4.2 miR-155: an oncomiR, as well as potential role in chemoresistance?

CAF-induced up-regulation of miR-155 in epithelial cancer cells was highlighted in this chapter, with the up-regulation implicated as an interferon-related mediator of chemoresistance.

The role of miR-155 in the CAF compartment has been explored previously. Expression in CAFs is of clear relevance to my findings, as it has been reported that miR-155 can be transfered between cell types in exosomes (Santos et al, 2018), and therefore the CAFs could be the source of the up-regulated expression of miR-155 in breast cancer cells. MiR-155 has been shown to be up-regulated in ovarian CAFs when compared to matched NFs (Mitra et al, 2012). Furthermore, miR-155 was required for the conversion of NFs to CAFs in the presence of ovarian tumour cells (Mitra et al, 2012). However, up-regulation of miR-155 has not been observed in breast cancer CAFs when compared to NFs (Schoepp et al, 2017). The cell-to-cell transfer of miR-155 was actually reported in the context of breast cancer cells, with miR-155 apparently transferred from cancer stem-like cells and treatment resistant MCF-7 cells to treatment sensitive MCF-7 breast cancer cells in culture (Santos et al, 2018) via exosomes. However, transfer of miR-155 from CAFs has not been reported.

In the context of clinical breast cancer samples, miR-155 has been shown to be significantly up-regulated in breast cancer and is associated with tumour subtype (Mattiske et al, 2012). When comparing 76 breast cancer samples to 10 normal breast samples, miR-155 was one of the most consistently up-

regulated miRNAs in the cancer samples (lorio et al, 2005). Another study identifying differential expression of miRNAs in specific tumour subtypes, identified up-regulation of miR-155 in only HER2+ breast cancer in comparison to other breast cancer subtypes (Tsai et al, 2018). Although these are excellent examples of miR-155 up-regulation, it is important to note that these studies used tumour samples comprising both cancer cells, and cancer stroma, therefore miR-155 expression could be present in either the CAFs or breast cancer cells or both.

In my experiments (section 4.3.6), three TNBC cell lines were cultured on their own as well as in co-culture with CAFs. When grown on their own, relative expression levels of miR-155 were low (figure 4.6), however in this instance there is no comparison to any normal breast line to assess whether this level is "up-regulated" as would be compatible with the studies using cancer samples containing mixed cell types. Up-regulation is clearly observed in the presence of CAFs for two breast cancer cell lines (MDA-MB-231 and MDA-MB-157) but, again, miR-155 levels were not compared to levels of miR-155 in normal breast cancer cells with CAFs, so it is difficult to conclude where CAF induced upregulation of miR-155 fits in with what is already known. Expression of miR-155 was determined in cell lines, in this chapter, instead of breast samples. Expression of miRNAs has also been studied in multiple breast cancer cell lines. One study has shown molecular subtype specific differential expression of miR-155, with relatively high expression of miR-155-3p in the normallike/claudin-low cell lines (MDA-MB-231 and -157) that were estrogen receptor negative. More interestingly, miR-155-5p expression was high in basal-like estrogen receptor negative (MDA-MB-468) cell lines (Riaz et al, 2013), High expression levels of miR-155-5p in basal cell lines contradicts with my observations in section 4.3.6 and figure 4.6, in which I show higher expression levels of miR-155-5p in MDA-MB-231 and MDA-MB-157 cells in comparison to MDA-MB-468 cells. The changes in miR-155 expression observed in section 4.3.6 and figure 4.6, clearly highlight the importance of CAFs in inducing expression of miR-155-5p.

The functional role of miR-155 has previously been explored and interestingly miR-155 plays a role in interferon signaling. Not only is miR-155-5p transcription increased by interferon signaling, but miR-155-5p also plays an important role in the signaling cascade of Type I interferon signaling (Forster et al, 2015). Normally SOCS1 prevents the formation of the ISGF3 complex containing the STAT1/STAT2 heterodimer and IRF9, leading to inhibition of interferon signaling (Piganis et al, 2011). However, miR-155-5p has been shown to bind to the 3'UTR of SOCS1 and therefore enables the formation of the ISGF3 complex of STAT1/2 and IRF9 (Jiang et al, 2010). The identification of CAF-induced miR-155-5p up-regulation as well as up-regulation of interferon signaling (Table 4.3) fits with the known role of miR-155-5p in the interferon signaling pathway. Increased levels of miR-155-5p will therefore drive increased interferon signaling. Additionally, the activation of interferon signaling could be the stimulus that leads to CAF-induced enhanced expression of miR-155-5p, a hypothesis supported by the observation that miR-155-5p was not induced by CAFs in the one cell line that also did not activate interferon signaling (MDA-MB-468 cells; Figure 4.6).

In this chapter, the function of miR-155 was studied in terms of miR-155 effect on survival and growth of cells (sections 4.3.3 and 4.3.4) in order to identify the functional importance of increased miR-155 expression. It has already been shown that miR-155 is involved in many key cellular pathways that potentially promote cancer development or progression, including apoptosis, proliferation and epithelial-mesenchymal transition (Liu et al, 2015). MiR-155 also been linked to cellular functions in MCF-7 cells, with miR-155 promoting proliferation (Martin et al, 2014). MAPK signaling is also a target of miR-155. MiR-155 upregulates p38 and enhances MAPK signaling to promote phosphorylation of ERK1/2 and cell proliferation (Martin et al, 2014). My data do not support a role for miR-155 in proliferation in the cell lines tested (MDA-MB-231 and MDA-MB-157; Figure 4.4 and 4.5), although it is important to note that the experiments were not designed to test this hypothesis, rather to investigate any potential role in chemo-response.

Importantly, miR-155's role in chemoresistance was investigated, because of the CAF-induced increase of miR-155 expression (section 4.3.2) in TNBC cells following epirubicin treatment. Interestingly miR-155 has previously been implicated as a potential marker of chemoresistance in TNBC (Ouyang et al, 2014). It is thought that miR-155 can lead to drug resistance in multiple different ways. MiR-155 has been shown to bind directly to and repress the function of FOXO3a, which is involved in processes such as cell death and cell cycle arrest (Dijkers et al, 2000). p27, an inhibitor of cell cycle progression, is a downstream target of FOXO3a and the presence of miR-155 therefore prevents activation and function of p27. p27 normally leads to the arrest of cells in G₀/G₁ (Chiarle et al, 2001). Presence of miR-155 and inhibition of FOXO3a and subsequently decreased levels p27 have been shown to be associated with docetaxel resistance in breast cancer cell lines (Brown et al, 2004). Here, miR-155 demonstrated a trend towards contributing to chemoresistance, although the findings were formally not statistically significant. MiR-155 mimics increased chemoresistance and miR-155 inhibitors decreased chemoresistance in both MDA-MB-231 and MDA-MB-157 cells (Figure 4.4 and 4.5). This trend was both consistent with the published literature above, and with my hypothesis that miR-155 would provide chemo-protection, based on its CAF-induced up-regulation in the context of CAF-induced chemo-protection. The function of miR-155 in chemoresistance is studied further in chapter 5.

<u>4.4.3 miR-27a</u>

The second miRNA that was identified as up-regulated in breast cancer cells by the presence of CAFs was miR-27a (section 4.3.2). CAF-induced up-regulation of miR-27a in epithelial cells has not previously been reported, however, the roles of miR-27a in CAFs themselves have been investigated. MiR-27a has previously been shown to induce the transition of NFs to the CAF phenotype, which in turn induced chemoresistance in esophageal cancer cells (Tanaka et al, 2015). However, up-regulation of miR-27a was not observed in the esophageal cancer cells nor did transfection with miR-27a confer chemoresistance in these cells, therefore this observation may be unrelated to my findings.

The functional role of miR-27a has been studied in TNBC. MiR-27a has been shown to have prognostic value in breast cancer patients, with high expression of miR-27a associated with poorer survival (Tang et al., 2012) and has been shown to part of a four miRNA signature specific to TNBC (Gasparini et al, 2014). However, my aim was to identify potential mediators of chemoresistance, and in this regard miR-27a was not promising since it was up-regulated in chemoresistant cancer cells (those co-cultured with CAFs, section 4.3.2), yet miR-27a over-expression or knock-down was not associated with significant or consistent changes in survival after chemotherapy treatment (Figure 4.4). By contrast, miR-27a inhibition in ER+ breast cancer cells has previously been shown to increase the sensitivity of the cells to cisplatin treatment (Zhou et al, 2016). MiR-27a is a positive regulator of multidrug resistance 1 (MDR1) / P-glycoprotein drug pump and has been shown to increase expression of MDR1 (Zhu et al, 2008). Both of these studies highlight that miR-27a can impact on chemoresistance, although this does not appear to be the case in my assays.

4.5 Conclusions and aspects of future work

It is evident from data presented in this chapter, that the presence of CAFs can up-regulate activity of the interferon signaling pathway and, related to this, expression of miR-155. It is also clear that interferons alone are sufficient to induce chemoresistance in representative cell lines of the claudin-low TNBC subtype. In my next chapter, I explore whether it is possible to block this influence of CAFs in order to inhibit the chemoprotective effect of CAFs.

Chapter 5 – Inhibition of interferon signaling reduces fibroblast-induced chemoresistance in triple negative breast cancer

5.1 Abstract

In chapter 4, I showed that interferon signaling was activated in selected TNBC cell lines by the presence of CAFs, and that activation was sufficient to induce relative chemoresistance. My next hypothesis was that inhibition of interferon signaling would improve the efficacy of the chemotherapy agent epirubicin in TNBC cell lines co-cultured with CAFs. This could raise the possibility that this combination treatment could be effective in patients with tumours containing high proportions of CAFs. Antibodies directed against interferon receptors that inhibit their activation, a JAK/STAT small molecule inhibitor, or miR-155 inhibitors were used to inhibit CAF-induced interferon activation.

CAF-induced chemoresistance in MDA-MB-231 and/or MDA-MB-157 was significantly reduced by the addition of interferon blocking antibodies directed against Type I or Type II receptors. qPCR was used to confirm that interferon signaling was indeed down-regulated. Similarly, the use of the JAK/STAT inhibitor ruxolitinib reduced CAF-induced chemoresistance in MDA-MB-231 and MDA-MB-157 cells. Chemoresistance induced by both immortalized and primary CAFs was sensitive to these inhibitions. MiR-155 inhibition also reduced CAF-induced chemoresistance in MDA-MB-231 cells.

I concluded that the use of interferon blocking antibodies, a JAK/STAT inhibitor, or miR-155 inhibitors can all synergise with epirubicin treatment by inhibiting CAF-induced chemoresistance.

5.2 Introduction

Interferon activation in TNBC cells is sufficient to drive increased resistance to the chemotherapeutic epirubicin (Chapter 4). My next interest was to examine whether inhibition of this activation could reduce CAF-induced protection from chemotherapy and could therefore present a potential therapeutic approach to sensitize cells to chemotherapy.

Interferon signaling is classically known to be involved in antiviral responses, however more recently roles of interferons in cancer have been studied. An example is the observation that activation of the interferon signaling pathway is associated with myeloproliferative disorders and hematological cancers. Enhanced JAK-STAT activities, down-stream mediators of interferon signals, have been identified in acute T-cell lymphocytic leukemia, acute B-cell lymphocytic leukemia and acute myeloid leukemia (Furumoto and Gadina, 2013). Therefore, inhibitors of these signaling intermediates have been developed as potential cancer therapeutics. Ruxolitinib is a leading JAK-STAT inhibitor and inhibitor of interferon signaling (Harrison et al, 2012). Ruxolitinib is a targeted therapy that selectively binds to non-receptor janus kinases (JAK), therefore inhibiting JAK1 and JAK2 activation and signaling. Ruxolitinib is currently FDA approved for treatment of myelofibrosis and polycythaemia vera (Yi et al, 2015). As ruxolitinib is already in clinical use, it could potentially be used in breast cancer patients in the near term. Ruxolitnib is therefore an ideal inhibitor for me to use in combination with chemotherapy to assess whether inhibition of interferon signaling improves chemotherapy response.

I also identified miR-155 as a potential regulator of chemoresistance, through the screen of changes in miRNA expression completed in Chapter 4. MiR-155 has previously been found to be over-expressed in a large number of solid cancers, including colon, breast, pancreatic, thyroid, cervical and lung cancers (Faraoni et al, 2009). MiR-155 has an established role as a positive regulator of interferon signaling, through targeting the protein SOCS, which normally functions to down-regulate interferon signaling (Forster et al, 2012). MiR-155 is itself also a down-stream target of interferon signaling, with activation of the pathway leading to the transcription of further miR-155; this represents a positive feedback loop (Jiang et al, 2010). Inhibition of miR-155 function therefore provides a further strategy to inhibit interferon signaling.

An interesting aspect of the CAF-induced chemoprotection I have observed is that this may be a paracrine activity, with CAFs releasing signals (presumably interferons themselves) that are received by the epithelial cancer cells. This means that use of antibodies that block the ability of interferons to act on their receptors provides another strategy to inhibit activation of the pathway. Different blocking antibody strategies are available: those that bind to interferons themselves or those that bind to the receptors to prevent interferon binding. One study has shown successful blocking of IFN- α and IFN- β by targeting them directly using monoclonal antibodies and by targeting the IFNAR2 receptor chain. Blocking of IFN- γ directly and the IFNGR receptor chain also successfully neutralized IFN- γ activity (Moll et al, 2008).

In this chapter, I aimed to inhibit interferon signaling in breast cancer epithelial cells, using multiple different strategies as outlined above, and determine whether this reduced CAF-induced chemoresistance.

5.3 Results

5.3.1 Type I and/or Type II interferon blocking antibodies individually block CAF1-induced chemoresistance in TNBC cells but the combination of both antibodies has no increased effect

My first aim was to assess whether antibodies capable of blocking activation of interferon receptors would inhibit CAF-induced chemoresistance.

To achieve this, I adapted my standard colony formation assay protocol and introduced Type I and Type II receptor blocking antibodies. MDA-MB-231-GFP/luc cells were cultured either on their own or with 20% CAF1 cells with either an isotype control antibody or blocking antibody added. Blocking antibodies for either Type I or Type II signaling were used either individually or in combination. Cultures were incubated for 24h and then treated with 10nM epirubicin for 24h, with the antibodies re-added to ensure inhibition of the pathway was maintained. As before, the cultures were then sorted to purify the GFP-expressing epithelial cells with equal numbers of cells plated at a low seeding density and cultured to allow individual colonies to be counted (Figure 5.1).

As seen previously and as expected, the results in Figure 5.1 consistently demonstrate that epirubicin treatment significantly reduced cell survival, and that co-culture with CAF1s both significantly increased seeding efficiency and induced significant chemoresistance (p<0.05). However, the key results in this experiment are the comparison of survival after epirubicin treatment in cells treated with either control or IFN-blocking antibodies. In CAF1 co-cultures both the Type I blocking antibody (Figure 5.1A, right plot) and the Type II blocking antibody (Figure 5.1B, right plot) caused significant reductions in survival after epirubicin treatment (p<0.05), meaning that the CAF-dependent protection was completely inhibited. Surprisingly, however, the combination of the two blocking antibodies did not significantly inhibit the CAF-dependent protection (Figure 5.1C, right plot, p value 0.486).

It was also evident from these data that each of the Type I and Type II blocking antibodies also inhibited the CAF-dependent increase in seeding density (Figure 5.1, compare control with blocking antibody for epithelial + CAFs without epirubicin in the colony count data set [left plot]), indicating that interferon signaling is also required for this effect of CAFs that is unrelated to chemoresponse.

It was also important to confirm that the blocking antibodies were indeed inhibiting interferon signaling as expected. Therefore, in addition to the colony forming assays, qPCR analysis was performed on the remaining cells that had been sorted but were not required for colony plating to assess levels of interferon pathway activity using the interferon stimulated genes OAS1 and MX1 as indicators of activity (Figure 5.2). As previously seen and expected, coculture with CAFs led to activation of interferon signaling with epirubicin treatment enhancing this activation (p<0.05). The blocking antibodies, either individually (Figure 5.2A and B) or in combination (Figure 5.2C), consistently and significantly reduced (p<0.05) this activation, although levels were not reduced to basal.




Cultures of MDA-MB-231-GFP/luc cells were established in mono-culture or coculture with 20% immortalised breast CAF1s. Type I interferon signaling (1µg/mI) or Type II interferon signaling (5µg/mI) blocking antibodies were either added separately (A and B) or in combination (C) or appropriate isotype controls were added. Cultures were treated with 10nM epirubicin or untreated for 24h with antibodies re-added. Cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are shown as numbers of colonies (left plots), or as counts normalised to the no treatment value for each individual fibroblast proportion (right plots), and represent the means (+/-SE) of 3 independent experimental repeats. Two-tailed Mann-Whitney U tests were carried out and selected significant differences are shown. P values * <0.05.



Figure 5.2: Interferon neutralizing antibodies successfully blocked CAF1dependent interferon activation in MDA-MB-231 cells when treated with epirubicin

Cultures of MDA-MB-231-GFP/luc cells were established in mono-culture or coculture with 20% immortalised breast CAF1s. Type I interferon signaling (1µg/mI) or Type II interferon signaling (5µg/mI) blocking antibodies were either added separately (A and B) or in combination (C) or appropriate isotype controls were added. Cultures were treated with 10nM epirubicin or untreated for 24h with antibodies readded. Cultures were then separated into pure epithelial populations by FACS. RNA was extracted from the epithelial cells and equal amounts of RNA were used for qPCR to determine relative expression of IFN-target genes OAS1 (left plots) and MX1 (right plots). qPCRs were normalised to ACTB. Data represent the mean(+/- SE) for n=3. Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value *<0.05.

5.3.2 Type I, but not Type II, interferon blocking antibodies block CAF1-induced chemoresistance in MDA-MB-157 cells

Next, the same entire experimental approach of clonogenic assays and qPCR pathway activity assessment was repeated using the alternative TNBC cells line MDA-MB-157, in which I had also observed CAF-induced chemoprotection (Chapter 3). Clonogenic survival data are presented in Figure 5.3, while matched qPCR data to assess interferon pathway activity are presented in Figure 5.4.

As previously, epirubicin treatment reduced survival, while co-culture with CAFs afforded chemoprotection in the absence of blocking antibodies (p<0.05 and p<0.01). In accordance with the findings in MDA-MB-231 cells, the Type I blocking antibody significantly (p<0.01) and completely inhibited this protection in MDA-MB-157 (Figure 5.3A, right panel). However, in contrast to findings in MDA-MB-231 cells, the Type II blocking antibody had no effect on CAF-dependent protection (Figure 5.3B, right panel, p value 0.486). Analysis of interferon pathway activity by qPCR confirmed that the Type I blocking antibody inhibited interferon signaling (Figure 5.4A, p<0.01), as expected. However, it was evident that in this cell line the Type II blocking antibody failed to inhibit the CAF-dependent activation of the interferon pathway (Fig 5.4B, p values 0.486 and >1), which provided an explanation for why this antibody had also had no functional effect on the CAF-dependent chemoprotection (Figure 5.3B, right panel).



Figure 5.3: Type I but not Type II interferon neutralization blocks immortalised CAF1 induced chemoresistance in MDA-MB-157 cells

Cultures of MDA-MB-157 cells were established in mono-culture or co-culture with 20% immortalised breast GFP-CAF1s. Type I interferon signaling (1µg/mI) or Type II interferon signaling (5µg/mI) neutralising antibodies were added separately (A and B) or appropriate isotype controls were added. Cultures were treated with 10nM epirubicin or untreated for 24h with antibodies readded. Cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are shown as numbers of colonies (left plots), or as counts normalised to the no treatment value for each individual fibroblast proportion (right plots), and represent the means (+/- SE) of 3 independent experimental repeats. Two-tailed Mann-Whitney U tests were carried out and selected significant differences are shown. P values * <0.05.



Figure 5.4: Type I but not Type II interferon neutralizing antibodies block CAF1dependent interferon activation in MDA-MB-157 cells

Cultures of MDA-MB-157 cells were established in mono-culture or co-culture with 20% immortalised breast GFP-CAFs. Type I interferon signaling (1µg/mI) or Type II interferon signaling (5µg/mI) blocking antibodies were either added separately (A and B) or appropriate isotype controls were added. Cultures were treated with 10nM epirubicin or untreated for 24h with antibodies readded. Cultures were then separated into pure epithelial populations by FACS. RNA was extracted from the epithelial cells and equal amounts of RNA were used for qPCR to determine relative expression of IFN-target genes OAS1 (left plots) and MX1 (right plots). qPCRs were normalised to ACTB. Data represent the mean(+/- SE) for n=3. Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value *<0.05.

5.3.3 Type I interferon blocking antibodies block pCAF-induced chemoresistance in MDA-MB-231 cells

The same approach was then taken to examine whether_interferon blocking antibodies could block chemoresistance induced by *primary* CAFs (pCAFs) derived from a TNBC patient. These pCAFs have already been shown to induce chemoprotection of MDA-MB-231-GFP/luc cells (Figure 3.5). Clonogenic assays and qPCR assessment were performed as above in section 5.3.1 and 5.3.2, in this case with the Type I interferon blocking antibody, pCAFs and MDA-MB-231 cells (Figure 5.5).

As previously, epirubicin treatment reduced survival, while co-culture with pCAFs afforded chemoprotection in MDA-MB-231 cells in the absence of blocking antibody (p<0.05). In accordance with the findings in MDA-MB-231 and MDA-MB-157 cells using immortalized CAF1s, the Type I blocking antibody significantly (p<0.05) and completely inhibited pCAF-induced protection in MDA-MB-231 (Figure 5.5A, right panel). Analysis of interferon pathway activity by qPCR confirmed that the Type I blocking antibody inhibited the CAF-induced interferon signaling as expected, with reduced induction of OAS1 and MX1 (Figure 5.5B, p values 0.029 and 0.343 respectively).



Figure 5.5: Type I interferon neutralization blocks pCAF-induced chemoresistance, and pCAF-induced interferon activation in MDA-MB-231 cells Cultures of MDA-MB-231-GFP/luc cells were established in mono-culture or coculture with 20% pCAFs. Type I interferon signaling (1µg/ml) blocking antibody was added or appropriate isotype controls were added (A). Cultures were treated with 10nM epirubicin or untreated for 24h with antibodies readded. Cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are shown as numbers of colonies (left plots), or as counts normalised to the no treatment value for each individual fibroblast proportion (right plots), RNA was also extracted from the epithelial cells and equal amounts of RNA were used for qPCR (B) to determine relative expression of IFN-target genes OAS1 (left plot) and MX1 (right plot). gPCRs were normalised to ACTB. Data represent the mean(+/- SE) for n=3. Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value *<0.05.

5.3.4 The JAK-STAT inhibitor ruxolitinib reduces CAF1-induced chemoresistance in MDA-MB-231 and MDA-MB-157 cells

Blocking antibodies successfully inhibited CAF-dependent protection above, however, the process for moving these blocking antibodies forward towards assessing their effect in cancer patients is likely to be prohibitively slow and expensive. Therefore, I searched for an alternative that might potentially make it into clinic more quickly if appropriate. A JAK-STAT inhibitor, ruxolitinib, was chosen as it has been shown to have high efficacy for inhibiting interferon signaling via binding to JAK2 (Thomas et al, 2015). Ruxolitinib is already available within the NHS and is used for treatment of myelofibrosis.

Experiments were performed broadly as above to assess colony formation in MDA-MB-231 or MDA-MB-157 cells after mono-culture or co-culture with CAF1s, and after treatment with epirubcin and/or ruxolitinib. qPCR was again used to assess interferon pathway activity. In this case, mono-cultures or co-cultures with CAF1s were pre-treated for 24h with either vehicle control or 10nM ruxolitinib and new vehicle or ruxolitinib was added with epirubicin treatment for 24h. Colony forming potential of the epithelial cells was then determined as previously (Figures 5.6, MDA-MB-231 cells, and 5.7, MDA-MB-157 cells).

As previously, epirubicin reduced survival in both MDA-MB-231 and MDA-MB-157 cells, and the presence of CAF1 cells induced relative chemoresistance (p values 0.067 and p<0.01 respectively). More interestingly, ruxolitinib significantly inhibited CAF-induced protection in MDA-MB-157 (Figure 5.7A, right panel, p<0.01), and a trend towards the same result was seen in MDA-MB-231 cells, although this inhibition was not significant (Figure 5.6A, right panel, p value 0.114). Analysis of interferon pathway activity by qPCR highlighted a trend that ruxolitinib inhibited CAF-induced interferon signaling in both MDA-MB-231 and MDA-MB-157 cells, as indicated by reduced levels of OAS1 and MX1, although this was not significant (Figure 5.6B and 5.7B, 231 p values 0.132 and 0.240 respectively, 157 p values 0.589 and 0.394 respectively). It was also evident from these data that inhibition of interferon signaling, in this case with ruxolitnib, inhibited the ability of CAF1 cells to enhance the clonogenic plating efficiency of MDA-MB-231 and MDA-MB-157 cells in the absence of epirubicin (Figure 5.6A and 5.7A).





Cultures of MDA-MB-231-GFP/luc cells were established in mono-culture or coculture with 20% immortalised breast CAF1s. Ruxolitinib was added to cultures at 10nM or vehicle control (A). Cultures were treated with 10nM epirubicin or untreated for 24h with ruxolitinib/vehicle readded. Cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are shown as numbers of colonies (left plots), or as counts normalised to the no treatment value for each individual fibroblast proportion (right plots), RNA was also extracted from the epithelial cells and equal amounts of RNA were used for qPCR (B) to determine relative expression of IFN-target genes OAS1 (left plot) and MX1 (right plot). qPCRs were normalised to ACTB. Data represent the mean(+/- SE) for n=3. Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value *<0.05.



Figure 5.7: Ruxolitinib significantly reduces CAF1-induced chemoresistance in MDA-MB-157 cells

Cultures of MDA-MB-157 cells were established in mono-culture or co-culture with 20% immortalised breast GFP-CAF1s. Ruxolitinib was added to cultures at 10nM or vehicle control (A). Cultures were treated with 10nM epirubicin or untreated for 24h with ruxolitinib/vehicle readded. Cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are shown as numbers of colonies (left plots), or as counts normalised to the no treatment value for each individual fibroblast proportion (right plots), RNA was also extracted from the epithelial cells and equal amounts of RNA were used for qPCR (B) to determine relative expression of IFN-target genes OAS1 (left plot) and MX1 (right plot). qPCRs were normalised to ACTB. Data represent the mean(+/- SE) for n=3. Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value *<0.05.

The same experiment as in section 5.3.4 was also performed using pCAFs derived from a TNBC, and MDA-MB-231-GFP/luc cells (Figure 5.8). Ruxolitinib significantly inhibited pCAF-induced chemoresistance in MDA-MB-231 cells (Figure 5.8A, right panel, p<0.05), and also inhibited pCAF-induced interferon signaling, although this was not significant (Figure 5.8B, p values 0.200 and 0.200 respectively).



Figure 5.8: Ruxolitinib significantly inhibits pCAF-induced chemoresistance in MDA-MB-231

Cultures of MDA-MB-231-GFP/luc cells were established in mono-culture or coculture with 20% primary breast CAFs. Ruxolitinib was added to cultures at 10nM or vehicle control (A). Cultures were treated with 10nM epirubicin or untreated for 24h with ruxolitinib/vehicle readded. Cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. Plates were then stained with crystal violet and individual colonies were counted. Data are shown as numbers of colonies (left plots), or as counts normalised to the no treatment value for each individual fibroblast proportion (right plots), RNA was also extracted from the epithelial cells and equal amounts of RNA were used for qPCR (B) to determine relative expression of IFN-target genes OAS1 (left plot) and MX1 (right plot). qPCRs were normalised to ACTB. Data represent the mean(+/- SE) for n=3. Two tailed Mann-Whitney U tests were carried out and significant differences shown. P value *<0.05.

5.3.6 MiR-155 inhibition reduces CAF1-induced chemoresistance of MDA-MB-231 cells

Inhibition of miR-155 is another potential mechanism of inhibiting CAF-induced interferon activity and thereby CAF-induced chemoresistance, based on its known role as a positive regulator of interferon signaling, and my previous observations that miR-155 was up-regulated by CAFs (section 4.3.2). Furthermore, I previously demonstrated that manipulation of miR-155 levels showed a trend to alter sensitivity of MDA-MB-231 and MDA-MB-157 cells to epirubicin treatment in the absence of CAFs (section 4.3.3/4).

A slightly adapted version of the clonogenic assay that has previously been used was carried out to determine whether inhibiting miR-155 in epithelial cells could inhibit CAF1-induced effects. Prior to seeding MDA-MB-231-GFP/luc cells in mono-culture or co-culture with CAF1 cells, MDA-MB-231-GFP/luc cells were transfected with either control inhibitor miRNAs or targeted miR-155 inhibitors. The protocol for the previous experiments was then followed for both assessment of clonogenic potential, and use of qPCR to monitor interferon pathway activity. In this case, additionally, qPCR was also used to monitor miR-155 expression.

Inhibition of miR-155 in the MDA-MB-231 cells significantly inhibited CAF1induced chemoresistance in the MDA-MB-231 cells when compared to control transfected cells (Figure 5.9A, right panel, p<0.05). qPCR analysis confirmed the expected CAF-induced up-regulation of miR-155, particularly when treated with epirubicin (p<0.0001), and the successful inhibition of miR-155 expression by the specific miR inhibitor (Figure 5.9B, p<0.05). Although not significant, qPCR also suggested that miR-155 inhibition in epirubicin-treated cultures reduced CAF-induced interferon signaling activation, as demonstrated by reduced levels of OAS1 and MX1 (Figure 5.9C p values 0.393 and 0.588 respectively).





MDA-MB-231-GFP/luc were transfected with control or miR155 inhibitor. Transfected MDA-MB-231-GFP/luc cells were plated in mono-culture or coculture with 20% immortalised breast CAF1s. MDA-MB-231-GFP/luc had been labelled with GFP, so the cell types could be differentiated by flow-cytometry. Co-cultures were treated with 10nM epirubicin or untreated for 24h. Co-cultures were then separated into pure epithelial populations by FACS. 500 epithelial cells were then seeded at low density and left for 14 days for colonies to form. RNA was extracted from remaining cells. Plates were then stained with crystal violet and individual colonies were counted. Data are normalised to the no treatment value for each individual fibroblast proportion and represent the means (+/- SE) of 3 independent experimental repeats (A). RNA was extracted from the MDA-MB-231 cells and relative expression of miR-155 (B) or OAS1 and MX1 (C) determined by qPCRs. Data represent means (+/- SE) of 3 independent experimental repeats. Two-tailed Mann-Whitney U tests were carried out with any significant differences shown between no treatment 0% to 20% fibroblasts, treated 0% to 20% fibroblasts and treated control 20% to miR155 20%. P values * <0.05, **<0.01 and ****<0.0001.

5.4 Discussion

The aims of the work in this chapter were to identify potential mechanisms to inhibit CAF-induced interferon signaling and to determine if this would reduce CAF-induced chemoresistance in TNBC cells.

5.4.1 Use of antibodies for inhibition of interferon signaling

Antibodies provide successful current, and promising future therapeutic possibilities through their ability to bind selectively to cancer cells or to soluble targets, and thereby influence cancer cell function. Use of blocking antibodies to inhibit the interferon signaling pathway could therefore be a viable option in preventing interferon activation and chemoresistance. This could be done by either blocking soluble interferon ligands or by blocking interferon receptors. The use of blocking antibodies to block soluble ligands represents a viable option as evidenced by the fact that an antibody drug has been successful in sequestering VEGF-A. Bevacizumab, which binds and sequesters VEGF-A, inhibits angiogenesis, which in turn inhibits tumour growth and proliferation (Ferrara et al, 2004). *In vitro*, blocking of soluble interferons has also been successful in targeting soluble IFN- β (Hosein et al, 2015). Although soluble ligand blocking can be extremely effective and could have been used to block soluble interferons, this approach was not used because of the diversity of the interferon ligands, which might require multiple different antibodies.

Antibodies that bind to and block surface receptors have proved succesful in the treatment of cancer, and this was the approach I chose to use in this chapter. Blocking antibodies act by neutralizing the effects of oncogenic pathways and are extremely effective treatment options in breast cancer with various monoclonal antibodies currently being used in the clinic (Simpson and Caballero, 2014). Cetuximab is used in breast cancer treatment to reduce activation of the EGFR pathway by binding to the extracellular domain of EGFR. Binding of cetuximab therefore prevents EGF binding to its receptor and transduction of the signaling cascade. In addition, cetuximab can lead to receptor internalization and degradation of EGFR and down-regulation of EGFR expression (Herbst et al, 2002). Both trastuzumab and pertuzumab are targeted treatments available for HER positive breast cancer patients. Trastuzumab acts by binding to the extracellular domain of HER2 receptors with high affinity (Vu et al, 2012). Pertuzumab binds to the extracellular dimerization domain II of HER2 receptors with pertuzumab and trastuzumab able to work in syngery with one another as they bind to different epitopes (Franklin et al, 2004). In addition, pertuzumab and trastuzumab can be given in combination with docetaxel (Baselga et al, 2010). The current use of blocking antibodies targeting cell surface receptors as effective breast cancer therapeutics was a determining factor for their use in this chapter to block interferon signaling.

A survey of relevant literature identified 2 previously published studies in which IFN blocking antibodies were used in the context of breast cancer, one of which even involved the presence of CAFs in co-culture with breast cancer cells. Therefore, these studies are highly-related to work covered within this chapter. Blocking antibodies have been used to bind to the Type I interferon receptor (IFNAR1, clone MMHAR2) to prevent activation of the intra-cellular signaling cascade in ER-positive aromatase inhibitor resistant breast cancer cells (Choi et al, 2015). The Type I blocking antibody clone used in this study is the same clone that I have used this chapter, although it is interesting to note that these authors used western blot analysis of expression of the ISG IFITM1 (the protein on which their study was focused) to confirm interferon signaling blockade, rather than my qPCR analysis of OAS1 and MX1. In their study, the Type I blocking antibody or IFITM1 knockdown both separately reduced cell proliferation and sensitized the resistant cells to estrogen deprivation (Choi et al, 2015). Type I blocking antibodies targeted to soluble IFN- β have also been used in the context of CAFs and breast cancer. Using gene expression profiling, a subgroup of interferon positive CAFs, which expressed ISGs and released IFN-β, were identified. Indirect co-culture of the interferon positive CAFs with MCF-7 cells promoted MCF-7 cell growth, with the blocking antibody inhibiting

this effect. However, confirmation of decreased interferon signaling in the target MCF-7 cells was not performed (Hosein et al, 2015). Targetting of soluble IFN- β differs from targeting the Type I receptor, as above and in my work, and – critically - does not target potential roles of other Type I interferons.

In the studies highlighted above by Choi et al and Hosein et al, the authors did not attempt to block both Type I and Type II signaling in combination, and therefore my work with combining antibodies to block both pathways appears to be unique in the context of breast cancer assays in tissue culture. Interestingly, the combination of both Type I and II did not have additive effects in terms of inhibiting chemoresistance (Figure 5.1), and curiously appeared to work less well in combination than individually. It is plausible that the addition of both antibodies caused some steric hindrance of the binding of each antibody to its target receptor. Therefore, this combination was not explored further. Furthermore, CAF-induced activation of interferon signaling and CAF-induced chemo-protection in MDA-MB-157 was not inhibited by Type II blocking approaches (Figure 5.4) suggesting that Type I signaling may be the dominant influence. However, it is possible that the dose of the Type II neutralization antibody was not enough in this cell line, although it proved sufficient in MDA-MB-231 cells, or it may be that the Type I activation is so dominant in the MDA-MB-157 cell line that the contribution of Type II is relatively unimportant. It is worth highlighting that MDA-MB-157 cells certainly can respond to Type II interferons, as the presence of recombinant IFN-y in section 4.3.5 and Figure 4.5 caused a (slight) increase in chemoresistance when used on its own and significantly increased chemoresistance in combination with IFN-a.

Based on *in vitro* data from this chapter, use of Type I neutralizing antibodies in combination with epirubicin has potential as a treatment option for TNBC patients with high proportions of CAFs within their tumours, as a method for reducing CAF-induced chemoresistance and therefore making chemotherapy more effective. However, a concern would be that use of IFN blocking antibodies could potentially inhibit tumour suppressing immune cell responses, given the immune cell roles of interferons previously highlighted in section 1.14.1. Nevertheless, interferon blockade during chemotherapy, when the

immune system is suppressed in any event, remains worthy of further exploration.

5.4.2 Use of small molecules for inhibition of interferon signaling

As well as the use of antibodies to inhibit interferon function at the level of ligand or receptor, small molecule inhibitors of the intra-cellular JAK/STAT cascade have also been developed. The main mechanism that these drugs employ is inhibition of the JAKs; some JAK inhibitors have selective potency for both the predominant JAKs (JAK1 and JAK2), whereas other drugs only target one particular JAK (JAK1, 2 or 3). One JAK inhibitor, Tofacitinib, can target all 3 JAKs (JAK1, 2 and 3) (Thomas et al, 2015).

JAK inhibitors are undergoing clinical trials to treat a variety of different diseases, with some inhibitors approved for clinical use in selected diseases. Tofacitinib, a JAK1, 2 and 3 inhibitor, is used to treat patients with rheumatoid arthritis, skin conditions including psoriasis, and irritable bowel syndrome (Miklossy et al, 2013). A further JAK inhibitor, ruxolitinib, has been approved for use myelofibrosis, but of more relevance to my work, ruxolitinib is also being trialled for use in treatment of different forms of leukemia: acute myeloid, chronic myeloid and acute lymphocytic leukemia (Furumoto et al, 2013). However, the use of JAK inhibitors in solid tumours is less advanced. Different tumours have been implicated as potential targets of JAK inhibitors, as a result of STAT up-regulation being associated with recurrences and reduced overall survival in NSCLC, prostate cancer and melanoma (Xu et al, 2014, Messina et al, 2008, Mirtti et al, 2013), but JAK inhibitors have not yet been used in these solid tumours.

Clinical trials have been carried out using JAK inhibitors in patients with metastatic breast cancer. Patients, in this clinical trial, had been heavily pretreated with up to 48% having received 3 or more regimens of chemotherapy in the metastatic setting (Stover et al, 2018). The use of Ruxolitinib, a JAK1/2 selective inhibitor, is currently being explored as a potential treatment option for these patients due to high expression levels of STAT3 in metastatic disease. Although up-regulation of STAT3 was observed in metastatic TNBC patients, no responses were observed in 21 out of 23 patients following ruxolitinib treatment despite on-target activity. The use of ruxolitinib as a single agent to improve treatment options for metastatic TNBC patients, therefore, did not meet the primary efficacy endpoint in this study (Stover et al, 2018).

Interestingly, ruxolitinib showed potential in my experiments, which are also in the context of TNBC, albeit with major differences in essentially all other ways. In section 5.3.4 and 5.3.5, ruxolitinib reduced CAF-induced chemoresistance in both MDA-MB-231 and MDA-MB-157 cells. These data support the idea that ruxolitinib may have some efficacy in combination with epirubicin, by reducing CAF-induced chemoresistance. The clinical trial carried out in metastatic breast cancer highlighted above used ruxolitinb as a single agent and it seems until only recently that studies have been reported regarding clinical use in combination with cytotoxic chemotherapy. Ruxolitinib has been used in combination with capecitabine in HER2 negative breast cancer, with an overall response rate of 28.9% for the combination in comparison to 13.7% in the capecitabine alone group (O'Shaughnessy et al, 2018); this appears to support the combination of ruxolitinib and traditional cytotoxics. Clinical trials of ruxolitinib in breast cancer are ongoing with ruxolitinib used in combination with trastuzumab in metastatic HER2 positive breast cancer, without the use of cytoxic chemotherapy (Cigler et al, 2018), and in stage III triple negative inflammatory breast cancer in combination with paclitaxel followed by doxorubicin/cyclophosphamide. The results of these studies are still being collated (Overmoyer et al, 2018).

5.4.3 MiRNAs or miRNA inhibitors (specifically miR-155 inhibitor) as viable therapeutic options.

In recent years, work has been carried out to develop RNAs, and particularly miRNAs, as potential cancer therapeutics (Mehrgou and Akouchekian, 2017). The advantages of this approach are the low cost of synthesis of miRNAs as well as their small molecular size and the ease of design. MiRNA therapeutics act by either mimicking tumour suppressor miRNAs or by inhibiting oncogenic

miRNAs. Due to the ease of production of miRNA therapeutics, the use of miRNAs and miRNA inhibitors are a potential therapeutic option (Yeung and Jeang, 2011), although they have not yet reached routine clinical practice. This is of potential interest here, since I have identified inhibitors of miR-155 as a potential therapeutic in combination with epirubicin (Figure 5.9).

Various different miRNAs are being tested *in vitro* and *in vivo* for their potential for clinical use. For example in breast cancer, both miR-132 and miR-10b have oncogenic properties aiding in cancer progression, therefore inhibitors of miRNAs have been tested in vivo to try to target these miRNAs and prevent their function (Anand et al, 2010). Anti-miR-21 has also been investigated as a potential treatment option in human breast cancer. Anti-miR-21 in MCF-7 and MDA-MB-231 cells in vitro and in vivo inhibited growth as well as migration of breast cancer cell (Yan et al, 2011). I investigated the results of inhibiting miR-155 (section 4.3.3/4 and 5.3.6), and inhibition of this specific miRNA has been previously investigated in the context of breast cancer. MiR-155 antisense oligonucleotides (mechanistically the same as the miR-155 inhibitor used in this thesis) were shown to reduce the viability of MDA-MB-157 cells in vitro, and to reduce cell growth of xenograft tumours (Zheng et al, 2013), although this function was not linked to interferon activity, as in my work. This study, and my own, do support the concept that inhibition of miR-155 could have therapeutic value in the context of breast cancer.

Due to the ease of production of miRNA therapeutics, the main area for work and a key difficulty with miR-based therapeutics is the delivery method. *In vitro* use of miRNA therapeutics is relatively simple with the ease of transfection into cells. However, the ability to target miRNA therapeutic entry into only cancer cells in cancer patients (without targeting normal cells) is problematic. Neverthesless, one miRNA-based therapeutic, miravirsen - an inhibitor of miR-122, is currently in clinical trials (van der Rhee et al, 2016). Miravirsen is designed to work without specific targeting in the belief that it will be taken up sufficiently by cancer cells for efficacy when delivered systemically, without excess toxicity from non-target tissues. While miRNA therapeutics clearly have potential - and specifically targeting miR-155 in the context of this work - efficient delivery remains a barrier to future development.

5.5 Conclusion and aspects of future work

It is clear that interferon signaling can be inhibited in multiple different ways, from targeting the receptor to prevent interferon binding and activating signaling, to targeting the JAK/STAT pathway to inhibit the intra-cellular signaling cascade, or inhibiting formation of the correct transcriptional complex by inhibiting miR-155. Inhibition of interferon signaling using any of these mechanisms can reduce CAF-induced chemoresistance in TNBC cells, at least *in vitro*. To identify if these inhibitory mechanisms could be used in combination with chemotherapeutic agents as a possible treatment option for patients, *in vivo* studies will be required.

Chapter 6 - What are the cellular mechanisms of cancer-associated fibroblast-induced chemoresistance in breast cancer cells?

6.1 Abstract

CAF-induced chemoresistance in TNBC was identified in previous chapters. In this chapter, I have attempted to investigate the direct effectors that drive this chemoresistance down-stream of interferon signaling. I have also explored the nature of the CAF-induced increase in clonogenicity observed in the absence of chemotherapy.

The first potential mechanism of CAF-induced chemoresistance to be tested was altered drug loading in the epithelial cells. Using fluorescent-assessment of epirubicin loading, it was shown – surprisingly – that CAFs-induced *increased* drug uptake in MDA-MB-231 cells, rather the decrease that would be expected if this were the mechanism of protection. The second potential mechanism of CAF-induced chemoresistance to be tested was alteration of the cell cycle in the epithelial cells. CAFs did not induce significant changes in the proportions of MDA-MB-231 cells in each phase of the cell cycle, although there was a trend for CAFs to induce reduced S-phase representation, which is potentially compatible with this as a resistance mechanism. Finally, the CAF-induced increase in clonogenicity was assessed; this was shown to be associated with an increase in anoikis resistance in both MDA-MB-231 and MDA-MB-157 cells.

I concluded that CAF-induced changes in epithelial cell cycle profiles, or in sensitivity to apoptotic stimuli remain plausible potential mechanisms by which CAFs induce chemoresistance in epithelial breast cancer cells.

6.2 Introduction

I have shown in previous chapters that CAFs induce chemoresistance in TNBC cells, probably through paracrine signaling, leading to up-regulation of the interferon signaling pathway. However, the molecular mechanisms that allow activated interferon signaling to result in chemoresistance are not understood.

Interferon signaling results in activation of the JAK/STAT signaling cascade as highlighted in section 1.12.1 and 1.12.2. It seems reasonable to assume that the mediators of the chemoresistance seen in my experiments must be downstream of JAK activation. A number of studies have investigated the downstream targets of interferon and JAK activation in various cell types. One of the major sets of kinases that have been shown to induce IFN stimulated responses are the MAP kinases. p38 MAP kinases have been shown to be activated by the Type I IFN receptor (Uddin, 1999). ERK is another MAP kinase family member activated by IFNs, although ERK appears to be activated by both Type I and Type II IFN receptors. MEK/ERK activation is driven by JAK1 activation in response to IFNs. JAK1 can activate Raf1 to enable downstream effectors of the MEK/ERK pathway (Platanias, 2005). Jnk kinase, a further member of the MAP kinase family, has also been reportedly activated by both Type I and Type II IFNs to mediate IFN responses (Zhao et al, 2011). Another alternative downstream effector of IFN signaling is the mTOR pathway. Activation of both Type I and II receptors led to activation of upstream effectors of mTOR, which have been identified as PI3K and Akt (Kaur et al, 2008, Kaur et al, 2008).

All of these potentially activated signaling pathways remain substantially upstream of the actual effector molecules or functions that could directly lead to chemoresistance. Multiple different direct chemoresistance mechanisms have been explored previously to gain increased mechanistic understanding, and in efforts to use these insights to bypass their potential influences to make chemotherapy treatment more effective (O'Reilly et al, 2015). One such mechanism that has been explored is the role of the xenobiotic membrane transporters of the ATP-binding cassette (ABC) transporter family. The main ABC transporters that have been investigated in TNBC are multi-drug resistant protein 1 (MRP1, ABCC1), breast cancer resistance protein (BCRP, ABCG2) and P-glycoprotein (MDR1, ABCB1) (Kovalev et al, 2013). High expression and activity of these transporters has been shown to contribute to chemoresistance in TNBC by reducing intra-cellular levels of the chemotherapeutic agents. BCRP, in particular, has been shown to reduce intra-cellular levels of doxorubicin in TNBC cells (Chen et al, 2014), while MDR1 acts on various different classes of chemotherapeutics, including paclitaxel, doxorubicin and epirubicin (Wind et al, 2011).

Another direct effector mechanism of chemoresistance in cancers is increased resistance to apoptosis. In a large number of cancers, apoptosis signaling is dysregulated leading to evasion of cell death and enhanced survival of cancer cells. It has been shown in multiple different types of cancer such as pancreatic, ovarian and lung that there is up-regulation of anti-apoptotic factors (Mohammed et al, 2015). For example, over-expression of Bcl-2, or other family members such as Bcl-xl or Mcl-1 has been shown to be associated with chemoresistance in cancers (Azmi et al, 2009, Quin et al, 2011).

Finally, another potential mechanism of chemoresistance is the induction of changes to cell cycle. A large number of chemotherapeutic drugs' main mechanism of action occurs during specific phases of the cell cycle, for example at mitosis (taxanes) or during S phase (anthracyclines) (Hassan et al, 2010). Therefore, the proportion of cells in each cell cycle phase, and the overall proliferation rate of cells, can be key factors in determining chemotherapeutic function. Studies in early breast cancer have shown patients with high proliferation rates (determined by high Ki67 expression) responded better to NAC and achieved a pCR of 40% in comparison to lower proliferation rates (determined by low Ki67 expression), where pCR was 19% with patients more resistant to chemotherapy (Alba et al, 2015).

Therefore, based on some of the acknowledged chemoresistance mechanisms, in this chapter I have investigated potential mechanisms that could be directly responsible for the CAF-induced chemoresistance I have seen in selected TNBC cell lines.

6.3 Results

6.3.1 CAF1 cells increase intra-cellular epirubicin levels in co-cultured MDA-MB-231 cells

A possible mechanism of CAF-induced chemoresistance is reduced drug uptake by the epithelial cells, perhaps mediated by enhanced expression or activity of xenobiotic transporters such as the ABC transporters. In order to examine this, drug loading assays were performed taking advantage of the fact that epirubicin is itself fluorescent.

MDA-MB-231-GFP/luc cells were either mono-cultured or co-cultured with 20% CAF1 cells for 24h as previously. As a further control CAF1 cells were also mono-cultured. Cultures were treated with epirubicin for 24h and then analysed by flow cytometry to quantify intra-cellular epirubicin loading (Figure 6.1).

Surprisingly, CAF1s induced MDA-MB-231 cells to contain significantly more epirubicin compared to the levels in MDA-MB-231 mono-culture (compare dark blue bars in Figure 6.1, p<0.01). Conversely, uptake of epirubicin appeared to decrease in the CAF1s from the co-culture with MDA-MB-231 cells, compared to CAF mono-cultures (compare light blue bars in Figure 6.1), although this difference was not significant (p value 0.224). I concluded that CAFs were not protecting TNBC cells from epirubicin by reducing the loading of the drug into the cells. In fact, quite the reverse appeared to be happening – CAFs induced additional drug loading in the cancer cells.



Figure 6.1: CAF1s enhance intra-cellular epirubicin levels in MDA-MB-231 Mono-cultures of MDA-MB-231-GFP/luc cells or CAF1 cells, and co-cultures of MDA-MB-231-GFP/luc cells and 20% CAF1 cells were established. Cultures were treated with 1µM epirubicin for 24h. Cells were then analysed by flow cytometry, gating on viable cells (using FSC v SSC) and on MDA-MB-231-GFP/luc cells or CAF1 cells separately (on the basis of GFP expression). Epirubicin loading was determined by red fluorescence as the median of BL-3 fluorescence. Co-culture values (denoted "20%") were normalised to the matched mono-cultures values. Data represent the means (+/- SE) of 3 independent biological replicates. Two-tailed Mann-Whitney U tests were carried out and significant differences to corresponding cell type are shown. P values **<0.01.

6.3.2 CAF1s may induce changes in cell cycle progression

CAFs are known to influence growth and metastatic properties of cancer cells (see section 1.9). Therefore, it was possible that the presence of CAFs was having an impact on the cell cycle of the co-cultured TNBC epithelial cells. A decrease in epithelial proliferation might provide protection from chemotherapy, since cytotoxic chemotherapy agents target proliferating cells. It should, however, be noted that the reported influence of CAFs on epithelial proliferation is generally that CAFs stimulate increases in proliferation rather than the reverse (Verghese et al, 2013). In order to examine this, cell cycle profiling of mono- or co-cultures of MDA-MB-231 and CAF1 cells was undertaken.

MDA-MB-231-GFP/luc cells were either mono-cultured or again co-cultured with CAF1 cells, as usual, and treated with epirubicin. After 24h, cells were treated with a fluorescent dye to allow quantification of cellular DNA and therefore measurement of the proportion of cells in G1, S or G2 phase of the cell cycle. Cell cycle profiles were determined by flow-cytometry in the GFP positive MDA-MB-231 cells. Following the production of the cell cycle profile, the proportion of cells in each stage of the cell cycle was determined using ModFit cell cycle software, and proportions of cells in each stage were determined (Figure 6.2).

MDA-MB-231 in co-culture with 20% CAF1 cells were compared with those in mono-culture; there was an increase in the co-culture of the proportion of cells in G0/G1 and in G2, and a corresponding decrease in the proportion of cells in S phase (compare the red bars to the matched purple bars in Figure 6.2), although none of these differences were statistically significant (p values 0.400, 0.200 and 0.200 respectively). I concluded that although CAF1 cells did not induce significant changes in epithelial cell cycle profiles, a potential trend for cell cycle changes was apparent.



Figure 6.2: CAF1s do not induce significant differences in epithelial cell cycle profile

Mono-cultures of MDA-MB-231-GFP/luc cells and co-cultures of MDA-MB-231-GFP/luc cells and 20% CAF1 cells were established. Cultures were treated with 10nM epirubicin for 24h. Dye cycle violet was then added to cultures for 30 min prior to flow cytometry. Cells were then analysed by flow cytometry gating on viable cells (using FSC v SSC) followed by gating on dyecycle loaded viable cells (VL1 vs FSC). Singlet Dyecycle violet postive cells were then gated (VL1-H vs VL1-A) followed by identification of Dyecycle violet MDA-MB-231-GFP/luc cells in a population with CAF1 (VL1 vs BL1). VL1 was then counted to produce a cell cycle profile. Using ModFit software the percentage of cells in the different stages of the cell cycle was determined. Data represent the means (+/- SE) of 3 independent experimental repeats. Two-tailed Mann-Whitney U tests were carried out with no significant differences observed between 0% and 20% cultures

6.3.3 CAF1s significantly increase anoikis resistance in MDA-MB-231 and MDA-MB-157

In Chapter 3, I found that co-culture of the TNBC cell lines MDA-MB-231 or MDA-MB-157 with CAFs enhanced their replating efficiency after cell sorting (and in the absence of chemotherapy agents) (section 3.2.2; Figure 3.3 and 3.4). One hypothesis based on this observation was that the CAFs may induce resistance to the pro-apoptotic effects of detachment from the substrate. Apoptosis induced by detachment from the substrate – something to which most epithelial cells are susceptible – is referred to as anoikis (Chiarugi et al, 2008). CAF-induced anoikis resistance would be of interest since it would explain the enhanced plating efficiency mentioned above, but it could also potentially explain the chemoresistance if anoikis resistance were a reflection of a more general resistance to apoptosis, since it is well known that chemotherapeutics kill cells in part by inducing apoptosis (Kaufmann et al, 2000). Therefore, I carried out assays to test whether CAFs were indeed inducing anoikis-resistance in co-cultured TNBC cancer cells, using an established anoikis assay.

MDA-MB-231-GFP/luc or MDA-MB-157 cells were cultured alone or with increasing proportions of breast fibroblasts. In the case of MDA-MB-231-GFP/luc cells, either NF1s or CAF1s were used, while for MDA-MB-157 only GFP-CAF1s were used. Cultures were incubated for 48h. Epithelial cells were then purified by FACS on the basis of GFP expression. As previously, even cultures without fibroblasts (ie epithelial cells only) were sorted, in order to allow for properly-controlled comparisons to be made with epithelial cells sorted from the co-cultures. Standard numbers of collected epithelial cells were then suspended in medium in non-adherent plates for 36h. MTT assays were performed to quantify cells remaining viable under these conditions (Figure 6.3).

Co-culture with CAF1s induced dose-dependent and significant increases in survival in both MDA-MB-231 (p<0.05) and MDA-MB-157 (p<0.05) cells (Figure 6.3 B and C). By contrast, when MDA-MB-231 cells were co-cultured with NF1s,

there was no increase in survival. I concluded that CAFs, but not NFs, can induce anoikis-resistance in TNBC cell lines.



Figure 6.3: CAF1s enhance anoikis resistance in MDA-MB-231 and -157 cells Co-cultures of MDA-MB-231-GFP/luc (A and B) or -157 (C) cells were established with various proportions of either immortalised normal breast fibroblasts (NF1) or immortalised breast CAFs (CAF1). Either fibroblasts or epithelial cells had been labelled with GFP, so the cell types could be differentiated by flow-cytometry. Co-cultures were cultured for 48h with a media change after 24h. Co-cultures were then separated into pure epithelial populations by FACS. 150,000 epithelial cells were then seeded on a low attachment plates for 36h and cells remaining viable after this were quantified using MTT assays. Survival is expressed relative to the epithelial mono-culture (0%) from each experimental repeat. Data represent the means (+/- SE) of 3 independent biological repeats. Linear regression analysis was carried out on no treatment for colony counts and normalised treated data with significant differences in the overall trend across the fibroblast proportions calculated. P values * <0.05

6.4 Discussion

The aims of the work in this chapter were to identify potential effector mechanisms of CAF-induced chemoresistance.

6.4.1 Anoikis resistance

In chapter 3 (section 3.3.2), I identified that co-culture with CAFs induced increases in epithelial plating efficiency. During this experiment, the cells remain in suspension for an extended period of time prior to replating for the clonogenic assays. Epithelial cells in suspension can respond by undergoing programmed cell death, which is termed anoikis in this context.

Anoikis occurs when epithelial cells become detached from the extracellular matrix (ECM), and it acts as a defense system to prevent any aberrant growth of cells in the absence of adhesion as well as preventing detached cells from colonizing inappropriate sites (Frisch and Francis, 1994). Commonly, when cells remain attached to the ECM, they are protected from anoikis, which is driven by anti-apoptotic and pro-survival signals from integrin signaling (Frisch and Ruoslahti, 1997). Even during migration of cells, integrin attachment to focal contacts can again activate signaling to produce pro-survival signals to prevent anoikis (Baum et al, 2008). Anoikis resistance is thought to be an essential prerequisite for cancer progression and metastasis to distant organs for the formation of secondary tumours (Kim et al, 2011). It is thought that anoikis resistance in cancer cells can be achieved through a number of different mechanisms. Integrins can be dysregulated by expression changes occurring in cancer cells, for example more metastatic breast cancer cell lines (MDA-MB-435 and MDA-MB-231) expressed higher levels of integrin β 5 or integrin β 3 (Taherian et al, 2011). Increased expression of $\alpha 6$ integrin is also related to poorer survival of breast cancer patients (Friedrichs et al, 1995). Increased expression of integrins is functionally important to enable enhanced integrin signaling and activation of MEK and FAK for survival. Increased expression of integrins can also enable breast cancers to adapt to their metastatic site and enable colonization of secondary tumours (Guo et al, 2004). Another potential

mechanism that cancer cells have exploited to gain anoikis resistance is EMT, as it is thought that anoikis resistance is associated with cancer cells gaining a mesenchymal phenotype (Cao et al, 2016). A stimulus such as the presence of stromal cells, oncogene activation through epigenetic changes, or hypoxia can trigger EMT induction. Stromal cells, in particular CAFs, can induce EMT in breast cancer cells. Breast cancer MCF-7 cells cultured with CAFs showed increased expression of vimentin and lower expression of E-cadherin, which are known expression changes that occur in EMT (Soon et al, 2013). CAF-conditioned media can also transform breast cancer cell lines (MCF-7, T47D and MDA-MB-231) into a more aggressive and invasive phenotype through secretion of TGF- β from the CAFs, something that can be reversed by blocking TGF- β signaling (Yu et al, 2014). EMT induction in cancer cells leads to expression of pro-survival and anti-apoptotic genes to enable anoikis resistance (Chiarugi et el, 2008).

Finally, another mechanism involved in anoikis resistance is constitutive activation of signaling pathways involved in anti-apoptotic pathways, particularly the PI3K/Akt pathway (Davies et al, 1999). Constitutive activation of the PI3K/Akt pathway can occur in multiple different ways, such as overexpression of receptor tyrosine kinase receptors, constitutive activation of Ras (which is common in many cancers), or loss of PTEN expression (which is common in TNBC) (Tokunaga et al, 2008). Activation of PI3K and the conversion of PIP₂ to PIP₃ enables activation of Akt, which is enhanced by the loss of PTEN (which would otherwise convert PIP₃ back to PIP₂) (Vitolo et al, 2009). Activation of Akt leads to decreases in expression of pro-apoptotic Bid, Bim, Bad and Bax, and increases in expression of anti-apoptotic genes Bcl-2 and IKK. Src kinases have also been implicated in anoikis resistance. Sustained activation of Src kinases has been shown to phosphorylate focal adhesion kinase (FAK), which recruits and activates PI3K and thereby the anti-apoptotic pathway described above leading to inhibition of Bad and reduced caspase-2, -3, -8 and -9 function (Bouchard et al., 2007). Finally, growth factor signaling and activation of MAPK signaling leads to enhancement of cell survival as MAPK signaling induces degradation of the pro-apoptotic protein Bim (Reginato et al, 2003). Importantly, CAFs have also been shown to be involved in anoikis resistance. One study

has shown that CAFs can secrete elevated levels of insulin-like growth factor binding proteins (IGFBP) that are necessary for inhibition of anoikis in breast cancer cells, through stabilization of the anti-apoptotic protein Mcl-1 (Weigel et al, 2014), although in my experiments this may not be relevant since the anoikis resistance I see appears to be dependent on interferon signaling.

Interestingly, miRNAs have also been implicated in anoikis resistance as they have been shown to control expression of pro-apoptotic genes in breast cancer. For example miR-19, miR-21, miR-27, miR-29, miR-155, miR-181, miR-206, miR-221/222 and miR-375 can target and inhibit the function of pro-apoptotic genes such as Bim, Bax and Puma (Halytskiy, 2017).

From experiments carried out in Chapter 6, there was no direct data concerning whether CAFs led to expression/activity changes in any of these anoikis resistant pathways highlighted above. However, from gene level microarray data produced in Chapter 4, pathways known to be involved in anoikis resistance were not identified as significantly altered by CAFs in the presence of epirubicin. The use of IFN blocking antibodies in Chapter 5 reduced interferoninduction and halted the anoikis phenotype observed in colony count data sets (change in plating efficiency). Therefore, it is likely that interferons are up stream of CAF-induced anoikis resistance rather than other secretory molecules such as IGFBP. Interestingly from the miRNA screen in chapter 4, miR-155 was upregulated in the presence of CAFs. MiR-155 has already been implicated as a mediator of anoikis resistance by inhibiting the function of pro-apoptotic genes (Bim, Bax, Puma and caspase-3) discussed previously, and could therefore play a role in the CAF-induced anoikis resistance in TNBC observed in this chapter. In addition, given miR-155's role in enhancing interferon signaling, it is plausible that interferon signaling and its downstream targets (PI3K pathway), as well as the direct targets of miR-155, could therefore be the best candidate mediators of anoikis resistance.

From this chapter, it is clear that CAF-induced activation of interferon signaling can induce anoikis resistance in TNBC cells. It is plausible that clinically, if patients present with a higher proportion of CAFs in the primary tumour, cells could become anoikis resistant and therefore more likely to survive in the circulation and metastasize to form secondary tumours.

6.4.2 Cell cycle changes and chemoresistance

My data suggest that CAFs do not induce significant changes in epithelial cell cycle profiles (section 6.3.2). However, these data contain some variance between individual repeats (n=3) and it is tempting to speculate that if the experiment could be made more reproducible, then significant findings may be revealed. The reason to discuss cell cycle arrest and chemoresistance in detail is that the results displayed a trend for reduced proliferation as a potential mechanism for chemoresistance.

Interestingly, the role of CAFs on the cell cycle has been investigated in cancer cells previously. As highlighted in section 1.10, CAFs can enhance the growth of breast cancer cells. The opposite influence has, however, been seen in other contexts. For examples, dermal fibroblasts have been shown to induce cell cycle arrest (in G0/G1), and block epithelial-mesenchymal transition in melanoma (Zhou et al, 2016) while CAFs in pancreatic cancer repressed cell growth in clonogenic assays and reduced the malignancy of pancreatic cells (Ding et al, 2018). However, CAF-induced cell cycle arrest has not directly been implied in breast cancer cells.

In cell cycle analysis performed in section 6.3.2, the cells were also treated with epirubicin, as has been standard in my chemoresistance studies. Previous studies have shown epirubicin itself induced cell cycle arrest occurring in G0/G1. Surprisingly, cell cycle arrest resulting from treatment with a number of agents all of which act in totally different ways, namely - quercetin, tamoxifen and epirubicin, has been attributed to changes in the expression levels of the same single molecule - p21. p21 is a CDK-inhibitor and can inhibit all stages of the cell cycle from inhibiting cyclin E–CDK2 and cyclin D-CDK4/6 activation in G1 to inhibiting Cdc2 activation in G2 (Karimian et al, 2016). Increased p21 levels have been shown to inhibit proliferation of breast cancer cells (Xiong et al, 2016), with down-regulation of p21 shown to remove tamoxifen induced cell

cycle arrest (Cariou et al, 2000). Critically, Type I and Type II interferons have been shown to regulate expression of p21 (Sangfelt et al, 2000). Type I interferons have been shown to induce cell cycle arrest in G0/G1. Increased levels of p21 were identified following IFN treatment (Subramaniam et al, 1997). More recently, Type I IFNs have been found to activate the p21 gene promoter and subsequent expression of p21. Both IFN- α 2b and IFN- β were able to enable p21 expression, with p21 down-regulation reversing cell cycle arrest (Katayama et al, 2007). Type II IFN- γ can also induce p21 expression in MCF-7, however in this study p21 expression was apparently not detectable in MDA-MB-231 cells (Gooch et al, 2000).

Therefore, three observations can be brought together to suggest a mechanistic model. Firstly, interferons have previously been shown to induce cell cycle arrest, as described above. Secondly, I have shown CAFs to induce interferon-dependent chemoresistance in co-cultured epithelial cells (Chapter 5). Thirdly, my data suggest that CAFs may induce reduced proliferation in these co-cultured epithelial cells (albeit this was not significant; Figure 6.2), which would – if true – potentially lead to chemoresistance. Thus, a plausible mechanistic pathway would be that CAFs induce chemo-protective cell cycle changes in co-cultured epithelial cells, via interferon signaling.

Chapter 7- Discussion

The aims of the work in this thesis were to identify if CAFs play a role in chemoresistance in TNBC, to determine the pathways responsible for CAFmediated chemoresistance and inhibit these as potential therapeutic options to improve chemotherapy response, and finally to investigate the effector mechanisms that enable CAF-induced chemoresistance.

7.1 CAF-induced therapy resistance in breast cancer

CAFs are known to enhance breast cancer progression, as highlighted in section 1.10. However, here, the primary goal was understanding whether CAFs are also involved in chemoresistance in breast cancer, focusing in particular on TNBC.

CAFs have previously been implicated in chemoresistance pathways in cancer, however in breast cancer, CAFs' role in chemoresistance is still poorly understood, with only a few mediators of chemoresistance identified. Breast CAFs have previously been shown to modulate tamoxifen resistance in ER-positive breast cancers, through activation of the PI3K and MAPK pathways (Shekhar et al, 2007). Co-culture experiments involving human skin fibroblasts and the ER-positive breast cancer line MCF-7 cells led to the identification of fibroblast-induced resistance to tamoxifen and fulvestrant in the MCF-7 cells (Martinez-Outschoorn et al, 2011). Although co-culture of fibroblasts and ER-positive breast cancer cells induced therapy resistance, the use of human skin fibroblast and not breast CAFs is a weakness in this work, whereas in this thesis both normal breast fibroblasts and breast CAFs are used.

CAF-secreted soluble factors have also been implicated in therapy-resistance in TNBC, the subtype on which I have focused. HGF, present in conditioned media from reduction mammoplasty fibroblasts, activated Met signaling to promote resistance in SUM102 and SUM149 TNBC cells to the EGFR inhibitor gefitinib (Mueller et al, 2012). Although this is a prime example of fibroblast driven resistance in TNBC, both the cell lines used and the class of drug are

dissimilar to my work, therefore there is no reason to expect the resistance pathways to be similar.

However, two studies have been published that are similar to my work in some main respects, in that they have used similar techniques (cell lines, *in vitro*, in co-culture), cell lines (MDA-MB-231, MDA-MB-468 and MDA-MB-157 cells) and therapeutic agents (use of cytotoxic chemotherapy, albeit not the same drug) as in my investigations. Importantly, they have also made highly-related findings about the role of CAFs in therapy resistance. Therefore, I have described below their findings in some detail, and have then compared these systematically with my own (below, and Table 7.1).

One study showed that CXCL12, a chemokine often regarded as a hallmark of the CAF phenotype, can enable paclitaxel resistance in MDA-MB-231/CAF coculture spheroids in comparison to MDA-MB-231/normal human mammary fibroblasts. The resistance involved activation of MAPK and PI3K pathways. down-stream of the CXCR4 receptor (Ham et al, 2018). Only one TNBC cell line was used in this study (MDA-MB-231), but the normal breast fibroblasts and CAFs used were extracted from breast cancer patients, representing an advance on studies using unrelated sources of fibroblasts. In order to inhibit CXCR4 receptor signaling and remove chemoresistance, an inhibitor of CXRC4 was used. In the second study, MRC5 normal human fibroblasts (isolated from lung tissue) were grown in co-culture with TNBC cell lines MDA-MB-231, MDA-MB-468 and MDA-MB-157 and cells were treated with radiation and the gamma-secretase inhibitor DAPT, or cisplatin chemotherapy. This showed that cell death from these therapies was significantly reduced in co-cultures with MRC5 in comparison to mono-cultures of TNBC cell lines (Boelens et al, 2014). Critically, this study also found interferon signaling to be involved in therapy resistance – this is discussed in detail in section 7.2. In addition, this study highlighted that MDA-MB-468 cells were not protected by MRC5 fibroblasts, as I found in my own work. However, clinically practicable methods of inhibiting the protection were not used, with only siRNA used to confirm dependence on the relevant pathways.
The similarities and differences between these two studies and my own work are summarised in Table 7.1. Some comparative strengths of my work are that are I used a range of TNBC cell lines (unlike Ham et al) paired with both primary and immortalised breast NFs and CAFs (unlike Boelens et al), a range of therapies (different chemotherapeutic drug classes) that are relevant to current chemotherapy treatments for primary breast cancer (unlike either study). I also investigated the potential for blocking the influence of fibroblasts using a range of approaches (blocking antibodies, JAK/STAT inhibitor and miR-155 inhibitor), some of which could potentially be used in patients (unlike either study to an extent, and certainly unlike Boelens et al). Work covered in this thesis therefore provides a comprehensive analsysis of the role of CAFs in TNBC in chemoresistance and approaches to block CAF-induced chemoresistance.

A further point for comparison is the assays used to assess chemo-response. I measured the ability of fibroblasts to impact on chemoresistance in both short-term chemo-survival assays and long-term clonogenic assays (section 3.3.1 and 3.3.2). Ham et al assessed chemoresistance by quantifying GFP expression levels in TNBC cells 6 days following treatment, representing a measure between the time-scale of my two assays. By contrast, Boelens et al analysed cell death in the short-term rather than cell survival. The clonogenic assay I have used in particular may be more relevant than other assays in a cancer setting, as this may reflect the ability of the surviving cancer cells following treatment, to regrow and form clinical recurrences. In any event, these studies used a single assay, while my use of two independent assays may provide a more in depth profile of CAFs' role in TNBC chemoresistance.

	Ham et al (2018)	Boelens et al (2014)	Thesis
1. Cell lines	Triple negative	Triple negative cell	Triple negative
used	cell line - MDA-	lines - MDA-MB-	cell lines -
	MB-231	231, -468 and -157	MDA-MB-231, -
			468 and -157
	Fibroblasts-	Fibroblasts-	Fibroblasts-
	Normal breast	MRC5 normal	Immortalised
	fibroblasts and	human fibroblasts	and primary
	breast CAFs	isolated from lung	breast normal
	extracted from		fibroblasts and
	patients		breast CAFs
2. Therapies	Paclitaxel	Radiation and	Epirubin and
investigated	chemotherapy	cisplatin	docetaxel
		chemotherapy	chemotherapy
3. Techniques	Co-culture	Co-culture cell	Co-culture
used	spheroids	death quantified	short-term
	quantifying GFP		survival assay
	expression after		and clonogenic
	6 days		assay
4. Pathway	CXCL12 and	Exosomal transfer	Interferon
responsible	CXCR4 signaling	of RIG-1 to induce	signaling
		interferon stimulated	
		genes	
5. Inhibition of	YES- inhibitor of	YES- siRNA	YES- interferon
pathway leads	CXCR4 signaling		blocking
to			antibodies,
sensitization			JAK/STAT
			inhibitor and
			miR-155
			inhibitor

Table 7.1: Key features of Ham et al (2018) and Boelens et al (2014) compared to work carried out in this thesis

In the clinical setting, the evidence for CAFs having these roles in therapy resistance is disappointingly poor – since very few studies have quantified CAFs in clinical samples. There has been no published data identifying that an increase in chemoresistance in breast cancer is attributed to increasing numbers of CAFs. A large number of studies identify simply the presence of CAFs but not whether a difference in the number of CAFs has an impact. In most cases, this is ascertained by using tumour stroma ratio as tumour stroma contains CAFs. Comparisons have been made in 10-year disease-free survival between patients with invasive breast tumours with higher than 50% tumour stroma ratio, or lower; the former group had much worse outcomes (17% 10year disease-free survival) in comparison to the latter (87%) (Vangangelt et al, 2018). However, in inflammatory breast cancer, the tumour stroma ratio has no impact on prognosis for patients (Downey et al, 2015). Therefore, tumour stroma ratio can play different roles in different breast cancer subtypes. Importantly, in TNBC, patients displaying higher than 50% tumour stroma had a poorer prognosis when compared to patients with less than 50% tumour stroma (Moorman et al, 2012). Therefore, this evidence could be extrapolated to support the idea that the presence of large numbers of CAFs correlates with relative chemoresistance, in agreement with my *in vitro* study.

7.2 Interferons as mediators of chemoresistance in TNBC

The interferon signaling pathway was identified as up-regulated in TNBC cells by the presence of CAFs during chemotherapy treatment, with up-regulation of various components of the signaling pathway itself (for example, STAT2 and IFR9) as well as of multiple interferon stimulated genes (ISGs), such as OAS1/2 and MX1/2 (section 4.3.5). The prototypical role for activation of the interferon pathway is in immune responses. Viral induction of interferon release enables interferon to bind to their respective Type I or Type II receptors and activation JAK/STAT signaling. Activation of JAK/STAT signaling and binding of respective STAT complexes to either ISRE (Type I) or GAS (Type II) elements, enables transcription of multiple ISGs that have antiviral, anti-proliferative and immune-modulatory functions as highlighted in section 1.13 (Tsuno et al, 2009). A number of related studies have been published that implicate interferon signaling as a factor in chemoresistance in breast cancer. Of most relevance, is work from the Minn group, as already cited above, which identified fibroblasts as an activator of interferon signaling in some TNBC cell lines, leading to therapy resistance (Weichselbaum et al, 2008, Boelens et al, 2014, Benci et al, 2016). However, these papers define that this is a result of exosomal transfer of RNA (largely non-coding RNA), which activates the latter stages of the interferon signaling pathway. From the genes that were up-regulated in the data set in section 4.3.5, there were increased levels of ISGs but also components of the signaling pathway itself which indicates that the resistance seen is potentially driven by interferon release from CAFs driving the very start of the pathway. Furthermore, up-regulation of IFN- β in CAFs in the presence of TNBC cells and chemotherapy was observed (Figure 4.11). The difference between IFN-driven activation or exosomal RNA-driven activation is critical, as only the first method of activation could be blocked therapeutically using antibodies targeting the IFN receptors. In my work, the activation of interferon signaling was blocked by inhibition of receptor activation (Figure 5.2 and 5.4). This suggest that in this thesis, in which I have used breast CAFs rather than the unrelated lung fibroblast line MRC5, paracrine secretion of interferons and activation via classical receptor-mediated pathways predominate, rather than exosomal transfer of RNA.

In addition, 2 further studies have been carried out that implicate interferon activation in chemo-response. Firstly, an interferon-related gene signature has been predictive for chemotherapy response in breast cancer. Using 34 cancer cell lines from the NCI60 panel, an IFN-related DNA damage resistance signature (IRDS) was associated with resistance to chemotherapy and/or radiation. The IRDS signature was then analysed in clinical samples following adjuvant therapy in a range of tumours. In breast cancer, a seven gene-pair classifier successfully predicted efficacy of adjuvant chemotherapy. The IRDS included the interferon-related genes ISG15, STAT1 and IFITM1 (Weichselbaum et al, 2008). A further paper has highlighted that activation of STAT1 predicts poor response to chemotherapy in breast cancer. This was determined by analyzing expression profiles in residual tumour cells in chemo-

responder patient derived xenografts following antracycline and cyclophosphamide treatment (Legrier et al, 2015). Both of these studies implicate IFN- γ as the main pathway involved in chemoresistance in breast cancer given the dependence of up-regulation of STAT1. However, work in this thesis points more towards activation of the Type I interferon pathway as up-regulation of STAT2 and IRF9 in section 4.3.5 was observed, which is specific to the Type I interferon pathway, as well as the fact that IFN- β was up-regulated when CAFs and MDA-MB-231 cells were combined (Figure 4.11) and blocking antibodies against Type I signaling worked more consistently in both MDA-MB-231 and -157 TNBC cells (Figures 5.1 and 5.3).

There are also other data in the literature that provide support for the idea that stromal fibroblasts can produce interferons, and that epithelial cells can potentially respond to such signals. Both murine and human normal fibroblasts have been shown to produce IFN- β when cultured with the bacterial species Shigella flexneri or Escherichia coli (Hess et al, 1987). Secretion of IFN-β has also been observed previously in CAFs (Slaney et al, 2013). In addition, studies have shown that normal human skin fibroblasts produce IFN-β when grown on collagen membranes blended with fibronectin and vitronectin, indictating fibroblast attachment to the ECM can enable secretion of IFN- β (Higuchi et al, 2002). The identification of the specific interferon responsible for CAF-induced chemoresistance was required and therefore investigated as there is no literature determining a direct role for a particular CAF-secreted interferon in chemoresistance. Interestingly, in section 4.3.9, interferons were not normally produced in either the fibroblasts or epithelial cells. Basal levels of IFN- α and IFN-y were undetermined in both CAFs and MDA-MB-231 cells, with basal levels of IFN-β in both NFs and CAFs undetectable or low and basal levels undetermined in MDA-MB-231. Surprisingly, IFN- β levels only increased in CAFs when co-cultured with MDA-MB-231 cells, with IFN-β expression remaining undetermined in MDA-MB-231 component. Induction of expression of IFN-β in CAFs only in the presence of TNBC cells is a key finding that has not previously been observed, and represents complex cross-talk between the two cell types, where the TNBC cells induce IFN- β expression in the CAFs in order for the TNBC cells then to respond to this paracrine signal.

In addition, there has been increasing evidence that epithelial cells can respond to interferons. IFN- γ has been shown to increase MUC1 expression in haematopoietic and epithelial cancer cells (Reddy et al, 2003) and breast cancer cells also have the ability to secrete interferons (Wan et al, 2012). The ability of breast cancer epithelial cells to respond to interferons is evident throughout much of my data set, with CAFs inducing interferon signaling activation (Figure 4.6), and breast cancer cells responding to recombinant interferons in terms of both activation of the interferon signaling pathway, and induction of chemoresistance (Figure 4.7 and 4.8).

Interestingly, IFN driven chemoresistance contradicts some of the previous literature cited. IFN- β has been shown to repress cancer stem cell properties in TNBC and can also reduce the aggressive nature of breast cancers by reducing migration and metastasis (Doherty et al, 2017). However, data in this thesis suggests that activation of the interferon signaling occurs through secretion of IFN- β from CAFs (Figure 4.11) with both Type I and Type II interferons sufficient for enhancing tumorigenesis and protecting TNBC cells from chemotherapy (Figure 4.7 and 4.8).

It is also worth commenting on the studies using fibroblast conditioned medium in section 3.3.5, which has been previously discussed in detail in section 3.4.2. However, this requires further discussion here, given results in the following chapters. Conditioned media from CAFs did not confer chemoresistance on TNBC cells (section 3.3.5) casting doubt that a secreted factor was responsible for chemoresistance. Another plausible explanation for this was the requirement of cross-talk between TNBC cells and CAFs. From section 4.3.9 and figure 4.11, it is clear that cross-talk between breast cancer cells and CAFs is required for chemoresistance to occur given CAF secretion of IFN- β was only upregulated *in the presence MDA-MB-231 cells*. This finding explains why conditioned media experiments did not enable protection in TNBC cells. To further improve the conditioned media experiment, cross-talk between the 2 cell lines could be taken into account by using conditioned media from CAF/TNBC co-cultures, or by using transwell approaches.

7.3 Claudin-low TNBC phenotypes in CAF-induced chemoresistance

As described in section 1.4.2, there are multiple different TNBC subtypes. In order to cover some of this diversity, three different TNBC cell lines were used in my experiments. Both MDA-MB-231 and MDA-MB-157 are characterized as claudin-low with negative ER, PR and HER2 status, high Ki67 and low expression of claudin-3, -4 and -7. MDA-MB-468 are classified as ER, PR and HER2 negative, but EGFR positive, high Ki67 and with cytokeratin 5/6 expression, thereby defining a basal subtype (Grigoriadias et al, 2012, Neve et al, 2006).

Interestingly, in section 3.3.1 and 3.3.2, differing responses could be observed between the claudin-low cell lines (MDA-MB-231 and MDA-MB-157) and the basal cell line (MDA-MB-468). CAFs induced consistent chemoresistance in the claudin-low triple negative phenotype but not in the basal triple negative phenotype. Here, I therefore highlight a novel role of CAF-induced chemoresistance in claudin-low TNBC to epirubicin and docetaxel, which is extremely relevant in the treatment of primary TNBC. In addition, up-regulation of ISGs OAS1 and MX1 (section 4.3.6) and the ability to respond to recombinant interferons and confer chemoresistance (section 4.3.7) were attributed only to the claudin-low TNBC subtype, and not the basal subtype (Figures 4.7, 4.9 and 4.10). ISG expression analysis has previously been carried in all 3 cell lines used here (MDA-MB-231, -157 and -468) when either in mono-culture or when cultured with MRC5 fibroblasts in the absence of treatment (Boelens et al. 2014). Work in this thesis is therefore further confirmation of the ability of the claudin-low TNBC subtype to be able to respond to and activate interferon signaling, while the single representative of the basal TNBC subtype was not able to.

Although CAF induced chemoresistance to epirubicin and docetaxel in the claudin-low subtype in comparison to the basal subtype is a novel finding, the two subtypes have been compared in many other respects previously, including clinical response to treatments, disease free survival and overall survival. In one

study, when comparing pCR of claudin-low and basal subtypes, claudin-low tumours had a 39% pCR whereas basal had a higher pCR 73%, which was not deemed significant (p value=0.08) (Prat et al, 2010). However, another study showed the pCR of claudin-low and basal at 32% and 33% respectively (Sabatier et al, 2014), therefore a firm conclusion as to whether the pCR rates differ remains impossible. When looking at 5-year disease free survival, claudinlow TNBC has higher disease free survival at 79.3% with basal TNBC at 73.1%, which was, again, not significantly different (p=0.40) (Dias et al, 2017). Another study had similar findings with disease free survival of 67% and 60% respectively, which was not significant (Sabatier et al, 2014). 5-year overall survival of claudin-low TNBC patients was also higher in at 89.7% compared to basal TNBC at 82.6% (Dias et al, 2017). Curiously, some of these differences in survival - if they are to be believed - are not what one would predict when extrapolating from my data that suggest that the claudin-low cancers may have an extra chemoresistance mechanism not available for the basal cancers. Based on data in section 3.3.1 and 3.3.2, predictions could be made that claudin-low breast cancer patients would have worse overall survival when compared to basal breast cancer patients as a result of this extra chemoresistance, while the reverse is actually seen clinically. It is also important to remember that this extra chemoresistance mechanism in claudinlow TNBC is CAF dependent (section 3.3.1 and 3.3.2), therefore will not be relevant to all claudin-low TNBCs.

7.2 Limitations

Although this work is promising, there is further scope to build on these data sets. A major limitation is the lack of translation of these findings into *in vivo* models. 2 key findings need to be tested - to back up these findings: firstly, a demonstration that CAFs induce chemoresistance in TNBC tumours; secondly, a demonstration that blocking CAF-induced interferon activation, using interferon blocking antibodies or small molecules, halts CAF-induced chemoresistance and thereby increases chemotherapy responses without major side-effects.

In addition, it is necessary to support the pathways proposed here using evidence from clinical breast cancer samples. Assessments could be made of whether tumours with higher proportions of CAFs show evidence of activation of interferon signaling in the cancer cells, and whether this correlates with relatively poor responses to chemotherapy or cancer survival.

7.3 Final conclusions

A novel role for CAFs has been identified in some TNBC cell lines. CAFs, but not NFs, induce chemoresistance in claudin-low TNBC lines. This appears to be through the paracrine release of IFN- β from the CAFs, which drives interferon signaling in the epithelial cells and chemoresistance, perhaps associated with reduced proliferation of the epithelial cells. In addition, CAFs can also induce anoikis resistance in claudin-low TNBC cells, likely through the interferon signaling pathway also. Both observations have potential clinical implications. First, CAF-induced chemoresistance could be reversed through blocking the interferon signaling pathway during chemotherapy, thereby improving chemotherapy responses. Secondly, CAF-induced anoikis-resistance could be targeted similarly to reduce the chances of metastasis, by reducing the survival of the cancer cells in the circulation. Further work is required to take advantage of these potential therapeutic directions.

Appendix

Target	Average Fold Change	Target	Average Fold Change
OAS1	30.21	VCAN	3.99
MX1	28.32	USP41	3.86
SPARC	26.53	XAF1	3.83
GREM1	25.35	BGN	3.65
GREM1	23.75	CXCL12	3.65
GREM1	23.26	ANPEP	3.61
POSTN	22.29	IFITM1	3.56
MX2	14.19	LDB2	3.49
COL1A2	13.88	CDH13	3.41
NID1	13.27	STAT2	3.4
MMP2	12.83	DPYSL3	3.36
IFI6	10.29	HIST1H4E	3.31
IFI44L	9.22	EPB41L3	3.31
IFI27	8.73	THBS2	3.26
COL1A1	8.73	MYH10	3.19
CLDN11	7.93	LYPD1	2.99
THY1	7.69	KIAA1549L	2.95
OAS2	7.66	EHD3	2.9
RAB3B	7.54	TAGLN	2.87
MFAP5	7.52	DPT	2.79
COL6A3	7.49	CCBE1	2.74
USP18	7.01	C19orf66	2.73
SPOCK1	6.89	WNT5A	2.7
RGS4	6.5	ASIC1	2.69
ITGA11	6.44	XIST	2.67
IRF9	6.16	PARP9	2.64
TERT	5.59	GPNMB	2.59
CCL2	5.59	AEBP1	2.58
TCF4	5.14	COL5A2	2.55
FGF2	5.11	TAP1	2.52
CDH6	5.06	PLAGL1	2.5
DCN	4.97	LOC353194	2.5
GREM1	4.75	SLC16A1	2.41
ADAM12	4.72	TRANK1	2.41
TLE4	4.71	LBH	2.41
SULF1	4.7	MSC	2.41
LGALS9	4.68	TGFBI	2.35
SEMA5A	4.65	FKBP10	2.35
TBX3	4.63	PTGFRN	2.3
CDH2	4.45	SYT11	2.29
KRT14	4.4	FYCO1	2.28
TRIM22	4.3	C16orf45	2.28
TMEM47	4.29	NUAK1	2.26
PRRX1	4.25	SYNC	2.22
COL3A1	4.15	PARP12	2.21

Target	Average Fold Change		
RTBDN	2.2		
OLFML3	2.19		
BVES	2.19		
MTA3	2.19		
PTGS2	2.17		
SNX33	2.17		
MYH16	2.15		
FLI1	2.08		
SIGLEC14	2.08		
TNFRSF9	2.06		
BNC2	2.06		
MRC2	2.06		
NLRC5	2.05		
SHROOM2	2.05		
RSAD2	2.04		
APOL1	2.03		
ARHGAP30	2.02		
ASPHD2	2.02		
AL445199.1	-2.07		
RP11-252K23.2	-2.08		
MTND4P32	-2.1		
AC107218.3	-2.17		
U3	-2.22		
RNA5SP405	-2.41		
AC093642.4	-2.57		

Appendix Table 1: Full list of gene expression changes between 0% and 20% cultures following Affymetrix analysis

MiRNAs present in 0% but not in 20%	MiRNAs present in 20% but not in 0%	
miR-let-7f	miR-27a	
miR-32	miR-31	
miR-452	miR-130b	
	miR-145	
	miR-146b	
	miR-152	
	miR-185	
	miR-196b	
	miR-199a-3p	
	miR-424	
	miR-433	
	miR-485-3p	
	miR-483	
	miR-510	
	miR-539	
	miR-551b	
	miR-674	
	miR-708	

<u>Appendix table 2: List of miRNAs differentially expressed that are present in</u> <u>0% cultures but not 20% and miRNAs present in 20% cultures but not 0%.</u>

MiRNAs decrease from 0 to 20%	MiRNAs increase from 0 to 20%		
miR-22	miR-34a		
miR-25	miR-98		
miR-106b	miR-132		
miR-125a-3p	miR-139-5p		
miR-193b	miR-200c		
miR-210	miR-324-3p		
miR-218	miR-324-5p		
miR-365	miR-155		
miR-532	miR-411		
miR-652-	miR-422a		
miR-671-3p	miR-454		
	miR-455-3p		
	miR-489		

Appendix table 3: List of miRNAs differentially expressed that are decreased from 0% to 20% and miRNAs that are increased from 0% to 20%

Target	Average Fold Change
COL3A1	4.15
COL5A2	2.55
COL6A3	7.49
DPT	2.79
SPARC	26.53
SPOCK1	6.89
WNT5A	2.7
SEMA5A	4.65
DCN	4.97
ADAM12	4.72
BGN	3.65
LGALS9	4.68
TGFBI	2.35
POSTN	22.29
MFAP5	7.52
THBS2	3.26
AEBP1	2.58
SULF1	4.7
CCBE1	2.74
VCAN	3.99
FGF2	5.11
NID1	13.27
CCL2	5.59
GREM1	25.53
MMP2	12.83
CXCL12	3.65
COL1A1	8.73
COL1A2	13.88

Appendix table 4: List of extracellular matrix genes up-regulated from 0% to 20% cultures from Affymetrix analysis

Pathway	P Value	Genes from	Genes from
		input	annotation
Interferon alpha/beta signaling	1.38E-14	12	69
Ensemble of genes encoding core ECM	1.26E-12	17	275
Ensemble of genes encoding ECM and ECM associated proteins	1.45E-11	28	1028
Interferon signaling	3.29E-11	14	202
ECM organisation	5.45E-09	14	298
Cytokine signaling in immune system	8.42E-09	21	763
Genes encoding collagen proteins	8.62E-06	5	44
Defective CHST14 causes ED, musculocontractual type	8.77E-07	3	7
Defective CHST3 causes SEDCJD	8.77E-07	3	7
Defective CHSY1 causes TPBS	8.77E-07	3	7
Collagen chain trimerization	1.20E-05	5	47
Binding and uptake of ligands by scavenger receptors	1.48E-05	5	49
Genes encoding structural ECM glycoproteins	3.60E-05	8	196
Assembly of collagen fibrils and other multimeric structures	4.00E-05	5	60
Dermatan sulphate biosynthesis	4.06E-05	3	11
Validated transcriptional targets of AP1 family members Fra1 and Fra2	6.33E-05	4	34
Genes encoding proteoglycans	7.11E-05	4	35
Collagen biosynthesis and modifying enzymes	8.43E-05	5	70
CS/DS degradation	8.84E-05	3	14
Influenza A	1.99E-04	7	173
Regulation of Wnt-mediated beta catenin signaling and target gene transcription	1.50E-04	5	79
ECM-receptor interaction	1.79E-04	5	82
Diseases of glycosylation	2.24E-04	5	86
Defective B3GAT3 causes JDSSDHD	2.29E-04	3	19

		-	
Defective B4GALT7 causes EDS, progeroid type	2.29E-04	3	19
Defective B3GALT6 causes EDSP2 and SEMDJL1	2.29E-04	3	19
Protein digestion and absorption	2.77E-04	5	90
Scavenging by class A receptors	3.13E-04	3	21
Collagen formation	3.23E-04	5	93
Chondroitin sulphate biogenesis	3.60E-04	3	22
AGE-RAGE signaling pathway in diabetic complications	4.31E-04	5	99
ECM proteoglycans	4.82E-04	4	57
Diseases associated with glycosaminoglycan metabolism	5.97E-04	3	26
A tetrasaccharide linker sequence is required for GAG synthesis	5.97E-04	3	26
Cell-cell junction organisation	6.65E-04	4	62
Integrin signaling pathway	7.05E-04	6	167
Interleukin-4 and 13 signaling	8.21E-04	5	114
Adherens junction interactions	1.01E-03	3	31
Syndecan-4 mediated signaling events	1.01E-03	3	31
Herpes simplex infection	1.20E-03	6	185
Type I (alpha/beta IFN) pathway	1.43E-03	2	9
ISG15 antiviral mechanism	1.50E-03	4	77
Antiviral mechanism by IFN-	1.50E-03	4	77
stimulated genes			
Hepatitis C	1.53E-03	5	131
Measles	1.69E-03	5	134
Cell junction organisation	2.55E-03	4	89
Interferon gamma signaling	3.11E-03	4	94

Appendix Table 5: List of dysregulated pathways taken from ToppGene pathway analysis. Top hits were then used to create Table 4.4.



Appendix Figure 1: Dose response curves used to calculate IC10, IC50 and IC75 epirubicin values for MDA-MB-231 and -468 cells

References

Aas, T., Børresen, A-L., Geisler, S., Smith-Sørensen, B., Johnsen, H., Varhaug, J.E., Akslen, L.A., Lønning, P.E. (1996) Specific P53 Mutations Are Associated With De Novo Resistance To Doxorubicin In Breast Cancer Patients. *Nature Medicine* **2:** 811-814

Adams, B.D., Kasinski, A.L., Slack, F.M. (2014) Aberrant Regulation And Function Of Micrornas In Cancer. *Curr Biol* **24:** 762-776

Alba, E., Lluch, A., Ribelles, N., Anton-Torres, A., Sanchez-Rovira, P., Albanell, J., Calvo, L., Garcia-Asenjo, J.A.L., Palacios, J., Chacon, J.I., Ruiz, A., De La Haba-Rodriguez, J., Segui-Palmer, M.A., Cirauqui, B., Margeli, M., Plazaola, A., Barnadas, A., Casas, M., Caballero, R., Carrasco, E., Rojo, F. (2015) High Proliferation Predicts Pathological Complete Response To Neoadjuvant Chemotherapy In Early Breast Cancer. *The Oncologist* **21**: 150-155

Alba, E., Chacon, J.I., Lluch, A., Anton, A., Estevez, E., Cirauqui, B. (2012) A Randomized Phase II Trial Of Platinum Salts In Basal-Like Breast Cancer Patients In The Neoadjuvant Setting. Results From The GEICAM/2006-03, Multicenter Study. *Breast Cancer Research And Treatment* **136**: 487-493

Ali, S., Rasool, M., Chaoudhury, H., Pushparaj, P.N., Jha, P., Hafiz, A., Mahfooz, M., Sami, G.A., Kamal, M.A., Bashir, S., Ali, A., Jamal, M.S. (2016) Molecular Mechanisms And Mode Of Tamoxifen Resistance In Breast Cancer. *Bioinformation* **12:** 135-139

Alley, M.C., Scudiero, D.A., Monks, A., Hursey, M.L., Czerwinski, M.J., Fine, D.L., Abbott, B.J., Mayo, J.G., Shoemaker, R.H., Boyd, M.R. (1988) Feasibility Of Drug Screening With Panels Of Human Tumour Cell Lines Using A Microculture Tetrazolium Assay. *Cancer Research* **48**: 589-601

Allocati, N., Masulli, M., Di Ilio, C., Federici, L. (2018) Glutathione Transferases: Substrates, Inhibitors And Pro-Drugs In Cancer And Neurodegenerative Diseases. *Oncogenesis* **7**: 8

American Cancer Society- Breast Cancer Staging <u>Https://Cancerstaging.Org/References-</u>

Tools/Quickreferences/Documents/Breastmedium.Pdf. Accessed June 2018

Anand, S., Majeti, B.K., Acevedo, L.M., Murphy, E.A., Mukthavaram, R., Scheppke, L. (2010) Microrna-132-Mediated Loss Of p120Ras-GAP Activates The Endothelium To Facilitate Pathological Angiogenesis. *Nature Medicine* **16**: 909-914

Ank, N., West, H., Bartholdy, C., Eriksson, K., Thomsen, A.R., Paludan, S.R. (2006) Lambda Interferon (IFN-Lambda), A Type III IFN, Is Induced By Viruses And Ifns And Displays Potent Antiviral Activity Against Select Virus Infections In Vivo. *J Virol* **80**: 4501-4509

Armstrong, T., Packham, G., Murphy, L.B., Bateman, A.C., Conti, J.A., Fine, D.R., Johnson, C.D., Benyon, R.C., Iredale, J.P. (2004) Type I Collagen Promotes The Malignant Phenotype Of Pancreatic Ductal Adenocarcinoma. *Clin Cancer Res* **10**: 7427-7437

Azmi, A.S., Mohammed, R.M. (2009) Non-Peptidic Small Molecule Inhibitors Against Bcl-2 For Cancer Therapy. *J Cell Physiol* **218:** 13-21

Bach, E.A., Aguet, M., Schreiber, R.D. (1997) The IFN Gamma Receptor: A Paradigm For Cytokine Receptor Signaling. *Annu Rev Immunol* **15**: 563-591

Bachelder, R.E., Wendt, M.A., Mercurio, A.M. (2002) Vascular Endothelial Growth Factors Promotes Breast Carcinoma Invasion In An Autocrine Manner By Regulating The Chemokine Receptor CXCR4. *Cancer Research* **62**: 7203-7206

Badaoui, M., Mimsy-Julienne, C., Saby, C., Van Gulick, L., Peretti, M., Jeannesson, P., Morjani, H., Ouadid-Ahidouch, H. (2018) Collagen Type 1 Promotes Survival Of Human Breast Cancer Cells By Overexpressing Kv10.1 Potassium And Orai1 Calcium Channels Through DDR1-Dependent Pathway. *Oncotarget* **9**: 24653-24671

Badve, S., Dabbs, D.J., Schnitt, S.J., Baehner, F.L., Decker, T., Eusebi, V., Fox, S.B., Ichihara, S., Jacquemier, J., Lakhani, S.R., Palacios, J., Rakha, E.A., Richardson, A.L., Schmitt, F.C., Tan, P.H., Tse, G.M., Weigelt, B., Ellis, I.O., Resi-Filho, J. (2011) Basal-Like And Triple Negative Breast Cancers :A Critical Review With An Emphasis On The Implications For Pathologists And Oncologists. *Modern Pathology* **24**: 157-167

Baig, E., Fish, E.N. (2008) Distinct Signature Type I Interferon Responses Are Determined By The Infecting Virus And The Target Cell. *Antivir Ther* **13**: 409-422

Bainbridge, P. (2013) Wound Healing And The Role Of Fibroblasts. J

Wound Care 22: 407-408

Balkwill, F., Watling, D., Taylor-Papadimitriou, J. (1978) Inhibition By Lymphoblastoid Interferon Of Growth Of Cells Derived From The Human Breast. *International Journal Of Cancer* **22**: 258-265

Bartel, D.P. (2004) Micrornas: Genomics, Biogenesis, Mechanism And Function. *Cell* **116**: 281-297

Baselga, J., Norton, J., Albanell, J., Kim, Y.M., Mendelsohn, J. (1998) Recombinant Humanized Anti-HER2 Antibody (Herceptin) Enhances The Antitumour Activity Of Paclitaxel And Doxorubicin Against HER2/Neu Overexpressing Human Breast Cancer Xenografts. *Cancer Research* **58**: 2825-2831

Baselga, J. (2001) Herceptin Alone Or In Combination With Chemotherapy In The Treatment Of HER2-Positive Metastatic Breast Cancer: Pivotal Trials. *Oncology* **61:** 14-21

Baselga, J., Swain, S.M. (2010) CLEOPATRA: A Phase III Evaluation Of Pertuzumab And Trastuzumab For HER2-Positive Metastatic Breast Cancer. *Clin Breast Cancer* **10**: 489-491

Baselga, J., Cortés, J., Kim, S.B., Im, S.A., Hegg, R., Im, Y.H., Roman, L., Pedrini, J.L., Pienkowski, T., Knott, A., Clark, E., Benyunes, M.C., Ross, G., Swain, S.M., CLEOPATRA Study Group. (2012) Pertuzumab Plus Trastuzumab Plus Docetaxel For Metastatix Breast Cancer. *N Engl J Med* **366**: 109-119

Baselga, J., Gomez, P., Greil, R, Braga, S., Climent, M.A., Wardley, A.M., Kaufman, B., Stemmer, S.M., Pego, A., Chan, A., Goeminne, J.C., Graas, M.P., Kennedy, M.J., Ciruelos Gil, E.M., Schneeweiss, A., Zubel, A., Groos, J., Melezinkova, H., Awada, A. (2013) Randomized Phase II Study Of The Anti-Epidermal Growth Factor Receptor Monoclonal Antibody Cetuximab With Cisplatin Versus Cisplatin Alone In Patients With Metastatic Triple-Negative Breast Cancer. *Journal Of Clinical Oncology* **31**: 2586-2592

Basho, R.K., Gilcrease, M., Murthy, R.K., Helgason, T., Karp, D.D., Meric-Bernstam, F., Hess, K.R., Herbich, S.M., Valero, V., Albarracin, C., Litton, J.K., Chavez-Mcgregor, M., Ibrahim, N.K., Murray, J.L., Koenig, H.B., Hong, D., Subbiah, V., Kurzrock, R., Janku, F., Moulder, S. L. (2017) Targetting The PI3K/AKT/Mtor Pathway For The Treatment Of Mesenchymal Triple-Negative Breast Cancer: Evidence From A Phase 1 Trial Of Mtor Inhibiton In Combination With Liposomal Doxorubicin And Bevacizumab. *JAMA Oncology* 3: 509-515

Baum, B., Settleman, J., Quinlan, M. P. (2008). Transitions Between Epithelial And Mesenchymal States In Development And Disease. *Seminars In Cell & Developmental Biology*, **19**: 294-308.

Bedard, P.L., Di Leo, A., Piccart-Gebhart, M.J. (2010) Taxanes: Optimizing Adjuvant Chemotherapy For Early-Stage Breast Cancer. *Nat Rev Clin Oncol* **7**: 22-36

Begg, A.C., Mooren, E. (1989) Rapid Fluorescence-Based Assay For Radiosensitivity And Chemosensitivity Testing In Mammalian Cells In Vitro. *Cancer Research* **49:** 565-569

Bernstein, E., Caudy, A.A., Hammond, S.M., Hannon, G.J. (2001) Role For A Bidentate Ribonuclease In The Initiation Step Of RNA Interference. *Nature* **409**: 363-369

Bhattacharyya, G.S., Basu, S., Agarwal, V., Malhotra, H., Pareekh, P.M., Babu, K.G, Aggarwala, D. (2009) 41LBA Single Institute Phase II Study Of Weekly Cisplatinum And Metronomic Dosing Of Endoxan And Methotrexate In Second Line Metastatic Breast Cancer Triple-Negative. *European Journal Of Cancer Supplements* **7**: 18-19

Bloom, H.J.G., Richardson, W.W. (1957) Histological Grading And Prognosis In Breast Cancer. *British Journal Of Cancer* **11:** 359-377

Boelens, M.C., Wu, T.J., Nabet, B.Y., Xu, B., Qiu, Y., Toon, T., Azzam, D.J., Victor, C. T-S., Wiemann, B.Z., Ishwaran, H., Ter Brugge, P.J., Jonkers, J., Slingerland, J., Minn, A.J. (2014) Exosome Transfer From Stromal To Breast Cancer Cells Regulated Resistance Pathways. *Cell* **159**: 499-513

Bohnsack, M.T., Czaplinski, K., Gorlich, D. (2004) Exportin 5 Is A Rangtp-Dependent Dsrna-Binding Protein That Mediates Nuclear Export Of Pre-Mirnas. *RNA* **10**: 185-191

Bouchard, V., Demers, M.-J., Thibodeau, S., Laquerre, V., Fujita, N., Tsuruo, T., Beaulieu, J.-F., Gauthier, R., Vézina, A., Villeneuve, L., Vachon, P. H. (2007) Fak/Src Signaling In Human Intestinal Epithelial Cell Survival And Anoikis: Differentiation State-Specific Uncoupling With The PI3-K/Akt-1 And MEK/Erk Pathways. *Journal Of Cellular Physiology*, **212**: 717-728 Bravata, V., Minafra, L., Cammarate, F.P., Pisciotta, P., Lamia, D., Marchese, V., Petringa, G., Manti, L., Cirrone, G.A., Gilardi, M.C., Cuttone, G., Forte, G.I., Russo, G, (2018) Gene Expression Profiling Of Breast Cancer Cell Lines Treated With Proton And Electron Radiations. *Br J Radiol* **5**: 20170934

Breast Cancer Now, <u>Http://Breastcancernow.Org/About-Breast-</u> Cancer/Have-You-Recently-Been-Diagnosed-With-Breast-

Cancer/Understanding-Your-Results/Grades-And-Stages, Accessed May 2018

Brown, I., Shalli, K., Mcdonald, S.L., Moir, S.E., Hutcheon, A.W., Heys, S.D., Schofield, A.C. (2004) Reduced Expression Of P27 Is A Novel Mechanism Of Docetaxel Resistance In Breast Cancer Cells. *Breast Cancer Research* **6**: R601-607

Bryant, H.E., Schultz., Thomas, H.D., Parker, K.M., Flower, D., Lopez, E., Kyle, S., Meuth, M., Curtin, N.J., Helleday, T. (2005) Specific Killing Of BRCA2-Deficient Tumours With Inhibitors Of Poly(ADP-Ribose) Polymerase. *Nature* **434:** 913-917

Buch, K., Peters, T., Nawroth, T., Sanger, M., Schmidberger, H., Langguth, P. (2012) Determination Of Cell Survival After Irradiation Via Clonogenic Assay Versus Multiple MTT Assay – A Comparative Study. *Radiation Oncology* **7:** 1

Buchholz, T.A., Hunt, K.K., Whitman, G.J., Sahin, A.A., Hortobagyi, G.N. (2003) Neoadjuvant Chemotherapy For Breast Carcinoma: Multidisciplinary Considerations Of Benefits And Risks. *Cancer* **98**: 1150-1160

Bussard, K.M., Mutkus, L., Stumpf, K., Gomez-Manzano, C., Marini, F.C. (2016) Tumour-Associated Stromal Cells As A Key Contributor To The Tumour Microenvironment. *Breast Cancer Research* **18**: 84

Burstein, M.D., Polyak, K., Wong, J.S., Lester, S.C., Kaelin, C.M. (2004) Ductal Carcinoma In Situ Of The Breast. *N Eng J Med* **350**: 1430-1441

Burstein, M.D., Tsimelzon, A., Poage, G.M., Covington, K.R., Contreras, A., Fugua, S.A., Savage, M.I., Osborne, C.K., Hilsenbeck, S.G., Chang, J.C., Mills, G.B., Lau, C.C., Brown, P.H. (2015) Comprehensive Genomic Analysis Identifies Novel Subtypes And Targets Of Triple-Negative Breast Cancer. *Clinical Cancer Research* **21**: 1688-1698

Cai, X., Hagedorn, C.H., Cullen, B.R. (2004) Human Micrornas Are Processed From Capped, Polyadenylated Transcripts That Can Also Function As Mrnas. RNA 10: 1957-1966

Camp, T.J., Elloumi, F., Roman-Perez, E., Rein, J., Stewart, D.A., Harrell, C.J., Perou, C.M., Troester, M.A. (2011) Interactions With Fibroblasts Are Distinct In Basal-Like And Luminal Breast Cancers. *Molecular Cancer Research* **9:** 3-13

Cancer Genome Atlas Network. (2012) Comprehensive Molecular Portraits Of Human Breast Tumours. *Nature* **490:** 61-70

Cancer Research UK, <u>Http://Www.Cancerresearchuk.Org/Health-</u> <u>Professional/Cancer-Statistics/Statistics-By-Cancer-Type/Breast-</u> <u>Cancer?_Ga=2.238148544.1199982128.1524849134-</u> 1968803846.1412951190, Accessed April 2018

Cancer Research UK, Https://Www.Cancerresearchuk.Org/Health-Professional/Cancer-Statistics/Statistics-By-Cancer-Type/Breast-Cancer/Diagnosis-And-Treatment#Heading-Four, Accessed June 2018

Cao, Z., Livas, T., Kyprianou, N. (2016) Anoikis And EMT: Lethal "Liaisons" During Cancer Progression. *Critical Reviews In Oncogenesis*, **21**: 155-168.

Capparelli, C., Chiavarina, B., Whitaker-Menezes, D., Pestell, T.G., Pestell, R.G., Hulit, J., Ando, S., Howell, A., Martinez-Outschoorn, U.E., Sotgia, F., Lisanti, M.P. (2012) CDK Inhibitors (P16/P19/P21) Induce Senescence And Autophagy In Cancer-Associated Fibroblasts, "Fueling" Tumour Growth Via Paracrine Interactions, Without An Increase In Neo-Angiogenesis. *Cell Cycle* **11:** 3599-3610

Carey, L.A, Perou, C.M., Livasy, C.A., Dressler, L.G., Cowan, D., Conway, K., Karaca, G., Troester, M.A., Tse, C.K., Edmiston, S., Deming, S.L., Geradts, J., Cheang, M.C.U., Nielsen, T.O., Moorman, P.G., Earp, S., Millikan, R.C. (2006) Race, Breast Cancer Subtypes And Survival In The Carolina Breast Cancer Study. *JAMA* **295**: 2492-2502

Carey, L.A., Dees, C.E., Sawyer, L., Gatti, L., Moore, D.T., Collichio, F., Ollila, D.W., Sartor, C.I., Graham, M.L., Perou, C.M. (2007) The Triple Negative Paradox: Primary Tumour Chemosensitivity Of Breast Cancer Subtypes. *Clinical Cancer Research* **13**: 2329-2334

Carey, L.A., Rugo, H.S., Marcom, P.K., Mayer, E.L., Esteva, F.J., Ma, C.X. (2012) TBCRC 001: Randomized Phase II Study Of Cetuximab In

Combination With Carboplatin In Stage IV Triple-Negative Breast Cancer. *Journal Of Clinical Oncology* **30:** 2615-2623

Carey, L. A. (2011) Directed Therapy Of Subtypes Of Triple-Negative Breast Cancer. *Oncologist*, **16**: Suppl 1, 71-8.

Cariou, S., Donovan, J.C.H., Flanagan, W.M., Milic, A., Bhattacharya, N., Slingerland, J.M. (2000) Down-Regulation Of P21^{waf1/CIP1} Or p27^{Kip1} Abrogates Antiestrogen-Mediated Cell Cycle Arrest In Human Breast Cancer Cells. *PNAS* **97:** 9042-9046

Carney, D.N., Winkler, C.F. (1985) In Vitro Assays Of Chemotherapeutic Sensitivity. *Important Advances In Oncology* 78-103

Chambon, P., Weill, J.D., Mandel, P. (1963) Nicotinamide Mononucleotide Activation Of New DNA-Dependent Polyadenylic Acid Synthesizing Nuclear Enzyme. **11:** 39-43

Chen, J., Bardes, E.E., Aronow, B.J., Jegga, A.G. (2009) Toppgene Suite For Gene List Enrichment Analysis And Candidate Gene Prioritization. *Nucleic Acids Res* **37**: 305-311

Chen, Q.X., Wang, X.X., Lin, P.Y., Zhang, J., Li, J.J., Song, C.G., Shao, Z.M. (2017) The Different Outcomes Between Breast-Conserving Surgery And Mastectomy In Triple Negative Breast Cancer: A Population Based Study From The SEER 18 Database. *Oncotarget* **8**: 4773-4780

Chen, D.R., Lu, D.Y., Lin, H.Y., Yeh, W.L. (2014) Mesenchymal Stem-Cell Induced Doxorubicin Resistance In Triple Negative Breast Cancer. *Biomed Res Int* **2014:** 532161

Chen, J.H., Feig, B., Agrawal, G., Yu, H., Carpenter, P.M., Mehta, R.S., Nalcioglu, O., Su, M.Y. (2007) MRI Evaluation Of Pathologically Complete Response And Residual Tumors In Breast Cancer After Neoadjuvant Chemotherapy. *Cancer* **112:** 17-26

Chen, Y., Jiang, L., Gao, B., Cheng, Z.Y., Jin, J., Yang, K.H. (2016) Survival And Disease-Free Benefits With Mastectomy Versus Breast Conservation Therapy For Early Breast Cancer: A Meta-Analysis. *Breast Cancer Research And Treatment* **Epub:** 1-9

Chendrimada, T.P., Gregory, R.I., Kumaraswamy, E., Norma, J., Cooch, N., Nishikura, K., Shiekhattar, R. (2005) TRBP Recruits Dicer Complex To Ago2 For Microrna Processing And Gene Silencing. *Nature* **436**: 740-744

Chernyy, V., Pustylnyak, V., Kozlov, V., Gulyaeva, L. (2018) Increased Expression Of Mir-155 And Mir-222 Is Associated With Lymph Node Positive Status. *J Cancer*, **9:** 135-140.

Chew, H.K. (2001) Adjuvant Therapy For Breast Cancer: Who Should Get What? Western Journal Of Medicine **174:** 284-287

Chiarugi, P., Giannoni, E. (2008) Anoikis: A Necessary Death Program For Anchorage-Independent Cells. *Biochem Pharmacol* **76:** 1352-1364

Chiarle, R., Pagnano, M., Inghirami, G. (2001) The Cyclin Dependent Kinase Inhibitor P27 And Its Prognostic Role In Breast Cancer. *Breast Cancer Res* **3**:91-94

Chittaranjan, S., Bortnik, S., Dragowska, W.H., Xiu, J., Abeysundara, N., Leung, A., Go, N.E., Devorkin, L., Weppler, S.A., Gelmon, K., Yapp, D.T., Bally, M.B, Gorski, S.M. (2014) Autophagy Inhibition Augments The Anticancer Effects Of Epirubicin Treatment In Anthracycline-Sensitive And Resistant Triple Negative Breast Cancer. *Clinical Cancer Research* **20**: 3159-3173

Choi, J.A., Kim, J.Y., Lee, J.Y., Kang, C.M., Kwon, H.J., Yoo, Y.D., Kim, T.W., Lee, Y.S., Lee, S.J. (2001) Induction Of Cell Cycle Arrest And Apoptosis In Human Breast Cancer Cells By Quercetin. *Int J Oncol* **19:** 837-844

Choi, H. J., Lui, A., Ogony, J., Jan, R., Sims, P. J., Lewis-Wambi, J. (2015) Targeting Interferon Response Genes Sensitizes Aromatase Inhibitor Resistant Breast Cancer Cells To Estrogen-Induced Cell Death. *Breast Cancer Res* **17:** 6.

Cigler, T. Joint Clinical Trials Office <u>Https://Jcto.Weill.Cornell.Edu/Open_Clinical_Trials/Phase-12-Trial-Of-</u> <u>Ruxolitinib-In-Combination-With-Trastuzumab-In-Metastatic-Her2-Positive-</u> Breast-Cancer.

Clarke, R., Tyson, J. J. & Dixon, J. M. (2015) Endocrine Resistance In Breast Cancer – An Overview And Update. *Molecular And Cellular Endocrinology* **418**: 220-234.

Clayton, A., Evans, R.A.M Petit, E., Hallett, M., Williams, J.D., Steadman, R. (1998) Cellular Activation Through The Ligation Of Intercellular Adhesion Molecule-1. *Developmental Biology* **111**: 443-453

Cleator, S., Heller, W., Coombes, R.C. (2007) Triple-Negative Breast

Cancer: Therapeutic Options. The Lancet Oncology 374: 235-244

Concepcion, C.P., Han, Y., Mu, P., Bonetti, C., Yao, E., D'Andrea, A., Vidigal, J.A., Maughan, W.P., Ogrodowski, P., Ventura, A. (2012) Intact P-53 Dependent Responses In Mir-34 Deficient Mice. *Plos Genetics* **8:** E1002797

Cooke, T., Reeves J., Lanigan, A., Stanton, P. (2001) Her2 As A Prognostic And Predictive Marker For Breast Cancer. *Annals Of Oncology* **12**: 23-28

Cooper, C. S. (2001) Applications Of Microarray Technology In Breast Cancer Research. *Breast Cancer Res* **3:** 158-75.

Cortazar, P., Zhang, L., Uuntch, M., Mehta, K., Costantino, J. P., Wolmark, N., Bonnefoi, H., Cameron, D., Gianni, L., Valagussa, P., Swain, S. M., Prowell, T., Loibl, S., Wickerham, D. L., Bogaerts, J., Baselga, J., Perou, C., Blumenthal, G., Blohmer, J., Mamounas, E. P., Bergh, J., Semiglazov, V., Justice, R., Eidtmann, H., Paik, S., Piccart, M., Sridhara, R., Fasching, P. A., Slaets, L., Tang, S., Gerber, B., Geyer, C. E., Pazdur, R., Ditsch, N., Rastogi, P., Eiermann, W. & Von Minckwitz, G. (2014.) Pathological Complete Response And Long-Term Clinical Benefit In Breast Cancer: The Ctneobc Pooled Analysis. *The Lancet*, **384**: 164-172.

Cowell, C.F., Weigelt, B., Sakr, R.A., Ng, C.K.Y., Hicks, J., King, T.A., Reis-Filho, J.S. (2013) Progression From Ductal Carcinoma In Situ To Invasive Breast Carcinoma: Revisted. *Molecular Oncology* **7**: 859-869

Crowley, L.C., Waterhouse, N.J. (2016) Measuring Survival Of Hematopoietic Cancer Cells With The Colony Forming Assay In Soft Agar. *Cold Spring Harbor Protoc* **8:** Epub

Crowley, L.C., Christensen, M.E., Waterhouse, N.J. (2016) Measuring Survival Of Adherent Cells With Colony Forming Assay. *Cold Spring Harbor Protoc* 8: Epub

Cullen, K.J., Yee, D., Bates, S.E., Brunner, N., Clarke, R., Dickson, R.E., Huff, K.K., Paik, S., Kosen, N., Valverius, E., Zugmaier, G., Lippman, M.E. (1989) Regulation Of Human Breast Cancer By Secreted Growth Factors. *Acta Oncologica* **28**: 835-839

D'Angelo, R.C., Ouzounova, M., Davis, A., Choi, D., Tchuenkam, S.W., Kim, G., Luther, T., Quraishi, A.A., Senbabaoglu, Y., Conley, S.J., Clouthier, S.G., Hassan, K.A., Wicha, M.S., Korkaya, H. (2015) Notch Reporter Activity In Breast Cancer Cell Lines Identifies A Subset Of Cells With Stem Cell Activity. *Molecular Cancer Therapy* **14:** 779-787

Darby, I.A., Laverdet, B., Bonte, F., Desmouliere, A. (2014) Fibroblasts And Myofibroblasts In Would Healing. *Clinical, Cosmetic And Investigational Dermatology* **7:** 301-331

Darnell, J.E Jr. (1997) Stats And Gene Regulation. Science 277: 1630-1635

Darnell, J.E Jr., Kerr, I.M., Stark, G.R. (1994) Jak-STAT Pathways And Transcriptional Activation In Response To Ifns And Other Extracellular Signaling Proteins. *Science* **264**: 1415-1421

Davies, M.A., Koul, D., Dhesi, H., Berman, R., Mcdonnell, T.J., Mcconkey, D., Yung, T.K., Steck, P.A. (1999) Regulation Of Akt/PKB Activity, Cellular Growth, And Apoptosis In Prostate Carcinoma Cells By MMAC/PTEN. *Cancer Res* **59**: 2551-2556

Davis, M.E. (2009) The First Delivery Of Sirna In Humans Via A Self Assembling Cyclodextrin Polymer Based Nanoparticle: From Concept To Clinic. *Molecular Pharmocology* **6:** 659-668

De Kruijf, E.M., Van Nes, J.G.H., Van De Velde, C.J.H., Putter, H., Smit, V.T.H.B.M., Liefers, G.J., Kuppen, P.J.K., Tollenaar, R.A.E.M., Mesker, W.E. (2011) Tumour-Stroma Ratio In The Primary Tumour Is Prognostic Factor In Early Breast Cancer Patients, Especially In Triple- Negative Carcinoma Patients. *Breast Cancer Research And Treatment* **125**: 687-696

De Weerd, N. A. & Nguyen, T. (2012) The Interferons And Their Receptors--Distribution And Regulation. *Immunol Cell Biol*, **90:** 483-91.

Dean, C.T., Jubelirer, S.J., Plants, B.A., Welch, C.A. (2009) Use Of Radiation After Breast Conserving Surgery (BCS) For DCIS And Early Invasive Breast Cancer At Charleston Area Medical Center (CAMC). A Study Of Compliance With National Comprehensive Cancer Network (NCCN) Guidelines. *W V Med J* **105:** 34-38

Denli, A.M., Tops, B.B., Plasterk, R.H., Ketting, R.F., Hannon, G.J. (2004) Processing Of Primary Micrornas By The Microprocessor Complex. *Naure* **432**: 231-235

Dias, K., Dvorkin-Gheva, A., Hallett, R.M., Wu, Y., Hassell, J., Pond, G.R., Levine, M., Whelan, T., Bane, A.L. (2017) Claudin-Low Breast Cancer; Clinical & Pathological Characteristics. *Plos One* **12**: E0168669

Diaz, M.O., Pomykala, H.M., Bohlander, S.K., Maltepe, E., Malik, K., Brownstein, B., Olopade, O.I. (1994) Structure Of The Human Type-I Interferon Gene Cluster Determined From A YAC Clone Contig. *Genomics* **22**: 540-552

Dijkers, P.F., Medema, R.H., Pals, C., Banerji, L., Thomas, N.S., Lam, E.W., Burgering, B.M., Raaijmakers, J.A., Lammers, J.W., Koenderman, L., Coffer, P.J. (2000) Forkhead Transcription Factor FKHR-L1 Modulates Cytokine Dependent Transciprtional Regulation Of P27 (KIP1). Mol Cell Biol **20**: 9138-9148

Ding, S-M., Lu, A-L., Zhang, W., Zhou, L., Xie, H-Y., Zheng, S-S., Li, Q-Y. (2018) The Role Of Cancer-Associated Fibroblast MRC-5 In Pancreatic Cancer. *J Cancer* **9:** 614-628

Doherty, M.R., Cheon, H., Junk, D.J., Vinayak, S., Varadan, V., Teil, M.L., Ford, J.M., Stark, G.R., Jackson, M.W. (2017) Interferon-Beta Repressed Cancer Stem Cell Properties In Triple-Negative Breast Cancer. *Proc Nati Acad Sci USA* **114**: 13792-13797

Donnelly, S.M., Paplomata, E., Peake, B.M., Sanabria, E., Chen, Z, Nahta, R.(2014) P38 MAPK Contributes To Resistance And Invasiveness Of HER2 Overexpressing Breast Cancer. *Curr Med Chem* **21**: 501-510

Downey, C.L., Simpkins, S.A., White, J., Holliday, D.L., Jones, J.L., Jordan, J.B., Kulka, J., Pollock, S., Rajan, S.S., Thygesen, H.H., Hanby, A.M., Speirs, V. (2014) The Prognostic Significance Of Tumour Stroma Ratio In Oestrogen-Receptor Positive Breast Cancer. *British Journal Of Cancer* **110**: 1744-1747

Downey, C.L., Thygesen, H.H., Sharma, N., Shaaban, A.M. (2015) Prognostic Significance Of Tumour Stroma Ratio In Inflammatory Breast Cancer. *Springerplus* **4**: 68

Doyle, L.A., Yang, W., Abruzzo, L.V., Krogmann, T., Gao, Y., Rishi, A.K., Ross, D.D. (1998) A Multidrug Resistance Transporter From Human MCF-7 Breast Cancer Cells. *Proc Nati Acad Sci USA* **95:** 15665-70

Dubey, P. (2012) Reporter Gene Imaging Of Immune Responses To Cancer: Progress And Challenges. *Theranostics* **2:** 355-362 Duenwald, S., Zhou, M., Wang, Y., Lejnine, S., Kulkami, A., Graves, J., Smith,R., Castle, J., Tokiwa, G., Fine, B., Dai, H., Fare, T., Marton, M. (2009) Development Of A Microarray Platform For FFPET Profiling: Application To The Classification Of Human Tumors. *J Transl Med* **28**: 65

Duffy, M.J., Macguire, T.M., Hill, A., Mcdermott, E., O'Higgings, N. (2000) Metalloproteinases: Role In Breast Carcinogenesis, Invasion And Metastasis. *Breast Cancer Research* **2:** 252-257

Dumont, N., Liu, B., Defilippis, R.A., Chang, H., Rabban, J.T., Karnezis, A.N., Tjoe, J.A., Marx, J., Parvin, B., Tlsty, T.D. (2013) Breast Fibroblasts Modulate Early Dissemination, Tumorigenesis, And Metastasis Through Alteration Of Extracellular Matrix Characteristics. *Neoplasia* **15**: 249-262

Dvorak, K. M., Pettee, K. M., Rubinic-Minotti, K., Su, R., Nestor-Kalinoski, A., Eisenmann, K. M. (2018) Carcinoma Associated Fibroblasts (Cafs) Promote Breast Cancer Motility By Suppressing Mammalian Diaphanous-Related Formin-2 (Mdia2). *PLOS ONE* **13:** E0195278.

Early Breast Cancer Trialists' Collaborative Group (EBCTCG). (2018) Long-Term Outcomes For Neoadjuvant Versus Adjuvant Chemotherapy In Early Breast Cancer: Meta-Analysis Of Individual Patient Data From Ten Randomised Trials. *The Lancet Oncology* **19**: 27-39

Egeblad, M., Ewald, A.J., Askautrud, H.A., Truitt, M.L., Welm. B.E., Bainbridge, E., Peeters, G., Krummel, M.F., Werb, Z. (2008) Visualizing Stromal Cell Dynamics In Different Tumor Microenvironments By Spinning Disk Confocal Microscopy. *Dis Model Mech* **1:** 155-167

Elston, C.W., Ellis, I.O. (1991) Pathological Prognostic Factors In Breast Cancer. I. The Value Of Histological Grade In Breast Cancer: Experience From A Large Study With Long-Term Follow-Up. *Histopathology* **19:** 403-410

Espinosa, C.E.S., Slack, F.J. (2006) The Role Of Micrornas In Cancer. Yale J Biol Med **79:** 131-140

Faraoni, I., Antonetti, F. R., Cardone, J. & Bonmassar, E. (2009) Mir-155 Gene: A Typical Multifunctional Microrna. *Biochimica Et Biophysica Acta (Bba) - Molecular Basis Of Disease*, **1792:** 497-505.

Faraoni, I., Antonetti, F. R., Cardone, J., Bonmassar, E. (2009) Mir-155 Gene: A Typical Multifunctional Microrna. *Biochimica Et Biophysica Acta (BBA) - Molecular Basis Of Disease*, **1792:** 497-505. FDA-

Https://Www.Fda.Gov/Drugs/Informationondrugs/Approveddrugs/Ucm592357 Accessed October 2018

Ferrara, N., Hillan, K.J., Gerber, H.P., Novotny, W. (2004) Discovery And Development Of Bevacizumab, An Anti-VEGF Antibody For Treating Cancer. *Nat Rev Drug Discov* **3**: 391-400

Fathers, K.E. (2012) Crk Adaptor Proteins Act As Key Signaling Integrators For Breast Tumourigenesis

Fedier, A., Schwarz, V.A., Walt, H., Carpini, R.D., Haller, U., Fink, D. (2001) Resistance To Topoisomerase Poisons Due To Loss Of DNA Mismatch Repair. *Int Journ Of Cancer* **93:** 571-576

Feng, X., Reder, N.P., Yanamandala, M., Hill, A., Franek, B.B., Niewold, T.B., Reder, A.T., Javed, A. (2012) Type I Interferon Signature Is High In Lupus And Neuromyelitis Optica But Low In Multiple Sclerosis. *J Neurol Sci* **313**: 48-53

Fisher, B., Brown, A.M., Dimitrov, N.V., Poisson, R., Redmond, C., Margolese, R.G., Bowman, D., Wolmark, N.,Wickerham, D.L., Kardinal, C.G. (1990) Two Months Of Doxorubicin-Cyclophosphamide With And Without Interval Reinduction Therapy Compared With 6 Months Of Cyclophosphamide, Methotrexate, And Fluorouracil In Positibe-Node Breast Cancer Patients With Tamoxifen-Nonresponsive Tumors: Results From The National Surgical Adjuvant Breast And Bowel Project B-15. *J Clin Oncol* **8**: 1483-1496

Forster, S.C., Tate, M.D., Hetzog, P.J. (2015) Microrna As Type I Interferon-Regulated Transcripts And Modulators Of The Innate Immune Response. *Front Immunol.* **6:** 334

Franken, N.A., Rodermond, H.M., Stap, J., Haveman, J., Van Bree, C. (2006) Clonogenic Assays Of Cells In Vitro. *Nature Protocols* **5**: 2315-2319

Franklin, M.C., Carey, K.D., Vajdos, F.F., Leahy, D.J., De Vos, A.M., Silwkowski, M.X. (2004) Insights Into Erbb Signaling From The Structure Of The Erbb2-Pertuzumab Complex. *Cancer Cell* **5:** 317-328

Friedman, R.C., Farh, K.K., Burge, C.B., Bartel, D.P. (2009) Most Mammalian Mrnas Are Conserved Targets Of Micrornas. *Genome Res* **19:** 92-105 Friedrichs, K., Ruiz, P., Franke, F., Gille, I., Terpe, H-J., Imhof, B.A. (1995) High Expression Level Of α 6 Integrin In Human Breast Carcinoma Is Correlated With Reduced Survival. *Cancer Research* **55**: 901-906

Frisch, S. M., Francis, H. (1994) Disruption Of Epithelial Cell-Matrix Interactions Induces Apoptosis. *The Journal Of Cell Biology*, **124:** 619.

Frisch, S. M., Ruoslahti, E. (1997) Integrins And Anoikis. *Current Opinion In Cell Biology*, **9:** 701-706.

Fruman, D.A., Rommel, C. (2014) PI3K And Cancer: Lessons, Challenges And Opportunities. *Nat Rev Drug Discov* **13**: 140-156

Fu, X.Y., Schindler, C., Improta, T., Aebersold, R., Darnell, J.E Jr. (1992) The Proteins Of ISGF-3, The Interferon Alpha-Induced Transcriptional Activator, Define A Gene Family Involved In Signal Transduction. *Proc Nati Acad Sci USA* **89:** 7840-7843

Fujita, K. (2006) Cytochrome P450 And Anticancer Drugs. *Curr Drug Metab* **7**:23-37

Fulford, L. G., Reis-Filho, J. S., Ryder, K., Jones, C., Gillett, C. E., Hanby, A., Easton, D., And Lakhani, S. R. (2007) Basal-Like Grade III Invasive Ductal Carcinoma Of The Breast: Patterns Of Metastasis And Long-Term Survival. *Breast Cancer Res* **9**: R4

Furumoto, Y., Gadina, M. (2013) The Arrival Of JAK Inhibitors: Advancing The Treatment Of Immune And Hematologic Disorders. *Biodrugs* **27:** 431-8.

Gabbiani, G., Ryan, G.B., Majne, G. (1971) Presence Of Modified Fibroblasts In Granulation Tissue And Their Possible Role In Wound Contraction. *Experientia* **27**: 549-550

Gao, S., Vond Der Malsburg, A., Dick, A., Faelber, K., Schröder, G.F., Haller, O., Kochs, G., Daumke, O. (2011) Structure Of Myxovirus Resistance Protein A Reveals Intra- And Intermolecular Domain Interactions Required For The Antiviral Function. *Immunity* **35:** 514-525

Gao, Y., Yin, S-P., Xie, X-S., Xu, D-D., Du, W-D. (2017) The Relationship Between Stromal Cell Derived SPARC In Human Gastriv Cancer Tissue And Its Clinocopathologic Significance. *Oncotarget* **8:** 86240-86252 Gascard, P., Tlsty, T.D (2016) Carcinoma-Associated Fibroblasts: Orchestrating The Composition Of Malignancy. *Genes & Development* **30**: 1002-1019

Gasparini, P., Cascione, L., Fassan, M., Lovat, F., Guler, G., Balci, S., Irkkan, C., Morrison, C., Croce, C. M., Shapiro, C. L., Huebner, K. (2014) Microrna Expression Profiling Identifies A Four Microrna Signature As A Novel Diagnostic And Prognostic Biomarker In Triple

Negative Breast Cancers. Oncotarget 5: 1174-1184.

Gee, H.E., Buffa, F.M., Camps, C., Ramachandran, A., Leek, R., Tayor, M., Patil, M., Sheldon, H., Betts, G., Homer, J., West, C., Ragoussis, J., Harris, A.L. (2011) The Small-Nucleolar Rnas Commonly Used For Microrna Normalisation Correlate With Tumour Pathology And Prognosis. *Br J Cancer* **104:** 1168-1177

Gianni, L., Pienkowski, T., Im, Y.H., Tseng, L.M., Liu, M.C., Lluch, A., Staroslawska, E., De La Haba-Rodriguez, J., Im, S.A., Pedrini, J.L., Poirier, B., Morandi, P., Semiglazov, V., Srimuninnimit, V., Bianchi, G.V., Magazzù., Mcnally, V., Douthwaite, H., Ross, G., Valagussa, P. (2016) 5-Year Analysis Of Neoadjuvant Pertuzumab And Trastuzumab In Patients With Locally Advanced, Inflammatory, Or Early-Stage HER2-Positive Breast Cancer (Neosphere): A Multicentre, Open-Label, Phase 2 Randomised Trial. *Lancet Oncol* **17**: 791-800

Gibbert, K., Schlaak, J.F., Yang, D., Dittmer, U. (2013) IFN-A Subtypes: Distinct Biological Activities In Anti-Viral Therapy. *Br J Pharmacol* **168:** 1048-1058

Goliwas, K.F., Richter, J.R., Pruitt, H.C., Araysi, L.M., Anderson N.R., Samant, R.S., Lobo-Ruppert, S.M., Berry, J.L., Frost, A.R. (2017) Methods To Evaluate Cell Growth, Viability And Response To Treatment In A Tissue-Engineered Breast Cancer Model. *Scientific Reports* **7**: 1-14

Gooch, J.L., Herrera, R.E., Yee. D. (2000) The Role Of P21 In Interferon Gamma-Mediated Growth Inhibition Of Human Breast Cancer Cells. *Cell Growth Differ* **11:** 335-342

Govindarajan, R., Duraiyan, J., Kaliyappan, K., Palanisamy, M. (2012) Microarray And Its Applications. *Journal Of Pharmacy & Bioallied Sciences* **4:** S310-S312.

Grandvaux, N., Tenoever, B. R., Servant, M. J., Hiscott J. (2002) The Interferon Antiviral Response: From Viral Invasion To Evasion. *Current Opinion Infectious Disease* **15:** 259-267.

Grigoriadis, A., Mackay, A., Noel, E., Wu, P.J., Natrajan, R., Frankum, J., Reis-Filho, J.S., Tutt, A. (2012) Molecular Characterisation Of Cell Line Models For Triple-Negative Breast Cancers. *BMC Genomics* **13:** 619

Guo, W., Giancotti, F.G. (2004) Integrin Signaling During Tumour Progression. *Nat Rev Mol Cell Biol* **5:** 816-826

Hall, A.G., Tilby, M.J. (1992) Mechanism Of Action Of, And Modes Of Resistance To, Alkylating Agents Used In The Treatment Of Haematological Malignancies. *Blood Review* **6:** 163-173

Haller, O., Kochs, G. (2011) Human Mxa Protein: An Interferon-Induced Dynamin-Like Gtpase With Broad Antiviral Activity. *Journal Of Interferon Cytokine Research* **31:** 79-87

Haller, O., Sterz, S., Kochs, G. (2007) The Mx Gtpase Family Of Interferon-Induced Antiviral Proteins. *Microbes Infect* **9**: 1636-1643

Halytskiy, V. (2017) Shifts In Mirnaome Underlie Anoikis Resistance Of Breast Cancer Cells. *The Breast* **32**: S33.

Ham, S.L., Thakuri, P.S., Plaster, M., Li, J., Luker, K.E., Luker, G.D., Tavana, H. (2018) Three-Dimensional Tumour Model Mimics Stromal-Breast Cancer Cells Signaling. *Oncotarget* **9**: 249-267

Hamam, R., Ali, A. M., Alsaleh, K. A., Kassem, M., Alfayez, M., Aldahmash, A., Alajez, N. M. (2016) Microrna Expression Profiling On Individual Breast Cancer Patients Identifies Novel Panel Of Circulating Microrna For Early Detection. *Sci Rep* **6**: 25997.

Han, J., Lee, Y., Yeom, K.H., Kim, Y.K., Jin, H., Kim, V.N. (2004) The Drosha-DGCR8 Complex In Primary Microrna Processing. *Genes Dev* 18: 3016-3027

Han, T., Liu, Z., Li, H., Xie, W., Zhang, R., Zhu, L., Guo, F., Han, Y., Sheng, Y., Xie, X. (2015) High Expression Of UBD Correlates With Epirubicin Resistance And Indicates Poor Prognosis In Triple Negative Breast Cancer. *Oncotargets And Therapy* **8:** 1643-1649

Hanahan, D., Weingber, R.A. (2000) Hallmarks Of Cancer. *Cell* **100**: 57-70 Hanamura, N., Yoshida, T., Matsumoto, E., Kawarada, Y., Sakakura, T. (1997) Expression Of Fibronectin And Tenascin-C Mrna By Myofibroblasts, Vascular Cells And Epithelial Cells In Human Colon Adenomas And Carcinomas. *Int J Cancer* **73**: 10-15

Harbeck, N., Beckmann, M.W., Rody, A., Schneeweiss, A., Müller, V., Fehm, T., Marschner, N., Gluz, O., Schrader, I., Heinrich, G., Untch, M., Jackisch, C. (2013) HER Dimerization Inhibitor Pertuzumab – Mode Of Action And Clinical Data In Breast Cancer. *Breast Care (Basel)* **8:** 49-55

Harrison, C., Vannucchi, A.M. (2012) Ruxolitinib: A Potent And Selective Janus Kinase 1 And 2 Inhibitor In Patients With Myelofibrosis. An Update For Clinicians. *Ther Adv Hematol* **3:** 341-354

Hart, I.R., Fidler, I.J. (1980) Cancer Invasion And Metastasis. *Q Rev Biol* **55:** 121-142

Hassan, M.S., Ansari, J., Spooner, D., Hussain, S.A. (2010) Chemotherapy For Breast Cancer (Review). *Oncol Rep* **24:** 1121-1131

Hennigs, A., Riedel, F., Gondos, A., Sinn, P., Schirmacher, P., Marme, F., Jager, D., Kauczor, H.U., Stieber, A., Lindel, K., Debus, J., Golatta, M., Schutz, F., Sohn, C., Heil, J., Schneeweiss, A. (2016) Prognosis Of Breast Cancer Molecular Subtypes In Routine Clinical Care: A Large Prospective Cohort Study. *BMC Cancer* **16**: 734

Herbst, R.S., Shin, D.M. (2002) Monoclonal Antibodies To Target Epidermal Growth Factor Receptor-Positive Tumors: A New Paradigm For Cancer Therapy. *Cancer* **94:** 1593-1611

Herchenhan, A., Uhlenbrock, F., Eliasson, P., Weis, M., Eyre, D., Kadler, K.E., Magnusson, S.P., Kjaer, M. (2015) Lysyl Oxidase Activity Is Required For Ordered Collagen Fibrillogenesis By Tendon Cells. *J Biol Chem* **290**: 16440-16450

Hess, C.B., Niesel, D.W., Cho, Y.J., Klimpel, G.R. (1987) Bacterial Invasion Of Fibroblasts Induced Interferon Production. *J Immunol* **138**: 3949-3953

Higuchi, A., Takanashi, Y., Ohno, T., Asakura, T., Cho, C.S., Akaike, T., Hara, M. (2002) Production Of Interferon-Beta By Fibroblast Cells On Membranes Prepared By Extracellular Matrix Proteins. *Cytotechnology* **39:** 131-137 Hobeika, A.C., Subramaniam, P.S., Johnson, H.M. (1997) Ifnα Induces The Expression Of The Cyclin-Dependent Kinase Inhibitor P21 In Human Prostate Cancer Cells. *Oncogene* **14**: 1165-1170

Honke, N., Shaabani, N., Zhang, D-E., Hardt, C., Lang, K.S. (2016) Multiple Functions Of USP18. *Cell Death & Disease* **7**: E2444

Horisberger, M.A., Gunst, M.C. (1991) Interferon-Induced Proteins: Identification Of Mx Proteins In Various Mammalian Species. *Virology* **180:** 185-190

Hortobágyi, G.N. (1997) Anthracyclines In The Treatment Of Cancer. *Drugs* **54:** 1-7

Hosein, A. N., Livingstone, J., Buchanan, M., Reid, J. F., Hallett, M., Basik, M. (2015) A Functional In Vitro Model Of Heterotypic Interactions Reveals A Role For Interferon-Positive Carcinoma Associated Fibroblasts In Breast Cancer. *BMC Cancer* 15: 130.

Hrdlickova, R., Toloue, M., Tian, B. (2017) RNA-Seq Methods For Transcriptome Analysis. *Wiley Interdiscip Rev RNA* **8:** 1.

Hu, M., Yao, J., Carroll, D.K., Weremowicz, S., Chen, H., Carrasco, D., Richardson, A., Violette, S., Nikolskaya, T., Nikolsky, Y., Baurlein, E.L., Hanh, W.C., Gelman, R.S., Allred, C., Bissell, M.J., Schnitt, S., Polyak, K. (2008) Regulation Of In Situ To Invasive Breast Carcinoma Transition. *Cancer Cell* **13**: 393-406

Hugo, H.J., Lebret, S., Tomaskovic-Rook, E., Ahmed, N., Blick, T., Newgreen, D.F., Thompson, E.W., Ackland, M.L. (2012) Contribution Of Fibroblast And Mast Cell (Afferent) And Tumour (Efferent) IL-6 Effects Within The Tumour Microenvironment. *Cancer Microenvironment* **5**: 83-93

Iorio, M.V., Ferrachin, M., Liu, C-G., Veronese, A., Spizzo, R., Sabbioni, S., Magi, E., Pedriali, M., Fabbri, M., Campiglio, M., Ménard, S., Palazzo, J.P., Rosenberg, A., Musiani, P., Volina, S., Nenci, I., Calin, G.A., Querzoli, P., Negrini, M., Croce, C.M. (2005) Microrna Gene Expression Deregulation In Human Breast Cancer. *Cancer Research* **65**: 7065-7070

Inic, Z., Zegarac, M., Inic, M., Markovic, I., Kozomara, Z., Djurisic, I., Inic, I., Pupic, G., Jancic, S. (2014) Difference Between Luminal A And Luminal B Subtypes According To Ki-67, Tumour Size, And Progesteron Receptor Negativity Providing Prognostic Information. *Clin Med Insights Oncol* **8:** 107-111 Isakoff, S.J. (2010) Triple Negative Breast Cancer: Role Of Specific Chemotherapy Agents. *Cancer Journal* **16:** 53-61

Iseri, O.D., Kars, M.D., Arpaci, F., Gündüz, U. (2010) Gene Expression Analysis Of Drug-Resistant MCF-7 Cells: Implication For Relation To Extracellular Matrix Proteins. *Cancer Chemother Pharmacol* **65**: 447-455

Ishikawa, F., Ushida, K., Mori, K., Shibanuma, M. (2015) Loss Of Anchorage Primarily Induces Non-Apoptotic Deathin A Human Mammary Epithelial Cell Line Under Atypical Focal Adhesion Kinase Signaling. *Cell Death* & Disease 6: E1619

Jansson, M.D., Lund, A.H. (2012) Microrna And Cancer. *Molecular Oncology* **6:** 590-610

Jiang, S., Zhang, H.W., Lu, M.H., He, X.H., Li, Y., Gu, H., Liu, M.F., Wang, E.D. (2010) Microrna-155 Fucntions As An Oncomir In Breast Cancer By Targetting The Suppressor Of Cytokine Signaling 1 Gene. *Cancer Res* **70**: 3119-3127

Johansson, J., Berg, T., Kurzejamska, E., Pang, M.F., Tabor, V., Jansson, M.,Roswall, P., Pietras, K., Sund, M., Religa, P., Fuxe, J. (2013) Mir-155-Mediated Loss Of C/Ebpβ Shifts The TGF-B Response From Growth Inhibition To Epithelial-Mesenchymal Transition, Invasion And Metastasis In Breast Cancer *Oncogene* **32**: 5614-5624

Johnson, S.M., Grosshans, H., Shingara, J., Byrom, M., Jarvis, R., Cheng, A., Labourier, E., Reinert, K.L., Brown, D., Slack, F.J. (2005) Ras Is Regulated By The Let-7 Microrna Family. *Cell* **120:** 635-647

Jones, S.E., Dean, J.C., Young, L.A., Salmon, S.E. (1985) The Human Tumour Clonogenic Assay In Human Breast Cancer. *Journal Of Clinical Oncology* **3:** 92-97

Kaklamani, V.G., Gradishar, W.J. (2003) Epirubicin Vs Doxorubicin: Which Is The Anthracycline Of Choice For The Treatment Of Breast Cancer? *Clin Breast Cancer* **1:** 26-33

Kalluri, R. (2016) The Biology And Function Of Fibroblasts In Cancer. *Nature Reviews Cancer*, **16:** 582.

Kalluri, R. & Zeisberg, M. (2006) Fibroblasts In Cancer. *Nat Rev Cancer,* **6**:392-401.

Kalra, J., Anantha, M., Warburton, C., Waterhouse, D., Yan, H., Yang, Y., Strut, D., Osooly, M., Masin, D., Bally, M.B. (2011) Validating The Use Of A Luciferase Labeled Breast Cancer Cell Line, MDA435LLC6, As A Means To Monitor Tumour Progression And To Assess The Therapeutic Activity Of An Established Anticancer Drug, Docetaxel (Dt) Alone Or In Combination With ILK Inhibitor, QLT0267. *Cancer Biol Ther* **11**: 826-838

Karimian, A., Ahmadi, Y., Yousefi, B. (2016) Multiple Functions Of P21 In Cell Cycle, Apoptosis And Transcriptional Regulation After DNA Damage. *DNA Repair (Amst)* **42:** 63-71

Kassam, F., Enright, K., Dent, R., Dransitsaris, G., Myers, J., Flynn, C., Fralick, M., Kumar, R., Clemons, M. (2009) Survival Outcomes For Patients With Metastatic Triple Negative Breast Cancer: Implications For Clinical Practice And Trial Design. *Clinical Breast Cancer* **9**: 29-33

Katayama, T., Nakanishi, K., Nishihara, H., Kamiyama, N., Nakagawa, T., Kamiyama, T., Iseki, K., Tanaka, S., Todo, S. (2007) Type I Interferon Prolongs Cell Cycle Progression Via P21waf1/CIP1 Induction In Human Colon Cancer Cells. *Int J Oncol,* 31, 613-20.

Kaufmann, S.H., Earnshaw, W.C (2000) Induction Of Apoptosis By Cancer Chemotherapy. *Exp Cell Res* **256**: 42-49

Kaur, S., Sassano, A., Dolniak, B., Joshi, S., Majchrzak-Kita, B., Baker, D.P., Hay, N., Fish, E.N., Platanias, L.C. (2008) Role Of The Akt Pathway In Mrna Translation Of Interferon Stimulated Genes. *Proc Nati Acad Sci USA* **105**: 4808-4813

Kaur, S., Sassano, A., Joseph, A.M., Majchrzak-Kita, B., Eklund, E.A., Verma, A., Brachmann, S.M., Fish, E.N., Platanias, L.C. (2008) Dual Regulatory Roles Of Phosphatidylinositol 3-Kinase In IFN Signaling. *J Immunol* **181:** 7316-7323

Kawada, K., Yonei, T., Ueoka, H., Kiura, K., Tabata, M., Takigawa, N., Harada, M., Tanimoto, M. (2002) Comparison Of Chemosensitivity Tests: Clonogenic Assay Versus MTT Assay. *Acta Medica Okayama* **56:** 129-134

Kawamata, T., Seitz, H., Tomari, Y. (2009) Structural Determinants Of Mirnas For RISC Loading And Slicer-Independent Unwinding. *Nat Struct Mol Biol* **16:** 953-960
Kerr, I.M., Brown, R.E. (1978) Pppa2'p5'a2'p5'a: An Inhibitor Of Protein Synthesis Synthesized With An Enzyme Fraction From Interferon-Treated Cells. *PNAS* **75**: 256-260

Kessenbrock, K., Plaks, V., Werb, Z. (2010) Matrix Metalloproteinases: Regulators Of The Tumour Microenvironment. *Cell* **141:** 52-67

Kharaishvili, G., Simkova, D., Bouchalova, K., Gachechiladze, M., Narsia, N., Bouchal, J. (2014) The Role Of Cancer-Assocated Fibroblasts, Solid Stress And Other Microenvironmental Factors In Tumour Progression And Therapy Resistance. *Cancer Cell International* **14**: 1-8

Kikuchi, Y., Kashima, T.G., Nishiyama, T., Shimazu, K., Morishita, Y., Shimazaki, M., Kii, I., Horie, H., Nagai, H., Kudo, A., Fukayama, M. (2008) Periostin Is Expressed In Pericryptal Fibroblasts And Cancer-Associated In The Colon. *J Histochem Cytochem* **56**: 753-764

Kim, B., Fatayer, H., Hanby, A.M., Horgan, K., Perry, S.L., Valleley, E.M., Verghese, E.T., Williams, B.J., Thorne, J.L., Hughes, T.A. (2013) Neoadjuvant Chemotherapy Induces Expression Levels Of Breast Cancer Resistance Protein That Predict Disease-Free Survival In Breast Cancer. *Plos One* **8**: E62766

Kim, B., Stephen, S.L., Hanby, A.M., Horgan, K., Perry, S.L., Richardson, J., Roundhill, E.A., Valleley, E.M., Verghese, E.T., Williams, B.J., Thorne, J.L., Hughes, T.A. (2015) Chemotherapy Induced Notch1-Dependent MRP1 Up-Regulation, Inhibition Of Which Sensitizes Breast Cancer Cells To Chemotherapy. *BMC Cancer* **15**: 634

Kim, Y-K., Kim, V.N. (2007) Processing Of Intronic Micrornas. *EMBO J* **26:** 775-783

Knabbe, C., Lippman, M.E., Wakefield, L.M., Flanders, K.C., Kasid, A., Derynck, R., Dickson, R.B. (1987) Evidence That Transforming Growth Factor-B Is Hormonally Regulated Negative Growth Factor In Human Breast Cancer Cells. *Cell* **48**: 417-428

Kojima, Y., Acar, A., Eaton, E.N., Mellody, K.T., Scheel, C., Ben-Porath, I., Onder, T.T., Wang, Z.C., Richardson, A.L., Weinberg, R.A, Orimo, A. (2010) Autocrine TGF-Beta And Stromal Cell-Derived Factor-1 (SDF-1) Signaling Drives The Evolution Of Tumour-Promoting Mammary Stromal Myofibroblasts. *Proc Nati Acad Sci USA* **16**: 20009-20014

Kotenko, S.V., Gallagher, G., Baurin, V.V., Lewis-Antes, A., Shen, M.,

Shah, N.K., Langer, J.A., Sheikh, F., Dickensheets, H., Donnelly, R.P. (2003) IFN-Lambdas Mediate Antiviral Protection Through A Distinct Class II Cytokines Receptor Complex. *Nat Immunol* **4:** 69-77

Kovalev, A.A., Tsvetaeva, D.A., Grudinskaja, T.V. (2013) Role Of ABC-Cassette Transporters (MDR1, MRP1, BCRP) In The Development Of Primary And Acquired Multiple Drug Resistance In Patients With Early And Metastatic Breast Cancer. *Exp Oncol* **35**: 287-290

Kratzke, R.A., Kramer, B.S. (1996) Evaluation Of In Vitro Chemosensitivity Using Human Lung Cancer Cell Lines. *Journal Of Cell Biochem Suppl* **24:** 160-164

Krol, J., Sobczak, K., Wilczynska, U., Drath, M., Jasinska, A., Kaczynska, D., Krzyzosiak, W.J. (2004) Structural Features Of Microrna (Mirna) Precursors And Their Relevance To Mirna Biogenesis And Small Interfering RNA/Short Hairpin RNA Design. *J Biol Chem* **279**: 42230-42239

Kukurba, K. R., Montgomery, S. B. (2015) RNA Sequencing And Analysis. *Cold Spring Harbor Protocols* **2015**: 951-969.

Kumar, R., Sharma, A. & Tiwari, R. K. (2012) Application Of Microarray In Breast Cancer: An Overview. *J Pharm Bioallied Sci* **4**: 21-6.

Kuroda, H., Takeno, M., Murakami, S., Miyazawa, N., Kaneko, T., Ishigatsubo, Y. (2010) Inhibition Of Heme Oxygenase-1 With An Epidermal Growth Factor Receptor Inhibitor And Cisplatin Decreases Proliferation Of Lung Cancer A549 Cells. *Lung Cancer* **67:** 31-36

Lacroix, M. (2006) Significance, Detection And Markers Of Disseminated Breast Cancer Cells. *Endocrine Related Cancer* **13**: 1033-1067

Lagios, M.D., Margolin, F.R., Westdahl, P.R., Rose, M.R. (1989) Mammographically Detected Duct Carcinoma In Situ. Frequency Of Local Recurrence Following Tyletomy And Prognostic Effect Of Nuclear Grade On Local Recurrence. *Cancer* **63**: 618-624

Lagos-Quintane, M., Rauhut, R., Lendeckel, W., Tuschi, T. (2001) Identification Of Novel Genes Coding For Small Expressed Rnas. *Science* **294**: 853-858

Lakhani SR, Ellis IO, Schnitt SJ, Tan PH, Van De Vijver MJ, Editors. WHO Classification Of Tumours Of The Breast. Fourth Ed. IARC; Lyon: 2012. ISBN.13. Lee, Y., Ahn, C., Han, J., Choi, H., Kim, J., Lee, J., Provost, P., Radmark, O., Kim, S., Kim, V. N. (2003) The Nuclear Rnase III Dorsha Initiates Microrna Processing. *Nature* **425:** 415-419

Lee, Y., Hur, I., Park, S.Y., Kim, Y.K., Suh, M.R., Kim, V.N. (2006) The Role Of PACT In The RNA Silencing Pathway. *EMBO J* **25**: 522-532

Lee, Y., Kim, M., Han, J., Yeom, K.H., Lee, S., Baek, S.H., Kim, V.N. (2004) Microrna Genes Are Transcribed By RNA Polymerase II. *EMBO J* 23: 4051-4060

Le Page, C., Génin, P., Baines, M.G., Hiscott, J. (2000) Interferon Activation And Innate Immunity. *Rev Immunogenet* **2:** 374-386

Legler, J., Zeinstra, L.M., Schuitemaker, F., Lanser, P.H., Bogerd, J., Brouwer, A., Vethaak, A.D., De Voogt, P., Murk, A.J., Van Der Burg, B. (2002) Comparison Of In Vivo And In Vitro Reporter Gene Assays For Short-Term Screening Of Estrogenic Activity. *Environ Sci Technol* **36**: 4410-4415

Legrier, M-E., Bièche, I., Gaston, J., Beurdeley, A., Yvonnet, V., Déas, O., Thuleau, A., Château-Joubert, S., Servely, J-L., Vacher, S., Lassalle, M., Depil, S., Tucker, G.C., Fontaine, J-J., Poupon, M-F., Roman-Roman, S., Judde, J-G., Decaudin, D., Cairo, S., Marangoni, E. (2015) Activation Of IFN/STAT1 Signaling Predicts Response To Chemotherapy In Oestrogen Receptor-Negative Breast Cancer. *British Journal Of Cancer* **114**: 177-187

Lehmann, B.D., Jovanovic, B., Chen, X., Estrada, M.V., Johnson, K.N., Shyr, Y., Moses, H.L., Sanders, M.E., Pietenpol, J.A. (2016) Refinement Of Triple Negative Breast Cancer Molecular Subtypes: Implications For Neoadjuvant Chemotherapy Selection. *Plos One* **16:** E0157368

Lehmann, B.D., Bauer, J.A., Schafer, J.M., Pendleton, C.S., Tang, L., Johnson, K.C., Chen, X., Balko, J.M., Gomez, H., Arteaga ,C.L., Mills, G.B., Sanders, M.E., Pietenpol, J.A. (2014) PIK3CA Mutations In Androgen-Receptor Positive Triple-Negative Breast Cancer Confer Sensitivity To The Combination Of PI3K And Androgen Receptor Inhibitors. *Breast Cancer Research* **16**: 406

Lehmann, B.D., Bauer, J.A., Chen, X., Sander, M.E., Chakavarthy, A.B., Shyr, Y., Pietenpol, J.A. (2011) Identification Of Human Triple-Negative Breast Cancer Subtypes And Preclinical Models For Selection Of Targeted Therapies. *Journal Of Clinical Investigation* **121**: 2750-2767 Lehmann, B.D., Pietenpol, J. A. & Tan, A. R. (2015) Triple-Negative Breast Cancer: Molecular Subtypes And New Targets For Therapy. *Am Soc Clin Oncol Educ Book*, E31-9.

Levental, K.R., Yu, H., Kass, L., Lakins, J.N., Egeblad, M., Erler, J.T., Fong, S.F., Csiszar, K., Giaccia, A., Weninger, W., Yamauchi, M., Gasser, D.L., Weaver, V.M. (2009) Matrix Crosslinking Forces Tumor Progression By Enhancing Integrin Signaling. *Cell* **139**: 891-906

Leyh, B., Dittmer, A., Lange, T., Martens, J.W.M., Dittmer, J. (2015) Stromal Cells Promote Anti-Estrogen Resistance Of Breast Cancer Cells Through An Insulin-Like Growth Factor Binding Protein 5 (IGFBP5)/B-Cell Leukemia/Lymphoma 3 (Bcl-3 Axis). *Oncotarget* **6:** 39307- 39328

Li, S., Shen, Y., Wang, M., Yang, J., Meng, L., Li, P., Chen, Z., Yang, J. (2017) Loss Of PTEN Expression In Breast Cancer: Association With Clinicopathological Characteristics And Prognosis. *Oncotarget* **8**: 32043-32054

Li, X., Gu, W., Mohan, S., Baylink, D. J. (2002) DNA Microarrays: Their Use And Misuse. *Microcirculation* **9:** 13-22.

Liedtke, C., Mazouni, C., Hess, K.R., Andre, F., Tordai, A., Mejia, J.A., Symmans, F.W., Gonzalez-Angulo, A.M., Hennessy, B., Green, M., Cristofanilli, M., Hortobagyi, G.N., Pusztai, L. (2008) Response To Neoadjuvant Therapy And Long-Term Survival In Patients With Triple Negative Breast Cancer. *Journal Of Clinical Oncology* **26**: 1275-1281

Lin., S., Miller, J.D., Ying, S. (2006) Intronic Microrna (Mirna) *Journal Of Biomedicine And Biotechnology* **26818:** 1-13

Littlepage, L.E., Egeblad, M., Werb, Z. (2005) Coevolution Of Cancer And Stromal Cellular Responses. *Cancer Cell* **7:** 499-500

Liu, C. G., Calin, G. A., Volinia, S., Croce, C. M. (2008) Microrna Expression Profiling Using Microarrays. *Nat Protoc,* **3:** 563-78.

Liu, J., Carmell, M.A., Rivas, F.V., Marsden, C.G., Thomson, J.M., Song, J.J., Hammond, S.M., Joshua-Tor, L., Hannon, G.J. (2004) Argonaute2 Is The Catalytic Engine Of Mammalian Rnai. *Science* **305**: 1437-1441

Liu, J., Huang, W., Yang, H., Luo, Y. (2015) Expression And Function Of Mir-155 In Breast Cancer. *Biotechnology & Biotechnological Equipment* **29**: 840-843. Livak, K.J., Schmittgen, T.D. (2001) Analysis Of Relative Gene Expression Data Using Real-Time Quantitative PCR And The 2(-Delta Delta C(T)) Method. *Methods* **25:** 402-408

Lorger, M., Felding-Habermann, B. (2010) Capturing Changes In The Brain Microenvironment During Initial Steps Of Breast Cancer Brain Metastasis. *AM J Pathol* **176:** 2958-2971

Lotti, F., Jarrar, A. M., Pai, R. K., Hitomi, M., Lathia, J., Mace, A., Gantt, G. A., Sukhdeo, K., Devecchio, J., Vasanji, A., Leahy, P., Hjelmeland, A. B., Kalady, M. F., Rich, J. N. (2013) Chemotherapy Activates Cancer-Associated Fibroblasts To Maintain Colorectal Cancer Initiating Cells By IL-17A. *The Journal Of Experimental Medicine* **210**: 2851-2872.

Lu, P., Weaver, V.M., Werb, Z. (2012) The Extracellular Matrix: A Dynamic Niche In Cancer Progression. *J Cell Biol* **196:** 395-406

Lu, L., Zhang, L., Wai, M.S., Yew, D.T., Xu, J. (2012) Exocytosis Of MTT Formazan Could Exacerbate Cell Injury. *Toxicol In Vitro* **26:** 636-644

Lund, E., Güttinger, S., Calado, A., Dahlberg, J.E., Kutay, U. (2004) Nuclear Export Of Microrna Precursors. *Science* **303**: 95-98

Ma, X., Helgason, E., Phung, Q.T., Quan, C.L., Iyer, R.S., Lee, M.W., Bowman, K.K., Starovasnik, M.A., Dueber, E.C. (2012) Molecular Basis Of Tank-Binding Kinase 1 Activation By Transautophosphorylation. *Proc Nati Acad Sci USA* **109:** 9378-9383

Macfarlane, L.-A. & Murphy, P. R. (2010) Microrna: Biogenesis, Function And Role In Cancer. *Current Genomics*, **11:** 537-561.

Madar, S., Goldstein, I., Rotter, V. (2013) 'Cancer Associated Fibroblasts'- More Than Meets The Eye. *Trends In Molecular Medicine* **19**: 447-453

Makki, J. (2015) Diversity Of Breast Carcinoma: Histological Subtypes And Clinical Relevance. *Clinical Medicine Insights. Pathology* **8:** 23-31.

Malhorta, G. K., Zhao, X., Band, H. & Band, V. (2010) Histological, Molecular And Functional Subtypes Of Breast Cancers. *Cancer Biology & Therapy* **10:** 955-960. Mankouri, J., Fragkoudis, R., Richards, K.H., Wetherill, L.F., Harris, M., Kohl, A., Elliott, R.M., Macdonald, A. (2010) Optineurin Negatively Regulates The Induction Of Ifnbeta In Response To RNA Virus Infection. *Plos Pathog* **19**: E1000778

Mao, Y., Keller, E. T., Garfield, D. H., Shen, K. & Wang, J. (2013) Stromal Cells In Tumor Microenvironment And Breast Cancer. *Cancer Metastasis Rev*, **32:** 303-15.

Marthandan, S., Priebe, S., Hemmerich, P., Klement, K., Diekmann, S. (2014) Long-Term Quiescent Fibroblasts Cells Transit Into Senescence. *Plos One* **9**: E115597

Martin, E,C., Krebs, A.E., Burks, H.E., Elliott, Maddoo, M., Collins-Burrow, B.M., Flemington, E.K., Burow, M.E. (2014) Mir-155 Induced Transcriptome Changes In The MCF-7 Breast Cancer Cell Line Leads To Enhance Mitogen Activated Protein Kinase Signaling. *Genes Cancer* **5**: 353-364

Martinez-Outschoorn, U.E., Goldberg, A., Lin, Z., Ko, Y.H., Flomenberg, N., Wang, C., Pavlides, S., Pestell, R.G., Howell, A., Sotgia, F., Lisanti, M.P. (2011) Anti-Estrogen Resistance In Breast Cancer Is Induced By The Tumour Microenvironment And Can Be Overcome By Inhibiting Mitochondrial Function In Epithelial Cancer Cells. *Cancer Biol Ther* **12**: 924-938

Masoud, V., Pagès, G. (2017) Targeted Therapies In Breast Cancer: New Challenges To Fight Against Resistance. *World J Clin Oncol* 8: 120-134 Matos, I., Dufloth, R., Alvarenga, M., Zeferino, L.C., Schmitt, F. (2005)

P63, Cytokeratin 5, And P-Cadherin: Three Molecular Markers To Distinguish Basal Pheno Type In Breast Carcinomas. *Virchows Arch* **447**: 688-694

Mastracci, T.L., Tjan, S., Bane, A.L., O'Malley, F.P., Andrulis, I.L. (2005) E-Cadherin Alterations In Atypical Lobular Hyperplasia And Lobular Carcinoma In Situ Of The Breast. *Mod Pathol* **18:** 741-751

Matsumoto, K., Nakamura, T. (2006) Hepatocyte Growth Factor And The Met System As A Mediator Of Tumour Stromal Interactions. *International Journal Of Cancer* **119:** 477-483

Mattiske, S., Suetani, R. J., Neilsen, P. M., Callen, D. F. (2012) The Oncogenic Role Of Mir-155 In Breast Cancer. *Cancer Epidemiolbiomarkers Prev* **21:** 1236-43.

Mavaddat, N., Barrowdale, D., Andrulisnano, I. L., Domchek, S. M., Eccles, D., Nevanlinna, H., Ramus, S. J., Spurdle, A., Robson, M., Sherman, M., Mulligan, A. M., Couch, F. J., Engel, C., Mcguffod, L., Healey, S., Sinilnikova, O. M., Southey, M. C., Terry, M. B., Goldgar, D., O'Malley, F., John, E. M., Janavicius, R., Tihomirova, L., Hansen, T. V. O., Nielsen, F. C., Osorio, A., Stavropoulou, A., Benitez, J., Manoukian, S., Peissel, B., Barile, M., Volorio, S., Pasini, B., Dolcetti, R., Putignano, A. L., Ottini, L., Radice, P., Hamann, U., Rashid, M. U., Hogervorst, F. B., Kriege, M., Van Der Luijt, R. B., Peock, S., Frost, D., Evans, D. G., Brewer, C., Walker, L., Rogers, M. T., Side, L. E., Houghton, C., Weaver, J., Godwin, A. K., Schmutzler, R. K., Wappenschmidt, B., Meindl, A., Kast, K., Arnold, N., Niederacher, D., Sutter, C., Deissler, H., Gadzicki, D., Preisler-Adams, S., Varon-Mateeva, R., Schonbuchner, I., Gevensleben, H., Stoppa-Lyonnet, D., Belotti, M., Barjhoux, L., Isaacs, C., Peshkin, B. N., Caldes, T., De Al Hoya, M., Canadas, C., Heikkinen, T., Heikkila, P., Aittomaki, K., Blanco, I., Lazaro, C., Brunet, J., Agnarsson, B. A., Arason, A., Barkardottir, R. B., Dumont, M., Simard, J., Montagna, M., Agata, S., D'Andrea, E., Yan, M., Fox, S., Rebbeck, T. R., Rubinstein, W., Tung, N., Garber, J. E., Wang, X., Fredericksen, Z., Pankratz, V. S., Lindor, N. M., Szabo, C., Offit, K., Sakr, R. (2012) Pathology Of Breast And Ovarian Cancers Among BRCA1 And BRCA2 Mutation Carriers: Results From The Consortium Of Investigators Of Modifiers Of BRCA1/2 (CIMBA). Cancer Epidemiology, Biomarkers & Prevention : A Publication Of The American Association For Cancer Research, Cosponsored By The American Society Of Preventive Oncology, 21: 134-147

Maycotte, P., Aryal, S., Cummings, C.T., Thorburn, J., Morgan, M.J., Thorburn, A. (2012) Chloroquine Sensitizes Breast Cancer Cells To Chemotherapy Independent Of Autophagy. *Autophagy* **8:** 200-212

McDonnell, D.P., Wardell, S.E. (2010) The Molecular Mechanisms Underlying The Pharmacological Actions Of ER Modulators: Implications For New Drug Discovery In Breast Cancer. *Current Opinion In Pharmacology* **10**: 620-628

McGowan, E.M., Ailing, N., Jackson, E.A., Yagoub, D., Haass, N.K., Allen, J.D., Martinello-Wilks, R. (2011) Evaluation Of Cell Cycle Arrest In Estrogen Responsive MCF-7 Breast Cancer Cells: Pitfalls Of The MTS Assay. *Plos One* **6:** E20623

McIlwain, D.R., Berger, T., Mak, T.W. (2013) Caspase Functions In Cell Death And Disease. *Cold Spring Harb Perspect Biol* **5:** A008656

Mechetner, E., Kyshtoobayeva, A., Zonis, S., Kim, H., Stroup, R., Garcia, R., Parker, R.J., Fruehauf, J.P. (1998) Levels Of Multidrug Resistance (MDR1) P-Glycoprotein Expression By Human Breast Cancer Correlate With In Vitro Resistance To Taxol And Doxorubicin. *Clin Cancer Res* **4**: 389-398

Medzhitov, R., Janeway Jr, C.A. (1997) Innate Immunity: Impact On The Adaptive Immune Response. *Curr Opin Immuno* **9**: 4-9

Mehrgou, A., Akouchekian, M. (2017) Therapeutic Impacts Of Micrornas In Breast Cancer By Their Roles In Regulating Processes Involved In This Disease. *Journal Of Research In Medical Sciences : The Official Journal Of Isfahan University Of Medical Sciences* **22**: 130.

Mei, M., Xie, D., Zhang, Y., Jin, J., You, F., Yan, L., Dai, J., Chen, X. (2014) A New 2α,5α,10β,14β-Tetraacetoxy-4(20),11-Taxadiene (SIA) Derivative Overcomes Paclitaxel Resistance By Inhibiting MAPK Signaling And Increasing

Paclitaxel Accumaulation In Breast Cancer Cells. *Plos One* **9**: E104317 Meinke, A., Barahmand-Pour, F., Wöhrl, S., Stoiber, D., Decker, T. (1996) Activation Of Different Stat5 Isoforms Contributes To Cell-Type-Restricted Signaling In Response To Interferons. *Mol Cell Biol* **16**: 6937-6944

Meister, G., Landthaler, M., Dorsett, Y., Tuschi, T. (2004) Sequence-Specific Inhibition Of Microrna- And Sirna- Induced RNA Silencing. *RNA* 544-550

Meister, G., Landthaler, M., Patkaniowska, A., Dorsett, Y., Teng, G., Tuschi, T. (2004) Human Argonaute2 Mediates RNA Cleavage Targeted By Mirnas And Sirnas. *Mol Cell* **15:** 185-197

Mendell, J.T. (2008) Miriad Roles For The Mir-17-92 Cluster In Development And Disease. *Cell* **2:** 217-222

Menyhart, O., Harami-Papp, H., Sukumar, S., Schafer, R., Magnani, L., De Barrios, O., Gyorffy, B. (2016) Guidelines For The Selection Of Functional Assays To Evaluate The Hallmarks Of Cancer. *Bioechemica Et Biophysica Acta* **1866:** 300-319

Merlino, G., Miodini, P., Callari, M., D'aiuto, F., Cappelletti, V.,

Daidone, M. G. (2017) Prognostic And Functional Role Of Subtype-Specific Tumor-Stroma Interaction In Breast Cancer. *Mol Oncol* **11**: 1399-1412.

Messina, J.L., Yu, H., Riker, A.I., Munster, P.N., Jove, R.L., Daud, A.I. (2008) Activated Stat-3 In Melanoma. *Cancer Control* **15:** 196-201

Miklossy, G., Hilliard, T. S., Turkson, J. (2013) Therapeutic Modulators Of STAT Signaling For Human Diseases. *Nat Rev Drug Discov,* **12:** 611-29.

Mirtti, T., Leiby, B.E., Abdulghani, J., Aaltonen, E., Pavela, M., Mamtani, A., Alanen, K., Egevad, L., Granfors, T., Josefsson, A., Stattin, P., Bergh, A., Nevalainen, M.T. (2013) Nuclear Stat5a/B Predicts Early Recurrence And Prostate Cancer-Specific Death In Patients Treated By Radical Prostatectomy. *Hum Pathol* **44**: 310-319

Mitra, A.K., Zillhardt, M., Hua, Y., Tiwari, P., Murmann, A.E., Peter, M.E., Lengyel, E. (2012) Microrna Reprogram Normal Fibroblasts Into Cancer-Associated Fibroblasts In Ovarian Cancer. *Cancer Discovery* **2:** 1100-1108

Mitra, A.K., Zillhardt, M., Hua, Y., Tiwari, P., Murmann, A.E., Peter, M.E., Lengyel, E. (2012) Micrornas Reprogram Normal Fibroblasts Into Cancer Associated Fibroblasts In Ovarian Cancer. *Cancer Disco* **2**: 1100-1108

Miyoshi, Y., Ando, A., Takamura, Y., Taguchi, T., Tamaki, Y., Noguchi, S. (2002) Prediction Of Response To Docetaxel By CYP3A4 Mrna Expression In Breast Cancer Tissues. *Int J Cancer* **97**: 129-132

Mogensen, K.E., Lewerenz, M., Reboul, J., Lutfalla, G., Uzé, G. (1999) The Type I Interferon Receptor: Structure, Function, And Evolution Of A Family Business. *J Interferon Cytokine Res* **19**: 1069-1098

Mohammad, R.M., Mugbil R., Lowe, L., Yedjou, C., Hsu, H.Y., Lin, L.T., Sieglin, M.D., Fimognari, C., Kumar, N.B., Dou, P.Q., Yang, H., Samadi, A.K., Russo, G.L., Spagnuolo, C., Ray, S.K., Chakrabarti, M., Morre, J.D., Coley, H.M., Honoki, K., Fujii, H., Georgakilas, A.G., Amedei, A., Niccolai, E., Amin, A., Ashraf, S.S., Helferich, W.G., Yang, X., Boosani, C.S., Guha, G., Bhakta, D., Ciriolo, M.R., Aquilano, K., Chen, S., Mohammed, S.I., Keith, W.N., Bilsland, A., Halicka, D., Nowsheen, S., Azmi, A.S. (2016) Broad Targeting Of Resistance To Apoptosis In Cancer. *Smin Cancer Biol* **35**: S78-103

Montesano, R., Orci, L. (1998) Transforming Growth Factor Beta Stimulates Collagen-Matrix Contraction By Fibroblasts: Implications For Wound Healing. PNAS 85: 4894-4897

Moll, H. P., Freudenthaler, H., Zommer, A., Buchberger, E., Brostjan, C. (2008) Neutralizing Type I Interferon Antibodies Trigger An Interferon-Like Response In Endothelial Cells. *Journal Of Immunology (Baltimore, Md. : 1950)*, **180:** 5250-5256.

Munger, W., Dejoy, S.Q., Jeyaseelan, R. Sr., Torley, L.W., Grabstein, K.H., Eisenmann, J., Paxton, R., Cox, T., Wick, M.M., Kerwar, S.S. (1995) Studies Evaluating The Anti-Tumour Activity And Toxicity Of Interleukin-15, A New T Cell Growth Factor: Comparison With Interleukin-2. *Cellular Immunology* **165:** 289-293

Moorman, A.M., Vink, R., Heijmans, H.J., Van Der Palen, J., Kouwenhoven, E.A. (2012) The Prognostic Value Of Tumour—Stroma Ratio In Triple Negative Breast Cancer. *Eur J Surg Oncol* **38:** 307-313

Morrow, M., Jagsi, R., Alderman, A.K., Griggs, J.J., Hawley, S.T., Hamilton, A.S., Graff, J.J., Katz, S.J. (2009) Surgeon Recommendations And Receipt Of Mastectomy For Treatment Of Breast Cancer. *JAMA* **302**: 1551-1556

Mossman, T. (1983) Rapid Colorimetric Assay For Cellular Growth And Survival: Application To Proliferation And Cytotoxicity Studies. *Journal Of Immunological Methods* **16:** 55-63

Mueller, K.L., Madden, J.M., Zoratti, G.L., Kuperwasser, C., List, K., Boerner, J.L. (2012) Fibroblast-Secreted Hepatocyte Growth Factor Mediates Epidermal Growth Factor Receptor Tyrosine Kinase Inhibitor Resistance In Triple Negative Breast Cancers Through Paracrine Activation Of Met. *Breast Cancer Research* **14**: 104

Muscella, A., Vetrugno, C., Fanizzi, F.P., Manca, C., De Pascali, S.A., Marsigliante, S. (2013) A New Platinum (II) Compound Anticancer Drug Candidate With Selective Cytotoxicity For Breast Cancer Cells. *Cell Death & Disease* **4**: E796

Nabholtz, J.M., Gligorov, J. (2005) The Role Of Taxanes In The Treatment Of Breast Cancer. *Expert Opin Pharmacother* **6:** 1073-1094

Nagalakshmi, U., Waern, K., Snyder, M, (2010) RNA-Seq: A Method For Comprehensive Transcriptome Analysis. *Curr Protoc Mol Biol* **4**: 1-13 National Cancer Registration & Analysis Service And Cancer Research UK: "Chemotherapy, Radiotherapy And Tumoour Resections In England: 2013-2014" Workbook. London: NCRAS:2017

Ness, S.A. (2007) Microarray Analysis: Basic Strategies For Successful Experiments. *Mol Biotechnol* **36:** 205-219

Netti, P.A., Berk, D.A., Swartz, M.A., Grodzinsky, A.J., Jain, R.K. (2000) Role Of Extracellular Matrix Assembly In Interstitial Transport In Solid Tumours. *Cancer Research* **60:** 2497-2503

Neve, R.M., Chin, K., Fridlyand, J., Yeh, J., Baehner, F.L., Fevr, T., Clark, L., Bayani, N., Coppe, J.P., Tong, F., Speed, T., Spellman, P.T., Devries, S., Lapuk, A., Wang, N.J., Kuo, W.L., Stilwell, J.L., Pinkel, D., Albertson, D.G., Waldman, F.M., Mccormick, F., Dickson, R.B., Johnson, M.D., Lippman, M., Ethier, S., Gazdar, A., Gray, J.W. (2006) A Collection Of Breast Cancer Cell Lines For The Study Of Functionally Distinct Cancer Subtypes. *Cancer Cell* **10**: 515-527

NICE Guidelines 2009 -

<u>Https://Www.Nice.Org.Uk/Guidance/Cg81/Chapter/Recommendations#Systemi</u> <u>c-Disease-Modifying-Therapy</u>. Accessed October 2018

NICE Guidelines 2018- <u>Https://Www.Nice.Org.Uk/Guidance/Ng10</u>. Accessed October 2018

Nielsen, T.O., Hsu, F.D., Jensen, K., Cheang, M., Karaca, G., Hu, Z., Hernandez-Boussard, T., Livasy, C., Cowan, D., Dressler, L., Akslen, L.A., Ragaz, J., Gown, A.M., Gilks, C.B., Van De Rijn, M., Perou, C.M. (2004) Immunohistochemical And Clinical Characterization Of The Basal-Like Subtype Of Invasive Breast Carcinoma. *Clin Cancer Res* **10**: 5367-5374

Nolvadex Adjuvant Trial Organisation. (1988) Controlled Trial Of Tamoxifen As A Single Adjuvant Agent In The Management Of Early Breast Cancer. *Br J Cancer* **57:** 608-611

Nomura, Y., Tashiro, H., Hisamatsu, K. (1989) In Vitro Clonogenic Growth And Metastatic Potential Of Human Operable Breast Cancer. *Cancer Research* **49**: 5288- 5293

O'Donnell, K.A., Wentzel, E.A., Zeller, K.I., Dang, C.V., Mendell, J.T. (2005) C-Myc-Regulated Micrornas Modulate E2F1 Expression. *Nature* **435**: 839-843

Okuma, H. S., Koizumi, F., Hirakawa, A., Nakatochi, M., Komori, O., Hashimoto, J., Kodaira, M., Yunokawa, M., Yamamoto, H., Yonemori, K., Shimizu, C., Fujiwara, Y., Tamura, K. (2016) Clinical And Microarray Analysis Of Breast Cancers Of All Subtypes From Two Prospective Preoperative Chemotherapy Studies. *Br J Cancer* **115**: 411-9.

O'Neill, L.A.J., Bowie, A.G. (2007) The Family Of Five: TIR-Domain-Containing Adaptors In Toll-Like Receptor Signaling. *Nature Reviews Immunology* **7:** 353-364

O'Reilly, E.A., Gubbins, L., Sharma, S., Tully, R., Guang, M.H.Z., Weiner-Gorzel, K., Mccaffrey, J., Harrison, M., Furlong, F., Kell, M., Mccann, A. (2015) The Fate Of Chemoresistance In Triple Negative Breast Cancer. *BBA Clinical* **3**: 257-275

Orimo, A., Gupta, P.B., Sgroi, D.C., Arenzana-Seisdedos, F., Delaunay, T., Naeem, R., Carey, V.J., Richardson, A.L., Weinberg, R.A. (2005) Stromal Fibroblasts Present In Invasive Human Breast Carcinomas Promote Tumour Growth And Angiogenesis Through Elevated SDF- 1/Cxcl12 Secretion. *Cell* **121:** 335-348

O'Shaughnessy, J., Demichele, A., Ma, C. X., Richards, P., Yardley, D. A., Wright, G. S., Kalinsky, K., Steis, R., Diab, S., Kennealey, G., Geschwindt, R., Jiang, W.,Rugo, H. S. (2018) A Randomized, Double-Blind, Phase 2 Study Of Ruxolitinib Or Placebo In Combination With Capecitabine In Patients With Advanced HER2-Negative Breast Cancer And Elevated C-Reactive Protein, A Marker Of Systemic Inflammation. *Breast Cancer Res Treat*, **170**: 547-557.

Osheroff, N. (1989) Biochemical Basis For The Interactions Of Type I And Type II Topoisomerases With DNA. *Pharmacology & Therapeutics* **41**: 223-241

Ostman, A., Augsten, M. (2009) Cancer-Associated Fibroblasts And Tumour Growth-Bystanders Turning Into Key Players. *Current Opinions In Genetics And Development* **19:** 67-73

Ottewell, P.D., Lefley, D.V., Coleman, R.E., Holen, I. (2008) Mechanisms Of Apoptosis And Cell-Cycle Arrest In Subcutaneous Breast Tumours Treated Sequentially With Doxorubicin Followed By Zoledronic Acid. *Breast Cancer Research* **10:** 81 Overmoyer, B., Regan, M., Polyak, K., Brock, J., Van Poznak, C., King, T., Haddad, T., Stearns, V., Hwang, S., Winer, E. (2018) Abstract OT3-05-01: TBCRC 039: Phase II Study Of Combination Ruxolitinib (INCB018424) With Preoperative Chemotherapy For Triple Negative Inflammatory Breast Cancer. *Cancer Research*, **78:** OT3-05-01.

Ouyang, M., Li, Y., Ye, S., Ma, J., Lu, L., LV, W., Chang, G., Li, X., Li, Q., Wang, S., Wang, W. (2014) Microrna Profiling Implies New Markers Of Chemoresistance Of Triple-Negative Breast Cancer. *Plos One* **9**: E96228

Pace, L. (2010) APC Activation By IFN-Alpha Decreases Regulatory T Cell And Enhances Th Cell Functions. *Journal Of Experimental Medicine* **184**: 5969-5979

Page, C., Lin, H.J., Jin, Y., Castle, V.P., Nunez, G., Huang, M., Lin, J. (2000) Overexpression Of Akt/AKT Can Modulate Chemotherapy-Induced Apoptosis. *Anticancer Res* **20**: 407-416

Page, D.L. (2003) Special Types Of Invasive Breast Cancer, With Clinical Implications. *Am J Surg Pathol* **27:** 832-835

Pan, S.T., Li, Z.L., He, Z.X., Qiu, J.X., Zhou, S.F. (2016) Molecular Mechanisms For Tumour Resistance To Chemotherapy. *Clin Exp Pharmacol Physiol* **43**: 723-737

Pankova, D., Chen, Y., Terajima, M., Schliekelman, M.J., Baird, B.N., Fahrenholtz, M., Sun, L., Gill, B.J., Vadakkan, T.J., Kim, M.P., Ahn, Y.H., Roybal, J.D, Liu, X., Parra Cuentas, E.R., Rodriguez, J., Wistuba, I.I., Creighton, C.J., Gibbons, D.L., Hicks, J.M., Dickinson, M.E., West, J.L., Grande-Allen, K.J., Hanash, S.M., Yamauchi, M., Kurie, J.M. (2016) Cancer-Associated Fibroblasts Induce A Collagen Cross-Link Switch In Tumour Stroma. *Mol Cancer Res* **14:** 287-295

Paradiso, A., Mangia, A., Chiriatti, A., Tommasi, S., Zito, A., Latorre, A., Scittulli, F., Lorusso, V. (2005) Biomarkers Predictive For Clinical Efficacy Of Taxol-Based Chemotherapy In Advanced Breast Cancer. *Annal Oncol* **16**: 14-19

Parise, C.A., Caggiano, V. (2014) Breast Cancer Survival Defined By ER/PR/Her2 Subtypes And A Surrogate Classification According To Tumour Grade And Immunohistochemical Biomarkers. *Journal Of Cancer Epidemiology* **2014:** 1-11 Park, S., Lee, S.K., Paik, H-J., Ryu, J.M., Kim, I., Bae, S.Y., Yu, J., Kim, S.W., Lee, J.E., Nam, S.J. (2017) Adjuvant Endocrine Therapy Alone In Patients With Node-Positive, Luminal A Type Breast Cancer. *Medicine (Baltimore)* **96:** E6777

Park, S.G., Jiang, Z., Mortenson, E.D., Deng, L., Radkevich-Brown, O., Yang, X., Sattar, H., Wang, Y., Brown, N.K., Greene, M., Liu, Y., Tang, J., Wang, S., Fu, Y-X. (2010) The Therapeutic Effect Of Anti-HER2/Neu Antibody Depends On Both Innate And Adaptive Immunity. *Cancer Cell* **18**: 160-170

Park, T., Yi, S-G., Kang, S-H., Lee, S.Y., Lee, Y-S., Simon, R. (2003) Evaluation Of Normalization Methods For Microarray Data. *BMC Informatics* **4**: 33

Parker, B.S., Rautela, J., Hertzog, P.J. (2016) Antitumour Actions Of Interferons: Implications For Cancer Therapy. *Nature Reviews Cancer* **16:** 131-144

Parker Thompson, K., Johnson, D.E., Stoute, D., Burow, M.E., Rhodes, L.V., Gray, M., Carriere, P., Tilghman, S.L., Mclachlan, J.A., Ochieng, J. (2013) In Vitro And In Vivo Evaluation Of Novel Anticancer Agents In Triple Negative Breast Cancer. *J Health Care Poor Undeserved*. **24**: 104-111

Parsonage, G., Filer, A.D., Haworth, O., Nash, G.B., Rainger G.E., Salmon, M., Buckley, C.D. (2005) A Stromal Address Code Defined By Fibroblasts. *Trend In Immunology* **26:** 150-156

Patel, J.N. (2015) Cancer Pharmacogenomics: Implications On Ethnic Diversity And Drug Response. *Pharmacogenet Genomics* **25**: 223-230

Pattyn, E., Van Ostade, X., Schauvliege, L., Verhee, A., Kalia, M., Vanderkerckhove, J., Tavernier, J. (1999) Dimerization Of The Interferon Type I Receptor Ifnar2-1 Is Sufficient For Induction Of Interferon Effector Genes But Not For Full Antiviral Activity. *J Biol Chem* **274**: 34838-34845

Peiris-Pagès, M., Sotgia, F., Lisanti, M. P. (2015) Chemotherapy Induces The Cancer-Associated Fibroblast Phenotype, Activating Paracrine Hedgehog-GLI Signaling In Breast Cancer Cells. *Oncotarget* **6**: 10728-10745.

Peng, Y., Croce, C.M. (2016) The Role Of Micrornas In Human Cancer. Signal Transduction And Targeted Therapy **1:** 1-9

Perou, C.M., Jeffrey, S.S., Van De Rijn, M., Rees, C.A., Eisen, M.B., Ross, D.T., Pergamenschikov, A., Williams, C.F., Zhu, S.X., Lee, J.C., Lashkari, D., Shalon, D., Brown, P.O., Botstein, D. (1999) Distinctive Gene Expression Patterns In Human Mammary Epithelial Cells And Breast Cancers. *Proc Nati Acad Sci USA* **96:** 9212-9217

Perou, C.M., Sorlie, T., Eisen, M.B., Van De Rijn, M., Jeffrey, S.S., Rees, C.A., Pollack, J.R., Ross, D.T., Johnsen, H., Akslen, L.A., Fluge, O., Pergamenschikov, A., Williams, C., Zhu, S.X., Lonning, P.E., Borresen-Dale, A., Brown, P.O., Botstein, D. (2000) Molecular Portraits Of Human Breast Tumours. *Nature* **406**: 747-752

Pettitt, S., Lord, C.J. (2018) PARP Inhibitors And Breast Cancer: Highlights And Hang-Ups. *Expert Review Of Precision Medicine And Drug Development* **3:** 83-94

Peshkin, B.N., Alabek, M.L., Isaacs, C. (2010) BRCA1/2 Mutations And Triple Negative Breast Cancers. *Breast Disease* **32:** 1-12

Pestka, S. (2007) The Interferons: 50 Years After Their Discovery, There Is Much More To Learn. *J Biol Chem*, **282**: 20047-51.

Pestka, S., Kotenko, S.V., Muthukumaran, G., Izotova, L.S., Cook, J.R., Garotta, G. (1997) The Interferon Gamma (IFN-Gamma) Receptor: A Paradigm For The Multichain Cytokine Receptor. *Cytokine Growth Factor Rev* **8:** 189-206

Pestka, S., Krause, C. D. & Walter, M. R. (2004) Interferons, Interferon-Like Cytokines, And Their Receptors. *Immunol Rev*, **202**: 8-32.

Pestka, S., Langer, J.A., Zoon, K.C., Samuel, C.E. (1987) Interferons And Their Actions. *Annu Rev Biochem* **56:** 727-777

Piccart-Gebhart, M.J., Procter, M., Leyland-Jones, B., Goldhirsch, A., Untch, M., Smith, I., Gianni, L., Baselga, J., Bell, R., Jackisch, C., Cameron, D., Dowsett, M., Barrios, C.H., Steger, G., Huang, C-S., Andersson, M., Inbar, M., Lichinitser, M., Láng, I., Nitz, U., Iwata, H., Thomssen, C., Lohrisch, C., Suter, T.M., Rüschoff, J., Sütő, T., Greatorex, V., Ward, C., Straehle, C., Mcfadden, E., Dolci, S., Gelber, R.D. (2005) Trastuzumab After Adjuvant Chemotherapy In HER2-Positive Breast Cancer. *N Engl J Med* **353**: 1659-1672

Piganis, R.A., De Weerd, N.A., Gould, J.A., Schindler, C.W., Mansell, A., Nicholson, S.E., Hertzog, P.J. (2011) Suppressor Of Cytokine Signaling (SOCS) 1 Inhibits Type I Interferon (IFN) Signaling Via The Interferon Alpha Receptor (IFNAR1)-Associated Tyrosine Kinase Tyk2. *J Biol Chem* **30**: 33811-33818 Pillai, R.S. (2005) Microrna Function: Multiple Mechanisms For A Tiny RNA? *RNA* **11:** 1753-1761

Pinder, S.E., Dugga, C., Ellis, I.O., Cuzick, J., Forbes, J.F., Bishop, H., Fentiman, I.S., George, W.D. (2010) A New Pathological System From Grading DCIS With Improved Prediction Of Local Recurrence: Results From The UKCCCR/ANZ DCIS Trial. *Br J Cancer* **103**: 94-100

Planche, A., Bacac, M., Provero, P., Fusco, C., Delorenzi, M., Stehle, J.C., Stamenkovic, I. (2011) Identification Of Prognostic Molecular Features In The Reactive Stroma Of Human Breast And Prostate Cancer. *Plos One* **6:**E18640

Platanias, L.C. (2005) Mechanisms Of Type-I And Type-II-Interferon-Mediated Signaling. *Nat Rev Immunol* **5:** 375-386

Platanias, L.C (2013) Interferons And Their Antitumour Properties. Journal Of Interferon Cytokine Research **33:** 143-144

Pomerantz, J.L., Baltimore, D. (1999) NF-Kb Activation By A Signaling Complex Containing TRAF2, TANK And TBK1, A Novel IKK-Related Kinase. *EMBO J* **18:** 6694-6704

Prat, A., Parker, J.S., Karginova, O., Fan, C., Livasy, C., Herschkowitz, J.I., He, X., Perou, C.M. (2010) Phenotypic And Molecular Characterization Of The Claudin-Low Intrinsic Subtype Of Breast Cancer. *Breast Cancer Res* **12**: R68

Pratt, A.J., Macrae, I.J. (2009) The RNA-Induced Silencing Complex: A Versatile Gene-Silencing Machine. *J Biol Chem* **284:** 17897-17901

Price, C., Chen, J. (2014) Micrornas In Cancer Biology And Therapy: Current Status And Perspectives. *Genes And Diseases* **1:** 53-63

Provost, P., Dishart, D., Doucet, J., Fredenwey, D., Samuelsson, B., Radmark, O. (2002) Ribonuclease Activity And RNA Binding Of Recombinant Human Dicer. *EMBO Journal* **21:** 5864-5874

Pusztai, L. (2005) Breast Cancer Molecular Subtypes Respond Differently To Preoperative Chemotherapy. *Clinical Cancer Research* **11**: 5678-5685.

Qiu, W., Hu, M., Sridhar, A., Opeskin, K., Fox, S., Shipitsin, M., Trivett, M., Thompson, E.R., Ramakrishna, M., Gorringe, K.L., Polyak, K., Haviv, I., Campbell, I.G. (2008) No Evidence Of Clonal Somatic Genetic Alterations In

Cancer-Associated Fibroblast From Human Breast And Ovarian Carcinomas. *Nat Genet* **40:** 650-655

Qu, H., Fang, L., Duan, L., Long, X. (2014) Expression Of ABCG2 And P-Glycoprotein In Residual Breast Cancer Tissue After Chemotherapy And Their Correlation With Epithelial-Mesenchymal Transition. *Zhonghua Bing Li Xue Za Zhi* **43**: 236-240

Quinn, B.A., Dash, R., Azab, B., Sarker, S., Das, S.K., Kumar, S., Oyesanya, R.A., Dasgupta, S., Dent, P., Grant, S., Rahmani, M., Curiel, D.T., Dmitriev, I., Hedvat, M., Wei, J., Stebbins, J.L., Reed, J.C., Pellecchia, M., Sarker D., Fisher, P.B. (2011) Targetting Mcl-1 For The Therapy Of Cancer. *Expert Opin Investig Drugs* **10:** 1397-1411

Qureshi, O.S., Bon, H., Twomey, B., Holdsworth, G., Ford, K., Bergin, M., Huang, L., Muzylak, M., Healy, L.J., Hurdowar, V., Johnson, T.S. (2017) An Immunofluorescence Assay For Extracellular Matrix Components Highlights The Role Of Epithelial Cells In Producing A Stable Fibrillar Extracellular Matrix. *Biol Open* **6**: 1423-1433

Rajkumar, V.S., Shiwen, X., Bostrom, M., Leoni, P., Muddle, J., Ivarsson, M., Gerdin, B., Denton, C.P., Bou-Gharios, G., Black, C.M., Abraham, D, J. (2006) Platelet Derived Growth Factor-B Receptor Is Essential For Fibroblast And Pericyte Recruitment During Cutaneous Wound Healing. *American Journal Of Pathology* **169:** 2254-2265

Rakha, E.A., El-Sayed, M.E., Lee, A.H., Elston, C.W., Grainge, M.J., Hodi, Z., Blamey, R.W., Ellis, I.O. (2008) Prognositic Significance Of Nottingham Histologic Grade In Invasive Breast Carcinoma. *Journal Of Clinical Oncology* **26:** 3153-3158

Rakha, E.A., Reis-Filho, J.S., Ellis, I.O. (2008) Basal-Like Breast Cancer: A Critical Review. *Journal Of Clinical Oncology* **26:** 2568-2581

Rakha, E.A., Reis-Filho, J.S., Baehner, F., Dabbs, D.J., Decker, T., Eusebi, V., Fox, S.B., Ichihara, S., Jacquemier, J., Lakhani, S.R., Palacios, J., Richardson, A.L., Schnitt, S.J., Schmitt, F.C., Tan, P-H., Tse, G.M., Badve, S., Ellis, I.O. (2010) Breast Cancer Prognostic Classification In The Molecular Era: The Role Of Histological Grade. *Breast Cancer Res* **12**: 207 Rampersad, S.N. (2012) Multiple Applications Of Alamar Blue As An Indicator Of Metabolic Function And Cellular Health In Cell Viability Bioassays. *Sensors (Basel)* **12:** 12347-12360

Randall, R. E. & Goodbourn, S. (2008) Interferons And Viruses: An Interplay Between Induction, Signaling, Antiviral Responses And Virus Countermeasures. *J Gen Virol*, **89:** 1-47.

Ravikanth, M., Soujanya, P., Manjunath, K., Saraswathi, T.R., Ramachandran, C.R. (2011) Heterogenecity Of Fibroblasts. *J Oral Maxillofac Pathol* **15:** 247-250

Rebouillat, D., Hovanessian, A.G. (1999) The Human 2',5'-Oligoadenylate Synthetase Family: Interferon-Induced Proteins With Unique Enzymatic Properties. *J Interferon Cytokine Res* **19:** 295-308

Reddy, P.K., Gold, D.V., Cardillo, T.M., Goldenberg, D.M., Li, H., Burton, J.D. (2003) Interferon-Gamma Upregulates MUC1 Expression In Haematopoietic And Epithelial Cancer Cell Lines, An Effect Associated With MUC1 Mrna Induction. *Eur J Cancer* **39**: 397-404

Reginato, M.J., Mills, K.R., Paulus, J.K., Lynch, D.K., Sgroi, D.C., Debnath, J., Muthuswamy, .K., Brugge, .S. (2003) Integrins And EGFR Coordinately Regulate The Pro-Apoptotic Protein Bim To Prevent Anoikis. *Nat Cell Biol* **5**: 733-740

Reinert, K.E. (1983) Anthracycline-Binding Induced DNA Stiffening, Bending And Elongation; Sterochemical Implications From Viscometric Investigations. *Nucleic Acids Res* **11:** 3411-3430

Riaz, M., Van Jaarsveld, M.T.M., Hollestelle, A., Prage-Van Der Smissen, W.J.C.,Heine, A.A.J., Boersma, A.W.M., Liu, J., Helmijr, J., Ozturk, B., Smid, M., Wiemer, E.A., Foekens, J.A., Martens, J.W.M. (2013) Mirna Expression Profiling Of 51 Human Breast Cancer Cell Lines Reveals Subytpe And Driver Mutation Specific Mirnas. *Breast Cancer Res* **15:** R33

Ricardo, S., Gerhard, R., Cameselle-Teijerio, J.F., Schmitt, F., Paredes, J. (2012) Claudin Expression In Breast Cancer: High Or Low, What To Expect? *Histopathol* **27**: 1283-95

Riss, T.L., Moravec, R.A., Niles, A.L., Duellman, S., Benink, H.A., Worzella, T.J., Minor, L. (2013) Assay Guidance Manual. *Eli Lilly & Company And The National Center For Advancing Translational Sciences.* Robson, M., Im, S-A., Senkus, E., Xu, B., Domchek, S.M., Masuda, N., Delaloge, S., Li, W., Tung, N., Armstrong, A., Wu, W., Goessl, C., Runswick, S., Conte, P. (2017) Olaparib For Metastatic Breast Cancer In Patients With A Germline BRCA Mutation. *N Engl J Med* **377**: 523-533

Rosai J. Rosai And Ackerman's Surgical Pathology. Tenth Ed. Elsevier; Lyon, France: 2011.

Ross, A.A., Cooper, B.W., Lazarus, H.W., Mackay, W., Moss, T.J., Ciobanu, N., Tallman, T.S., Kennedy, M.J., Davidson, N.E., Sweet, D. (1993) Detection And Viability Of Tumour Cells In Peripheral Blood Stem Cell Collections From Breast Cancer Patients Using Immunocytochemical And Clonogenic Assay Tecnhiques. *Blood* **82**: 2605-2610

Rothschild, E., Banerjee, D. (2015) Subverting Subversion: A Review On The Breast Cancer Microenvironment And Therapeutic Opportunities. *Breast Cancer: Basic And Clinical Research* **9:** 7-15

Rouzier, R., Perou, C. M., Symmans, W. F., Ibrahim, N.,

Cristofanilli, M., Anderson, K., Hess, K. R., Stec, J., Ayers, M.,

Wagner, P., Morandi, P., Fan, C., Rabiul, I., Ross, J. S.,

Hortobagyi, G. N., Pusztai, L. (2005) Breast Cancer Molecular Subtypes

Respond Differently To Preoperative Chemotherapy. Clin

Cancer Res 11: 5678-85.

Ruby, J.G., Jan, C.H., Bartel, D.P. (2007) Intronic Microrna Precursors That Bypass Drosha Processing. *Nature* **448:** 83-86

Sabatier, R., Finetti, P., Guile, A., Adelaide, J., Chaffanet, M., Viens, P., Birnbaum, D., Bertucci, F. (2014) Claudin-Low Breast Cancers: Clinical,

Pathological, Molecular And Prognostic Characterization. *Mol Cancer* 13: 228
Sachdev, J.C., Jahanzeb, M. (2016) Use Of Cytotoxic Chemotherapy In
Metastatic Breast Cancer: Putting Taxanes Into Perspective. *Clinical Breast Cancer* 16: 73-81

Sadler, A.J., Williams, B.R.G. (2008) Interferon-Inducible Antiviral Effectors. *Nature Reviews Immunology* **8:** 559-568

Sadlonova, A., Bowe, D.B., Novak, Z., Mukherjee, S., Duncan, V.E., Page, G.P., Frost, A.R. (2009) Identification Of Molecular Distinctions Between Normal Breast-Associated Fibroblasts And Breast Cancer-Associated Fibroblasts. *Cancer Microenviron* **2:** 9-21 Sajid, M.T., Ahmed, M., Azhar, M., Mustafa, Q.U., Shukr, I., Ahmed, M., Kamal, Z. (2014) Age-Related Frequency Of Triple Negative Breast Cancer In Women. *Journal Of The College Of Physicians And Surgeons Pakistan* **24:**400-403

Samoszuk, M., Tan, J., Chorn, G. (2005) Clonogenic Growth Of Breast Cancer Cells Co-Cultured In Direct Contact With Serum-Activated Fibroblasts. *Breast Cancer Research* **7**: 274-283

Samuel, C.E. (2001) Antiviral Actions Of Interferons. *Clin Microbiol Rev* **14:** 778-809

Sangfelt, O., Erickson, S., Grander, D. (2000) Mechanisms Of Interferon-Induced Cell Cycle Arrest. *Front Biosci*, **5:** D479-87.

Santos, J.C., Lima, N.S., Sarian, L.O., Matheu, A., Ribeiro, M.L., Derchain, S.F.M. (2018) Exosome-Mediated Breast Cancer Chemoresistance Via Mir-155 Transfer. *Scientific Reports* **8:** 1-11

Sarasin-Filipowicz, M., Wang, X., Yan, M., Duong, F.H., Poli, V., Hilton, D.J., Zhang, D.E., Heim, M.H. (2009) Alpha Interferon Induces Long-Lasting Refractoriness Of JAK-STAT Signaling In The Mouse Liver Through Induction Of USP18/UBP43. *Molecular Cell Biology* **29**: 4841-4851.

Sarrazin, D., Le, M.G., Arriagada, R., Contesso, G., Fontaine, F., Spielmann, M., Rochard, F., Le Chevalier, T., Lacour, J. (1989) Ten-Year Results Of A Randomized Trial Comparing A Conservative Treatment To Mastectomy In Early Breast Cancer. *Radiotherapy And Oncology* **14**: 177-184

Sau, A., Pellizzari Tregno, F., Valentino, F., Federici, G., Caccuri, A.M. (2010) Glutathione Transferase And Development Of New Principles To Overcome Drug Resistance. *Arch Biochem Biophys* **500**: 116-122

Saura, C., Roda, D., Roselló, S., Oliveria, M., Macarulla, T., Pérez-Fidalgo, J.A., Morales-Barrera, R., Sanchis-García, J.M., Musib, L., Budha, N., Zhu, J., Nannini, M., Chan, W.Y., Sanabria Bohórquez, S.M., Meng, R.D., Lin, K., Yan, Y., Patel, P., Baselga, J., Tabernero, J., Cervantes, A. (2017) A First-In Human Phase I Study Of The ATP-Competitive AKT Inhibitor Ipataserib Demonstrates Robust And Safe Targeting Of AKT In Patients With Solid Tumors. *Cancer Discov* **7**: 102-113

Schoepp, M., Ströse, A.J., Haier, J. (2017) Dysregulation Of Mirna Expression In Cancer Associated Fibroblasts (Cafs) And Its Consequences On The Tumour Microenvironment. Cancers (Basel) 9: 54

Schiavoni, G., Mattei, F., Gabriele, L. (2013) Type I Interferons As Stimulators Of DC-Mediated Cross-Priming: Impact On Anti-Tumour Response. *Frontiers Of Immunology* **4**: 483

Schneider, W. M., Chevillotte, M. D. & Rice, C. M. (2014) Interferon-Stimulated Genes: A Complex Web Of Host Defenses. *Annu Rev Immunol*, **32:** 513-45.

Schnitt, S.J. (2010) Classification And Prognosis Of Invasive Breast Cancer: From Morphology To Molecular Taxonomy. *Modern Pathology* **23:** 60-64

Scully, R., Anderson, S.F., Chao, D.M., Wei, W., Ye, L., Young, R.A., Livingston, D.M., Parvin, J.D. (1997) BRCA1 Is A Component Of The RNA Polymeraseiiholoenzyme. *PNAS* **94:** 5605-5610

Senthebane, D.A., Rowe, A., Thomford, N.E., Shipanga, H., Munro, D., Al Mazeedi, M.A.M., Almazyadi, H.A.M., Kallmeyer, K., Dandara, C., Pepper, M.S., Parker, M.I., Dzobo, K. (2017) The Role Tumour Microenvironment In Chemoresistance: To Survive, Keep Your Enemies Closer. *Int J Mol Sci* **18**: 1586

Sgroi, D.C., Teng, S., Robinson, G., Levangie, R., Hudson, J.R Jr., Elkahloun, A.G. (1999) In Vivo Gene Expression Profile Analysis Of Human Breast Cancer Progression. *Cancer Research* **59**: 5656-5661

Shekhar, M.P., Santner, S., Carolin, K.A., Tait, L. (2007) Direct Involvemtn Of Breast Tumour Fibroblasts In The Modulation Of Tamoxifen Sensitivity. *Am J Pathol* **170:** 1546-1560

Shi, Y-J., Tsang, J.Y.S., Ni, Y-B., Tse, G.M. (2017) Intratumoral Heterogeneity In Breast Cancer: A Comparison Of Primary And Metastatic Breast Cancers. *Oncologist* **22:** 487-490

Shivdasani, R.A. (2006) Micrornas: Regulators Of Gene Expression And Cell Differentiation. *Blood* **108**: 3646-3653

Shiga, K., Hara, M., Nagasaki, T., Sato, T., Takahashi, H. & Takeyama, H. (2015) Cancer-Associated Fibroblasts: Their Characteristics And Their Roles In Tumor Growth. *Cancers*, **7**: 2443-2458.

Shomron, N., Levy, C. (2009) Microrna-Biogenesis And Pre-Mrna Splicing Cross-Talk. *J Biomed Biotechnol* **2009:** 594678 Silvennoinen, O., Ihle, J.N., Schlessinger, J., Levy, D.E. (1993) Interferon-Induced Nuclear Signaling By Jak Protein Tyrosine Kinases. *Nature* **366:** 583-585

Silver, D.P., Livingston, D.M. (2012) Mechanisms Of BRCA1 Tumour Suppression. *Cancer Discovery* **2:** 679-684

Simpson, A., Caballero, O. (2014) Monoclonal Antibodies For The Therapy Of Cancer. *BMC Proceedings*, **8:** O6-O6.

Singer, C.F., Kronsteiner, N., Marton, E., Kubista, M., Cullen, K.J., Hirtenlehner, K., Seifert, M., Kubista, E. (2002) MMP-2 And MMP-9 Expression In Breast Cancer-Derived Human Fibroblasts Is Differentially Regulated By Stromal-Epithelial Interactions. *Breast Cancer Res Treat* **72**: 69-77

Slàdek, N.E., Kollander, R., Sreerama, L., Kiang, D.T. (2002) Cellular Levels Of Aldehyde Dehydrogenase (ALDH1A1 And ALDH3A1) As Predictors Of Therapeutic Response To Cyclophosphamide-Based Chemotherapy Of Breast Cancer: A Retorspective Study. *Cancer Chemotherapy And Pharmacology* **49:** 309-321

Slamon, D.J., Leyland-Jones, B., Shak, S., Fuchs, H., Paton, V., Bajamonde, A., Fleming, T., Eiermann, W., Wolter, J., Pegram, M., Baselga, J., Norton, L. (2001) Use Of Chemotherapy Plus A Monoclonal Antibody Against HER2 For Metastatic Breast Cancer That Overexpresses HER2. *N Engl J Med* **344:** 783-792

Slaney, C.Y., Möller, A., Hertzog, P.J., Parker, B.S. (2013) The Role Of Type I Interferons In Immunoregulation In Breast Cancer Metastasis To The Bone. *Oncoimmunology* **2**: E22339

Spaeth, E.L., Dembinski, J.L., Sasser, K.A., Watson, K., Klopp, A., Hall, B., Andreeff, M., Marini, F. (2009) Mesenchymal Stem Cell Transition To Tumour-Associated Fibroblasts Contributes To Fibrovascular Network Expansion And Tumour Progression. *Plos One* **4**: E4992

Sonnenberg, M., Van Der Kuip, H., Haubeiss, Fritz, P., Schroth, W., Friedel, G., Simon, W., Mürdter, T.E., Aulitzky, W.E. (2008) Highly Variable Response To Cytotoxic Chemotherapy In Carcinoma-Associated Fibroblasts (Cafs) From Lung And Breast. *BMC Cancer* **8:** 364 Soon, P.S.H., Kim, E., Pon, C.K., Gill, A.J., Moore, K., Spillane, A.J., Benn, D.E., Baxter, R.C. (2013) Breast Cancer-Associated Fibroblasts Induce Epithelial-Mesenchymal Transition In Breast Cancer Cells. *Endocrine-Related Cancer* **20:** 1-12

Sørlie, T., Perou, C.M., Tibshirani, R., Aas, T., Geisler, S., Johnsen, H., Hastie, T., Elsen, M.B., Van De Rijn, M., Jeffrey, S.S., Thorsen, T., Quist, H., Matese, J.C., Brown, P.O., Botstein, D., Lønning, P.E., Børresen-Dale, A.L. (2001) Gene Expression Patterns Of Breast Carcinomas Distinguish Tumour Subclasses With Clinical Implications. *Proc Nati Acad Sci USA* **98**: 10869-10874

Stadler, Z.K., Come, S.E. (2009) Reviw Of Gene-Expression Profiling And Its Clinical Use In Breast Cancer. *Crit Review Oncol Hematol* **69:** 1-11

Stagg, J., Allard, B. (2013) Immunotherapeutic Approaches In Triple-Negative Breast Cancer: Latest Research And Clinical Prospects. *Therapeutic Advances In Medical Oncology* **5:** 169-181

Stover, D. G., Gil Del Alcazar, C. R., Brock, J., Guo, H.,

Overmoyer, B., Balko, J., Xu, Q., Bardia, A., Tolaney, S. M.,

Gelman, R., Lloyd, M., Wang, Y., Xu, Y., Michor, F., Wang, V.,

Winer, E. P., Polyak, K., Lin, N. U. (2018) Phase II

Study Of Ruxolitinib, A Selective JAK1/2 Inhibitor, In Patients With Metastatic Triple-Negative Breast Cancer. *NPJ Breast Cancer*, **4:**10.

Strome, S. E., Sausville, E. A., Mann, D. (2007) A Mechanistic Perspective Of Monoclonal Antibodies In Cancer Therapy Beyond Target-Related Effects. *Oncologist* **12**: 1084-95.

Subramaniam, P.S., Johnson, H.M. (1997) A Role For The Cyclin-Dependent Kiase Inhibitor P21 In The G1 Cell Cycle Arrest Mediated By The Type I Interferons. *J Interferon Cytokine Res* **17**: 11-15

Sun, Y., Campisi, J., Higano, C., Beer, T. M., Porter, P., Coleman, I., True, L., Nelson, P. S. (2012) Treatment-Induced Damage To The Tumor Microenvironment Promotes Prostate Cancer Therapy Resistance Through WNT16B. *Nature Medicine* **18:** 1359-1368.

Swann, J.B. (2007) Type I IFN Contributes To NK Cell Homeostasis, Activation And Antitumour Function. *Journal Of Immunology* **178:** 7540-7549 Taherian, A., Li, X., Liu, Y., Haas, T.A. (2011) Differences In Integrin Expression And Signaling Within Human Breast Cancer Cells. *BMC Cancer* **13**: 293

Tanaka, K., Miyata, H., Sugimura, K., Fukuda, S., Kanemura, T., Yamashita, K., Miyazaki, Y., Takahashi, T., Kurokawa, Y., Yamasaki, M., Wada, H., Nakajima, K., Takiguchi, S., Mori, M., Doki, Y. (2015) Mir-27 Is Associated With Chemoresistance In Esophageal Cancer Through Transformation Of Normal Fibroblasts To Cancer-Associated

Fibroblasts. Carcinogenesis 36: 894-903.

Tang, W., Zhu, J., Su, S., Wu, W., Liu, Q., Su, F., Yu, F. (2012) Mir-27 As A Prognostic Marker For Breast Cancer Progression And Patient Survival. *PLOS ONE* **7:** E51702.

Tarin, D., Croft, C.B. (1970) Ultrastructural Studies Of Wound Healing In Mouse Skin. II. Dermo-Epidermal Interrelationships. *J Anat* **106**: 79-91

Thomas, S. J., Snowden, J. A., Zeidler, M. P., Danson, S. J. (2015) The Role Of JAK/STAT Signaling In The Pathogenesis, Prognosis And Treatment Of Solid Tumours. *Br J Cancer* **113**: 365-71.

Thompson, A.M., Moulder-Thompson, S.L. (2012) Neoadjuvant Treatment Of Breast Cancer. *Annals Of Oncology* **23:** 231-236

Thyrell, L. (2002) Mechanisms Of Interferon-Alpha Induced Apoptosis In Malignant Cells. Oncogene **21**: 1251-1262

Tokunaga, E., Oki, E., Egashira, A., Sadanaga, N., Morita, M., Kakeji, Y., Maehara, Y. (2008) Deregulation Of The Akt Pathway In Human Cancer. *Curr Cancer Drug Targets*, **8:** 27-36.

Tomasek, J.J., Gabbiani, G., Hinz, B., Chaponnier, C., Brown, R.A. (2002) Myofibroblasts And Mechano-Regulation Of Connective Tissue Remodeling. *Nature Reviews Molecular Cell Biology* **3:** 349-363

Trimble, E.L., Ungerleider, R.S., Abrams, J.A., Kaplan, R.S., Feigal, E.G., Smith, M.A., Carter, C.L., Friedman, M.A. (1993) Neoadjuvant Therapy In Cancer Treatment. *Cancer* **72**: 3515-3524

Tsai, H-P., Huang, S-F., Li, C-F., Chien, H-T., Chen, S-C. (2018) Differential Microrna Expression In Breast Cancer With Different Onset Age. *Plos One* **13**: E0191195

Tsoutsou, P.G., Sozzi, W. J., Ozsahin, M., Delaloye, J.F., Bourhis, J.

(2013) Radiotherapy Options After Breast-Conserving Surgery: How Can Selection Of Patients Be Refined. *Journal Of Clinical Oncology* **31:** 4570-4571

Tsuno, T., Mejido, J., Zhao, T., Schmeisser, H., Morrow, A. & Zoon, K. C. (2009) Irf9 Is A Key Factor For Eliciting The Antiproliferative Activity Of Ifn-Alpha. *J Immunother*, **32**: 803-16.

Tutt, A., Robson, M., Garber, J.E., Domchek, S., Audeh, M.W., Weitzel, J.N., Friedlander, M., Arun, B., Loman, N., Schmutzler, R.K., Wardley, A., Mitchell, G., Earl, H., Wickens, M., Carmichael, J. (2010) Oral Poly(ADP-Ribose) Polymerase Inhibitor Olaparib In Patients With BRCA1 Or BRCA2 Mutations And Advanced Breast Cancer: A Proof Of Concept Trial. *Lancet* **376**: 235-244

Tutt, A., Robson, M., Garber, J.E., Domchek, S., Audeh, M.W., Weitzel, J.N., Friedlander, M., Carmichael, J. (2009) Phase II Of The Oral PARP Inhibitor Olaparib In BRCA-Deficient Advanced Breast Cancer. *Journal Of Clinical Oncology* **27**: CRA501-CRA501

Uddin, S., Majchrzak, B., Woodson, J., Arunkumar, P., Alsayed, Y., Pine, R., Young, P.R., Fish, E.N., Platanias, L.C. (1999) Activation Of The P38 Mitogen-Activated Protein Kinase By Type I Interferons. *J Biol Chem* **274**: 30127-30131

Uehara, N., Kanematsu, S., Miki, H., Yoshizawa, K., Tsubura, A. (2012) Requirement Of P38 MAPK For A Cell-Death Pathway Triggered By Vorinostat In MDA-MB-231 Human Breast Cancer Cells. *Cancer Letters* **315**: 112-121

Uematsu, S., Akira, S. (2007) Toll-Like Receptors And Type I Interferons. *J Biol Chem* **282**: 15319-15323

Valencia-Sanchez, M.A., Liu, J., Hannon, G.J., Parker, R. (2006) Control Of Transaltion And Mrna Degradation By Mirnas And Sirnas. *Genes Development* **20:** 515-524

Vallejos, C.S., Gómez, H.L., Cruz, W.R., Pinto, J.A., Dyer, R.R., Velarde, R., Suazo, J.F., Neciosup, S.P., León, M., De La Cruz, M.A., Vigil, C.E. (2010) Breast Cancer Classification According To Immunohistochemistry Markers: Subtypes And Association With Clinicopathologic Variables In A Peruvian Hospital Database. *Clin Breast Cancer* **10**: 292-300

Van Der Rhee, M.H., Van Der Meer, A.J., Van Nuenen, A.C., De Bruijne, J., Ottosen, S., Janssen, H.L., Kootstra, N.A., Reesink, H.W. (2016) Miravirsen

Dosing In Chronic Hepatitis C Patients Resulted In Decreased Microrna-122 Levels Without Affecting Other Micrornas In Plasma. *Aliment Pharmacol Ther* **43:** 102-113

Vangangelt, K.M.H., Van Pelt, G.W., Engels, C.C., Putter, H., Liefers, G.J., Smit, V.T.H.B.M., Tollenaar, R.A.E.M., Kuppen, P.J.K., Mesker, W.E. (2018) Prognostic Value Of Tumour-Stroma Ration Combined With The Immune Status Of Tumours In Invasive Breast Carcinoma. *Breast Cancer Res Treat* **168**: 601-612

Van Tonder, A., Joubert, A.M., Cromarty, D.A. (2015) Limitations Of 3-(4,5-Dimethylthiazol-2-YI)-2,5-Diphenyl-2H-Tetrazolium Bromide (MTT) Assay When Compared To Three Commonly Used Cell Enumeration Assays. *BMC Res Notes* **8**: 47

Van't Veer, L.J., Dai, H., Van Der Vijver, M.J., He, Y.D., Hart, A.A., Mao, M., Peterse, H.L., Van Der Krooy, K., Marton, M.J., Witteveen, A.T., Schreiber, G.J., Kerkhoven, R.M., Roberts, C., Linsley, P.S., Bernards, R., Friend, S.H. (2002) Gene Expression Profiling Predicts Clinical Outcome Of Breast Cancer. *Nature* **31**: 530-536

Vegran, F., Boidot, R., Oudin, C., Riedinger, J-M., Bonnetain, F., Lizard-Nacol, S. (2006) Overexpression Of Caspase-3s Splice Variant In Locally Advanced Breast Carcinoma Is Associated With Poor Response To Neoadjuvant Chemotherapy. *Clin Cancer Research* **12:** 5794-5800

Veldwijk, M.R., Neumaier, C., Gerhardt, A., Giordano, F.A., Sutterlin, M., Herskind, C., Wenz, F. (2015) Comparison Of The Proliferative And Clonogenic Growth Capacity Of Would Fluid From Breast Cancer Patients Treated With And Without Intraoperative Radiotherapy. *Translational Cancer Research* **4**: 1-5

Verghese, E.T., Drury, R., Green, C.A., Holliday, D.A., Lu, X., Nash, C., Speirs, V., Thorne, J.L., Thygsen, H.H., Zougman, A., Hull M.A., Hanby, A.M., Hughes, T.A. (2013) Mir-26b Is Down-Regulated In Carcinoma-Associated Fibroblasts From ER-Positive Breast Cancers Leading To Enhanced Cell Migration And Invasion. *Journal Of Pathology* **231:** 388-399

Verghese, E.T., Shenoy, H., Cookson, V.J., Green, C.A., Howarth, J., Partanen, R.H., Pollock, S., Waterworth, A., Speirs, V., Hughes, T.A., Hanby, A.M. (2011) Epithelial-Mesenchymal Interactions In Breast Cancer: Evidence For A Role Of Nuclear Localized B-Catenin In Carcinoma-Associated Fibroblasts. *Histopathology* **59:** 609-618

Vinay K, Abul KA, Jon CA, Nelson F. Robbins And Cotran Pathologic Basis Of Disease. Eight Ed. Elsevier; Lyon, France: 2010.

Vitolo, M. I., Weiss, M. B., Szmacinski, M., Tahir, K., Waldman, T., Park, B. H., Martin, S. S., Weber, D. J., Bachman, K. E. (2009) Deletion Of PTEN Promotes Tumorigenic Signaling, Resistance To Anoikis, And Altered Response To Chemotherapeutic Agents In Human Mammary Epithelial Cells. *Cancer Research*, **69:** 8275.

Volinia, S., Calin, G.A., Liu, C.G., Ambs, S., Cimmino, A., Petrocca, F., Visone, R., Iorio, M., Roldo, C., Ferracin, M., Prueitt, R.L., Yanaihara, N., Lanza, G., Scarpa, A., Vecchione, A., Negrine, M., Harris, C.C., Croce, C.M. (2006) A Microrna Expression Signature Of Human Solid Tumours Defines Cancer Gene Targets. *PNAS* **14**: 2257-2261

Von Minkwitz, G., Proctor, M., De Azambuja, E., Zardavas, D., Benyunes, M., Viale, G., Suter, T., Arahmani, A., Rouchet, N., Clark, E., Knott, A., Lang, I., Levy, C., Yardley, D.A., Bines, J., Gelber, R.D., Piccart, M., Baselga, J. (2017) Adjuvant Pertuzumab And Trastuzumab In Early Her-2 Postive Breast Cancer. *The New England Journal Of Medicine* **377:** 122-131

Vu, T., Claret, F.X. (2012) Trastuzumab: Updated Mechanisms Of Action And Resistance In Breast Cancer. *Front Oncol* **2**: 62

Wadler, S., Fuks, J.Z., Wiernik P.H. (1986) Phase I And II Agents In Cancer Therapy: I. Anthracyclines And Related Compounds. *J Clin Pharmacol* **26:** 491-509

Wahba, H.A., El-Hadaad, H.A. (2015) Current Approaches In Treatment Of Triple-Negative Breast Cancer. *Cancer Biol Med* **12**: 106-116

Walter, M., Liang, S., Ghosh, S., Hornsby, P.J., Li, R. (2009) Interleukin 6 Secreted From Adipose Stromal Cells Promotes Migration And Invasion Of Breast Cancer Cells. *Oncogene* **28**: 2745-2755

Wan, S., Pestka, S., Jubin, R.G., Lyu, T.G., Tsai, Y.C., Liu, L.F. (2012) Chemotherapeutics And Radiation Stimulate MHC Class I Expression Through Elevated Interferon-Beta Signaling In Breast Cancer Cells. *Plos One* **7**: E32542

Wang, B., Xi, C., Liu, M., Sun, H., Liu, S., Song, L., Kang, H. (2018) Breast Fibroblasts In Both Cancer And Normal Tissues Induce Phenotypic Transformation Of Breast Cancer Stem Cells: A Preliminary Study. *Peer*, **6**: E4805.

Wang, Y., Cortez, D., Yazdi, P., Neff, N., Elledge, S.J., Qin, J. (2000) BASC, A Super Complex Of BRCA1-Associated Proteins Involved In The Recognition And Repair Of Aberrant DNA Structures. *Genes Dev* **14**: 927-939

Wang, Z., Gerstein, M., Snyder, M. (2009) RNA-Seq: A Revolutionary Tool For Transcriptomics. *Nature Reviews Genetics* **10**: 57.

Weichselbaum, R.R., Ishwaran, H., Yoon, T., Nuyten, D.S.A., Baker, S.W., Khodarev, N., Su, A.W., Shaikh, A.Y., Roach, P., Kreike, B., Roizman, B., Bergh, J., Pawitan, Y., Van De Vijver, M.J., Minn, A.J. (2008) An Interferon-Related Gene Signature For DNA Damage Resistance Is A Predictive Marker For Chemotherapy And Radiation For Breast Cancer. *PNAS* **105**: 18490-18495

Weigel, K. J., Jakimenko, A., Conti, B. A., Chapman, S. E., Kaliney, W. J., Leevy, W. M., Champion, M. M., Schafer, Z. T. (2014) CAF-Secreted Igfbps Regulate Breast Cancer Cell Anoikis. *Mol Cancer Res* **12**: 855-66.

Weigelt, B., Peterse, J.L., Van't Veer, L.J. (2005) Breast Cancer Metastasis: Markers And Models. *Nature* **5**: 591-602

Weldon, C.B., Burow, M.E., Rolfe, K.W., Clayton, J.L., Jaffe, B.M., Beckman, B.S. (2001) NF-Kappa B-Mediated Chemoresistance In Breast Cancer Cells. *Surgery* **130:** 143-150

Wellings, S.R., Jensen, H.M. (1973) On The Origin And Progression Of Ductal Carcinoma In The Human Breast. *J Nati Cancer Inst* **50**: 1111-1118

Wen, Z., Zhong, Z., Darnell, J.E Jr. (1995) Maximal Activation Fo Transcription By Stat1 And Stat3 Requires Both Tyrosine And Serine Phosphorylation. *Cell* **82:** 241-250

Wind, N.S., Holen, I. (2011) Multidrug Resistance In Breast Cancer: From In Vitro Models To Clinical Studies. *Int J Breast Cancer* **2011**: 967419

Winslow, S., Leandersson, K., Edsjo, A., Larsson, C. (2015) Prognostic Stromal Gene Signatures In Breast Cancer. *Breast Cancer Res* **17**: 23.

Winslow, S., Lindquist, K. E., Edsjo, A., Larsson, C. (2016) The Expression Pattern Of Matrix-Producing Tumor Stroma Is Of Prognostic Importance In Breast Cancer. *BMC Cancer* **16**: 841.

Wiseman, B.S., Werb, Z. (2002) Stromal Effects On Mammary Gland Development And Breast Cancer. *Science* **296**: 1046-1049 Woolf, D.K., Padhani, A.R., Makris, A. (2015) Assessing Response To Treatment Of Bone Metastases From Breast Cancer: What Should Be The Standard Of Care? *Annals Of Oncology* **26**: 1048-1057

Xiang, Y. J., Fu, Q. Y., Ma, Z. B., Gao, D. Z., Zhang, Q., Li, Y. Y., Li, L., Liu, L., Ye, C. M., Yu, Z. G., Guo, M. M. (2015) Screening For Candidate Genes Related To Breast Cancer With Cdna Microarray Analysis. *Chronic Dis Transl Med* **1:** 65-72.

Xiong, J., Fan, F., Zhang, Y., Chen, W., Mao, W. (2016) Epieubicin Inhibits Proliferation Of Breast Cancer Cells Through Upregulating P21^{cip1} Expression. *Int J Clin Exp Med* **9**: 22764-22772

Xu, Y.H., Lu, S. (2014) A Meta-Analysis Of STAT3 And Phosphor-STAT3 Expression And Survival Of Patients With Non-Small-Cell Lung Cancer. *Eur J Surg Oncol* **40**: 311-317

Xu, J.M., Song, S.T., Tang, Z.M., Jiang, Z.F., Liu, X.Q., Zhou, L., Zhang, J., Liu, X.W. (1999) Predictive Chemotherapy Of Advanced Breast Cancer Directed By MTT Assay In Vitro. *Breast Cancer Research Treatment* **53**: 77-85

Yan, L.X., Wu, Q.N., Zhang, Y., Li, Y.Y., Liao, D.Z., Hou, J.H., Fu, J., Zeng, M.S., Yun, J.P., Wu, Q.L., Zeng, Y.X., Shao, J.Y. (2011) Knockdown Of Mir-21 In Human Breast Cancer Cell Lines Inhibits Proliferation, In Vitro Migration And In Vivo Tumour Growth. *Breast Cancer Res* **13:** R2

Yang, S., Netterwald, J., Wang, W., Zhu, H. (2005) Characterization Of The Elements And Proteins Responsible For Interferon-Stimulated Gene Idncution By Cytomegalovirus. *J Virol* **79:** 5027-5034

Yershal, O., Bartuca, S. (2014) Biological Subtypes Of Breast Cancer. *World Journal Of Clinical Oncology* **5:** 412-424

Yeung, M. L., Jeang, K.T. (2011) Micrornas And Cancer Therapeutics. *Pharmaceutical Research* **28:** 3043-3049.

Yi, R., Qin, Y., Macara, I.G., Cullen, B.R. (2003) Exportin-5 Mediates The Nuclear Export Of Pre-Micrornas And Short Hairpin Rnas. *Genes Development* **17:** 3011-3016

Yi, C.A., Tam, C.S., Verstovsek, S. (2015) Efficacy And Safety Of Ruxolitinib In The Treatment Of Patients With Myelofibrosis. *Future Oncol* **11**: 719-733

Yoshimura, A., Naka, T., Kubo, M. (2007) SOCS Proteins, Cytokine

Signaling And Immune Regulation. Nature Reviews Immunology 7: 454-465

Young, S.R., Pilarski, R.T., Donenberg, T., Shapiro, C., Hammond, L.S., Miller, J., Brooks, K.A., Cohen, S., Tenenholz, B., Desai, D., Zandvakili, I., Royer, R., Li, S., Narod, S.A. (2009) The Prevalence Of BRCA1 Mutations Amoung Young Women With Triple Negative Breast Cancer. *BMC Cancer* **9**: 86-90

Yu, Y., Xiao, C-H., Tan, L-D., Wang, Q-S., Li, X-Q., Feng, Y-M. (2014) Cancer-Associated Fibroblasts Induce Epithelial-Mesenchymal Transition Of Breast Cancer Cells Through Paracrine TGF-B Signaling. *British Journal Of Cancer* **110:** 724-732

Yue, W., Wang, J., Li, Y., Fan, P., Liu, G., Zhang, N., Conaway, M., Wang, H., Korach, K.S., Bocchinfuso, W., Santen, R. (2010) Effects Of Estrogen On Breast Cancer Development: Role Of Estrogen Receptor Independent Mechanisms. *International Journal Of Cancer* **127**: 1748-1757

Zeisberg, E.M., Potenta, S., Xie, L., Zeisberg, M., Kalluri, R (2007) Discovery Of Endothelial To Mesenchymal Transition As A Source For Carcinoma-Associated Fibroblasts. *Cancer Research* **67:** 10123-10128

Zhang, H., Kolb, F.A., Brondani, V., Billy, E., Filipowicz, W. (2002) Human Dicer Preferentially Cleaves Dsrnas At Their Termini Without A Requirement For ATP. *EMBO J* **21:** 5875-5885

Zhang, J.G., Wang, J.J., Zhao, F., Liu, Q., Jiang, K., Yang, G.H. (2010) Microrna-21 (Mir-21) Represses Tumour Suppressor PTEN And Promotes Growth And Invasion In Non-Small Cell Lung Cancer (NSCLC). *Clinica Chimica Acta* **411**: 846-852

Zhang, J., Wang, Y., Yin, Q., Zhang, W., Zhang, T., Niu, Y. (2013) An Associated Classification Of Triple Negative Breast Cancer: The Risk Of Relapse And The Response To Chemotherapy. *International Journal Of Experimental Pathology* **6**: 1380-1391

Zhang, P., Zhang, P., Shi, B., Zhou, M., Jiang, H., Zhang, H., Pan, X., Gao, H., Sun, H., Li, Z. (2014) Galectin-1 Overexpression Promotes Progression And Chemoresistance To Cisplatin In Epithelial Ovarian Cancer. *Cell Death Dis* **5**: E991 Zhao, L.J., Hua, X., He, S.F., Ren, H., Qi, Z.T. (2011) Interferon Alpha Regulates MAPK And STAT1 Pathways In Human Hepatoma Cells. *Virol J* 8: 157

Zheng, S. R., Guo, G. L., Zhai, Q., Zou, Z. Y., Zhang, W. (2013) Effects Of Mir-155 Antisense Oligonucleotide On Breast Carcinoma Cell Line MDA-MB-157 And Implanted Tumors. *Asian Pac J Cancer Prev* **14:** 2361-6.

Zheng, H. (2017) The Molecular Mechanisms Of Chemoresistance In Cancers. Oncotarget 8: 59950-59964

Zhou, A., Hassel, B.A., Silverman, R.H. (1993) Expression Cloning Of 2-5A-Dependent Rnaase: A Uniquely Regulated Mediator Of Interferon Action. *Cell* **72**: 753-765

Zhou, S., Huang, Q., Zheng, S., Lin, K., You, J., Zhang, X. (2016) Mir-27a Regulates The Sensitivity Of Breast Cancer Cells To Cisplatin Treatment Via BAK-SMAC/DIABLO-XIAP Axis. *Tumour Biol* **37:** 6837-45.

Zhou, L., Yang, K., Wickett, R.R., Zhang, Y. (2016) Dermal Fibroblasts Induce Cell Cycle Arrest And Block Epithelial-Mesenchymal Transition To Inhibit The Early Stage Melanoma Development. *Cancer Med* **5**: 1566-1579

Zhu, H., Han, C., Lu, D., Wu, T. (2014) Mir-17-92 Cluster Promotes Cholangiocarcinoma Growth. Evidence For PTEN As Downstream Target And IL-6/Stat3 As Upstream Activator. *Am J Pathol* **184**: 2828-2839

Zhu, H., Wu, H., Liu, X., Evans, B. R., Medina, D. J., Liu, C.-G., Yang, J.-M. (2008) Role Of Microrna Mir-27a And Mir-451 In The Regulation Of MDR1/P-Glycoprotein Expression In Human Cancer Cells. *Biochemical Pharmacology* **76:** 582-588.

Zhu, J., Zheng, Z., Wang, J., Sun, J., Wang, P., Cheng, X., Fu, L., Zhang, L., Wang, Z., Li, Z. (2014) Different Mirna Expression Profiles Between Human Breast Cancer Tumors And Serum. *Front Genet* **5**: 149

Zoglmeier, C. (2011) Cpg Blocks Immunosuppression By Myeloid-Derived Suppressor Cells In Tumour-Bearing Mice. *Clinical Cancer Research* **17:** 1765-1775